Torsten Schoeneberg

SEMISIMPLE LIE ALGEBRAS AND THEIR CLASSIFICATION OVER p-ADIC FIELDS

MÉMOIRES DE LA SMF 151

Société Mathématique de France 2017

Comité de rédaction

Christine BACHOC Emmanuel BREUILLARD Yann BUGEAUD Jean-François DAT Marc HERZLICH O' Grady KIERAN Raphaël KRIKORIAN Julien MARCHÉ Laurent MANIVEL Emmanuel RUSS Christophe SABOT Wilhelm SCHLAG

Pascal HUBERT (dir.)

Diffusion

Maison de la SMF B.P. 67 13274 Marseille Cedex 9 France christian.smf@cirm-math.fr AMS P.O. Box 6248 Providence RI 02940 USA www.ams.org

Tarifs 2017

Vente au numéro : 35 € (\$ 52) Abonnement électronique : 113 € (\$ 170) Abonnement avec supplément papier : 162 €, hors Europe : 186 € (\$ 279) Des conditions spéciales sont accordées aux membres de la SMF.

Secrétariat : Nathalie Christiaën

Mémoires de la SMF Société Mathématique de France Institut Henri Poincaré, 11, rue Pierre et Marie Curie 75231 Paris Cedex 05, France Tél : (33) 01 44 27 67 99 • Fax : (33) 01 40 46 90 96 memsmf@ihp.fr • http://smf.emath.fr/

© Société Mathématique de France 2017

Tous droits réservés (article L 122–4 du Code de la propriété intellectuelle). Toute représentation ou reproduction intégrale ou partielle faite sans le consentement de l'éditeur est illicite. Cette représentation ou reproduction par quelque procédé que ce soit constituerait une contrefaçon sanctionnée par les articles L 335–2 et suivants du CPI.

> ISSN 0249-633-X (print) 2275-3230 (electronic) ISBN 978-2-85629-859-6

> > Stéphane SEURET Directeur de la publication

MÉMOIRES DE LA SMF 151

SEMISIMPLE LIE ALGEBRAS AND THEIR CLASSIFICATION OVER p-ADIC FIELDS

Torsten Schoeneberg

Société Mathématique de France 2017 Publié avec le concours du Centre National de la Recherche Scientifique $T. \ Schoeneberg$

Torsten Schoeneberg, Camosun College, 3100 Foul Bay Rd, Victoria BC V8P 5J2. E-mail:torsten.schoeneberg@gmail.com

2010 Mathematics Subject Classification. — 17B05, 17B20, 17B22.

Key words and phrases. — Semisimple Lie algebras, Classification, p-adic fields, Tits index.

SEMISIMPLE LIE ALGEBRAS AND THEIR CLASSIFICATION OVER p-ADIC FIELDS

Torsten Schoeneberg

Abstract. — We study semisimple Lie algebras over fields of characteristic zero, with emphasis on \mathfrak{p} -adic fields and aiming at classification. We first transfer parts of the structure theory of reductive algebraic groups to our setting, with some variations. Classifying invariants are attached to Lie algebras and visualised with Satake-Tits diagrams. We give necessary and sufficient criteria for these diagrams. Over general fields of characteristic zero, we then classify all quasi-split forms, and we adapt an older classification theory for the classical types A-D to our language. Finally we focus on \mathfrak{p} -adic fields, where we achieve a complete classification by combination of certain well-known properties of these fields with our general results and methods, and we discuss the relation of this with a theorem of Kneser. This extends work by Weisfeiler.

Résumé (Algèbres de Lie semi-simples et leur classification sur les corps p-adiques)

Nous étudions les algèbres de Lie semi-simples sur les corps de caractéristique nulle, où l'accent est mis sur les corps p-adiques, l'objectif étant leur classification. Nous transférons d'abord certaines parties de la théorie de la structure des groupes réductifs dans notre contexte, avec quelques variations. Des invariants classifiants sont attachés aux algèbres de Lie et sont visualisés à l'aide de diagrammes de Satake-Tits. Nous donnons des critères nécessaires et suffisants pour ces diagrammes. Sur les corps généraux de caractéristique nulle, nous classifions ensuite toutes les formes quasi-déployées et nous traduisons une théorie ancienne de classification pour les types classiques A-D dans notre langue. Nous mettons enfin l'accent sur les corps p-adiques, où nous obtenons une classification complète par combinaison de certaines propriétés bien connues sur ces corps avec nos résultats généraux et nos méthodes, et nous abordons la relation de ces résultats avec un théorème de Kneser. Tout cela prolonge un travail de Weisfeiler.

CONTENTS

1.	Introduction	3
2.	Prerequisites	9 9 15 16
3.	Structure theory and the isomorphism theorem3.1. Definition of the invariants3.2. Special cases and examples3.3. The isomorphism theorem3.4. Visualisation: Satake-Tits diagrams	19 19 47 54 61
4.	Classification over general fields	65 68 72 77 79 95
5.	Classification over special fields5.1. C1 fields and unramified splitting for p-adic fields5.2. Type AI and Kneser's theorem5.3. Tools for ruling out anisotropic forms: The prime degree case5.4. Types AII, B, C and D over p-adic fields5.5. Exceptional Lie algebras over p-adic fields	99 99 101 103 112 116
6.	Concluding remarks 6.1. The field $k = \mathbb{R}$ 6.2. <i>k</i> -rational approaches (Allison, Seligman) 6.3. Explicit constructions for exceptional types	141 141 141 142
Bi	Bibliography	

To the memory of Michel Lazard and Boris Weisfeiler

[...] Und es starben Noch andere viel. Mit eigener Hand Viel traurige, wilden Muths, doch göttlich Gezwungen, zulezt, die anderen aber Im Geschike stehend, im Feld. [...]

Hölderlin, Mnemosyne



CHAPTER 1

INTRODUCTION

In 1964, Boris Weisfeiler published the short note [We1] on the classification of semisimple Lie algebras over a p-adic field. It is the purpose of this work to explain this note, append proofs to it, and put it into mathematical context.

Let us first give some mathematical-historical background. The Killing-Cartan classification of semisimple Lie algebras over the complex numbers \mathbb{C} is one of the standard classification theorems of modern mathematics. Both Killing, who had the results, and E. Cartan, who gave streamlined proofs in his *Thèse* of 1894, were interested in "infinitesimal groups" – as they were called then – mainly as a means to study Lie groups. In this spirit, Cartan also achieved the classification over the ground field \mathbb{R} . It is no wonder that fundamental concepts in the theory are named after them.

Likewise, the Weyl group is named after Weyl, who introduced this powerful tool in his approach to the subject in the 1920s. Cartan had already reduced the classification to that of root systems. Whereas he had achieved this classification in a computational way, Weyl's methods paved a new route, which was then taken by Coxeter, Witt and Dynkin. It also became clear that the classification over \mathbb{C} really is one of the *split* semisimple Lie algebras, which are the only ones over an algebraically closed field. With the works of Chevalley and Cartier's "Séminaire Sophus Lie" in the 1950s, this theory seems to have reached its definite form, in which it was presented in Bourbaki's fundamental treatment [**Bo2**] and found its way in many textbooks. Most mathematicians will have heard of the four classical families A-D and the five exceptional types E_6, E_7, E_8, F_4 and G_2 .

But Weyl had paved more than one way. He was the one who (orally) introduced the name "Lie algebra", and he promoted the subject as a means in itself which could be treated analogously to associative algebras – which, at the same time, were researched thoroughly by algebraists and number theorists, over various ground fields. Such a study for Lie algebras was initiated independently by Landherr [Lan1] and Jacobson [Jac1] in 1935. Landherr made the fundamental observation that one can start from

the known classification over an algebraically closed field and then investigate forms of the respective types: that is, take one type from the Killing-Cartan list and then find, over your ground field k, those Lie algebras which fall in this type after scalar extension to an algebraic closure \bar{k} . In modern terminology, he achieved this a) for the *inner forms* of type A over a general field of characteristic 0, and b) for all forms of type A over \mathbb{R} (reproving Cartan's results) and all \mathfrak{p} -adic fields, thinking in the direction of number fields and a local-global principle like the one Hasse had introduced for associative algebras. Jacobson on the other hand found a method that could deal with (almost) all the other classical types, i.e. *B-D* and the *outer forms* of type A, over any field of characteristic 0. Both Landherr's and Jacobson's approach can be described as enveloping the Lie algebras with certain associative algebras and then using known classification results for these. Jacobson and others went on to find similar approaches to the exceptional types, using various other kinds of algebras.

Meanwhile, after the Lie algebras had stepped out of the shadow of Lie groups, the theory of *algebraic groups* had emerged and, with the revolution in algebraic geometry, soon adopted a generality which left Lie algebras behind. With Weil's fundamental paper [**Wei**], the Landherr-Jacobson approach could be seen as a variant of a special case of *Galois cohomology* at work.

Mathematicians now aimed at a classification of semisimple, or more generally reductive, algebraic groups over fields (and then, schemes) as general as possible. It was certainly understood, not only by Weisfeiler, that by their close connection in characteristic 0, this would comprise the classification of semisimple Lie algebras. Many people worked on the group case, where the Borel-Tits paper [**BT1**] was a highlight, not least because of the generality of the ground field. Although Tits had announced most of the results earlier, it was Satake who had first published (in [**Sat2**]) proofs for some central theorems, at least over perfect ground fields. He would later extend his approach in [**Sat3**]. On the other hand, the scope of Tits' "Table II" in [**Tit1**] is intimidating. Satake's and Tits' visualisations of the classification are remarkably similar, and these *Satake-Tits diagrams* will abound in this work. They are also at the heart of Weisfeiler's note.

We proceed as follows.

▷ Chapter 2 collects some preliminaries and gives references for results we will use freely. In particular, we assume the classification of the split case via root systems and Dynkin diagrams to be known.

 \triangleright In Chapter 3 we start with the structure theory of semisimple Lie algebras over any field k of zero characteristic. Our general approach is to translate Satake's method to our setting, which is certainly what Weisfeiler did. A central concept are the *toral subalgebras* which are the analogue of tori. There is, however, one major deviation from Satake's method: namely, we start with a self-contained construction of k-rational root systems. This generalisation of the well-known split case is due to Seligman [Sel2] and more or less straightforward, its most profound ingredient being the Jacobson-Morozov theorem. Then we translate a bulk of Satake's and Borel-Tits' work to Lie algebras. One major tool is the elementary but fruitful notion of Γ -bases which bring root system combinatorics in balance with Galois actions. A recurring theme is the treatment of the relative Weyl group: its realisation as a subquotient of the absolute Weyl group, and simultaneously as a quotient of certain automorphism groups of the Lie algebra, is a highlight of this chapter. To achieve it, we use the k-rational root system and thereby circumvent some technical parts of Satake's method.

The main goal, however, is to attach two invariants to a semisimple Lie algebra, its *index* and its *anisotropic kernel*. In Section 3.1, we define them and show that in a natural manner they only depend on the isomorphism class of the Lie algebra. In Section 3.2 we discuss opposite extremes of these invariants – the *anisotropic* and *(quasi-)split* cases –, which play a major role in the classification, and we give basic but instructive examples. In Section 3.3 we prove the isomorphism theorem which describes how the invariants indeed classify semisimple Lie algebras. In Section 3.4 we finally introduce the Satake-Tits diagrams that visualise large parts of the invariants. At this point, the reader should be able to read Weisfeiler's table.

 \triangleright In Chapter 4 we begin to prove its correctness. The question is now largely translated to: which diagrams can occur over a given field k? It turns out in Section 4.1 that one can reduce this question to connected diagrams. In Section 4.2, we bring Galois cohomology into play; it gives a powerful tool as well as the clarifying terminology of *outer and inner forms of various types*. Then we present, in Sections 4.3 and 4.4, some general principles which exclude or produce *admissible* diagrams, or certain forms, over *any* characteristic zero field.

In this generality we also present the classification of the classical types in Section 4.5. The key ideas here are still the ones of Landherr and Jacobson, polished in the spirit of Galois cohomology, so that e.g. the appearance of *Brauer groups* is not too surprising. This approach is covered in the textbooks [Jac6] and [Sel1], but we only follow it for a while. Namely, their classification translates to one of involutorial algebras, which in turn are classified by certain equivalence classes of (skew-)hermitian spaces. This is good and fine, but we prefer to classify with the theory of Chapter 3 by assigning the corresponding Satake-Tits diagrams. Proceeding this way, some of the more delicate uniqueness issues in the associative-algebraic classification can be avoided. Indeed its full strength will only be needed for inner forms of type A, which had already been treated completely in Landherr's very first paper. In Section 4.6, we discuss and construct all quasi-split Lie algebras, whose classification is equivalent to that of quadratic and cubic extensions of the ground field.

 \triangleright In Chapter 5 we turn to special fields. The p-adic fields are our primary target, but it is useful to look first at C1 fields. In the language of Serre's Galois cohomology [Ser3], they are one-dimensional, and with a result by Springer we show in

Section 5.1 that over these fields all semisimple Lie algebras are quasi-split. Now \mathfrak{p} -adic fields are not of this kind – cohomologically, they are rather two-dimensional –, but their maximal unramified extensions are, and this suffices to show Weisfeiler's first theorem:

THEOREM (1). — If \mathfrak{g} is a semisimple Lie algebra over a \mathfrak{p} -adic field k, then there is a finite unramified extension $\mathfrak{K}|k$ such that $\mathfrak{g}_{\mathfrak{K}}$ is quasi-split.

This implies that for most forms there exist unramified (in particular, cyclic) splitting extensions. In Section 5.2 we classify the inner forms of type A over \mathfrak{p} -adic fields, and then discuss Weisfeiler's second theorem which says that anisotropic forms over \mathfrak{p} -adic fields are very rare – we have already encountered all of them:

THEOREM (2). — All simple anisotropic Lie algebras over a \mathfrak{p} -adic field k are derived algebras of k-division algebras. In particular, they are inner forms of type A.

We connect it with Kneser's theorem $[\mathbf{Kne}]$ about the vanishing of a certain cohomology group. On the basis of what we have until there, Kneser's result implies both theorem (2) and Weisfeiler's table, hence the complete classification over \mathfrak{p} -adic fields. However, Weisfeiler's work had appeared earlier, and it is not clear from $[\mathbf{We1}]$ how he arrived at his results, or even whether he concluded the theorem from the table or vice versa.

We undertake the verification in Sections 5.3 to 5.5. Specifically, in 5.3. we collect some technical facts and rule out certain anisotropic exceptional forms. In 5.4 we verify both theorem (2) and the table for the classical types. ⁽¹⁾ The special **p**-adic ingredient here is a well-known theorem which says that all hermitian, skew-hermitian and related forms in more than very few variables are isotropic; apart from this, we only use our general results from Chapter 4. In Section 5.5 we attack the exceptional types. For some diagrams, suitable Lie algebras are constructed with the techniques of Chapter 4. It turns out that we can exclude all other diagrams with our general methods and the results for the classical types – except for the anisotropic ones. To exclude these, we extend the results of 5.3, which works for almost all the anisotropic forms. The only ones for which we still have to rely on Kneser's theorem are one inner form of type E_6 , and one trialitarian form of type D_4 (although to be fair, the classification in types E_7 and E_8 rests on the result for E_6).

 \triangleright In Chapter 6 we very briefly address the case $k = \mathbb{R}$ and finish with concluding remarks about other approaches.

^{1.} One outer form of type D_{2m} was overlooked by Weisfeiler, see Section 5.4.4.

Acknowledgements. — This work is an extended version of my doctoral thesis. Work on it was supported by the SFB 478 and 878 of the *Deutsche Forschungsgemeinschaft*, and by a scholarship of the *Deutsche Telekom Stiftung* which I gratefully acknowledge.

I thank my advisor Peter Schneider for his patient support and for giving me the opportunity to write this thesis. Thanks to all members of the AG, especially Marten Bornmann, Jan Kohlhaase and Matthias Wulkau, for several clarifying discussions concerning topics in and around this thesis. Dimitri Wegner and Esther Engberding provided support for T_EXing the diagrams, Nora Loges and Hsini Han translated [**Xue**] for me, and Dominic Enders and Olga Varghese made various helpful suggestions. Thanks for that (and for more). Finally, many thanks to the referee for helpful and motivating suggestions, to the editors for their patience, and to my wife Arelia for her support during the editing process.

Die Jahre der Promotion waren für mich lehrreich, mit einigen Höhen und Tiefen. Von vielen, einige schon genannt, habe ich in dieser Zeit Unterstützung erfahren. Kein anderer ist vergessen, und nennen will ich Anna, Daniel, Gerd, Sebastian, Selçuk, Sven, Wolfgang, Frau Schlegel, meine Eltern und meinen Bruder.

CHAPTER 2

PREREQUISITES

Notations:

 \triangleright n is a (natural) number and p is a prime.

 $\triangleright \mathbb{C}, \mathbb{R}, \mathbb{Q}, \mathbb{Z}, \mathbb{F}_p$ have the usual meanings, likewise $M_{m \times n}(.), M_n(.)$ and $GL_n(.)$.

 $\triangleright \mathbb{F}_q$ denotes a finite extension of \mathbb{F}_p .

 \triangleright We use Gauß' floor function $\lfloor x \rfloor$ and Euler's totient function $\phi(n)$.

▷ For a ring R, the group of units is denoted by R^* or R^{\times} . For a vector or matrix X, the transpose is denoted by ${}^{t}X$.

 \triangleright We write I_n for the unit matrix in GL_n , and E_{ij} is the matrix with (i, j)-th entry 1 and all other entries 0. A diagonal matrix is denoted by $diag(x_1, \ldots, x_n)$.

2.1. Lie algebras and root systems

DEFINITION 2.1.1. — Let R be a commutative unital ring. A *Lie algebra* over R is an R-module \mathfrak{g} together with an R-bilinear map – the *Lie bracket* –

$$[.,.]: \mathfrak{g} \times \mathfrak{g} \longrightarrow \mathfrak{g}, \quad (x,y) \longmapsto [x,y]$$

satisfying, for all $x, y, z \in \mathfrak{g}$, the relations

[x, x] = 0 and [x, [y, z]] + [y, [z, x]] + [z, [x, y]] = 0.

We will only deal with the case where R is a field, so that \mathfrak{g} is a vector space, which is most often assumed to be *finite dimensional*. In the following, a ring denoted by the letter k will always be meant to be a field.

For the concepts of *ideals* and *subalgebras*, *products* of Lie algebras, *abelian* (= commutative), nilpotent and solvable Lie algebras, as well as the centre $\mathfrak{z}(\mathfrak{g})$, the radical $\mathfrak{r}(\mathfrak{g})$, and the derived Lie algebra $\mathfrak{D}\mathfrak{g} = [\mathfrak{g}, \mathfrak{g}]$, the derived series $\mathcal{D}^i\mathfrak{g}$, descending central series $\mathcal{C}^i\mathfrak{g}$, and also representations, in particular the adjoint representation, and the Killing form of a Lie algebra, see [**Bo2**, I]. For subsets H, K of a Lie algebra, [H, K] denotes the subalgebra generated by all [h, k] for $h \in H, k \in K$. Further, for a subset H of the Lie algebra \mathfrak{g} , we have its

 \triangleright centraliser $\mathfrak{z}_{\mathfrak{g}}(H) := \{x \in \mathfrak{g} : [x, h] = 0 \text{ for all } h \in H\}; \text{ it is a subalgebra, and if } H \text{ is an ideal of } \mathfrak{g}, \text{ then so is } \mathfrak{z}_{\mathfrak{g}}(H);$

▷ normaliser $\mathfrak{n}_{\mathfrak{g}}(H) := \{x \in \mathfrak{g} : [x, h] \in H \text{ for all } h \in H\}$; if H is a subalgebra of \mathfrak{g} , then $\mathfrak{n}_{\mathfrak{g}}(H)$ is the largest among the subalgebras \mathfrak{n} of \mathfrak{g} such that H is an ideal in \mathfrak{n} .

We state the concept of semisimplicity, as well as the stronger one of simplicity and the weaker one of reductivity. For the assertions in the following, see [**Bo2**, I.6].

- DEFINITION 2.1.2. Let \mathfrak{g} be a finite-dimensional Lie algebra over k.
 - i. g is called *simple* if it is not abelian and its only ideals are g and {0}.
 Now let char(k) = 0.

ii. g is called *semisimple* if the following equivalent conditions are satisfied:

- (a) \mathfrak{g} is isomorphic to a product of simple Lie algebras.
- (b) The radical $\mathfrak{r}(\mathfrak{g})$ is zero.
- (c) \mathfrak{g} does not contain any non-zero abelian ideal.
- (d) \mathfrak{g} does not contain any non-zero solvable ideal.
- (e) The Killing form on \mathfrak{g} is non-degenerate.
- iii. \mathfrak{g} is called *reductive* if the following equivalent conditions are satisfied:
 - (a) \mathfrak{g} is isomorphic to the product of an abelian and a semisimple Lie algebra.
 - (b) \mathfrak{g} is isomorphic to the product of its centre $\mathfrak{z}(\mathfrak{g})$ and its derived algebra $\mathfrak{D}\mathfrak{g}$.
 - (c) $\mathfrak{D}\mathfrak{g}$ is semisimple.
 - (d) The adjoint representation of ${\mathfrak g}$ is a direct sum of irreducible representations.
 - (e) Radical and centre of \mathfrak{g} are equal: $\mathfrak{r}(\mathfrak{g}) = \mathfrak{z}(\mathfrak{g})$.
- iv. A subalgebra \mathfrak{h} of \mathfrak{g} is called *reductive in* \mathfrak{g} if the representation $\mathfrak{h} \to \mathfrak{gl}(\mathfrak{g})$, $h \mapsto \operatorname{ad}_{\mathfrak{g}} h$ is the direct sum of irreducible representations.

There are no semisimple Lie algebras of dimension 1 or 2, and over an algebraically closed k there is up to isomorphism a unique simple one of dimension 3, namely, $\mathfrak{sl}_2(k)$.

In a semisimple Lie algebra \mathfrak{g} , we have well-behaved notions of *semisimple* and *nilpotent elements*, and we have *Jordan decomposition*: every element $x \in \mathfrak{g}$ can be written uniquely as the sum $x = x_s + x_n$ of a semisimple and a nilpotent one commuting with each other, called the *semisimple* and *nilpotent component* of x, respectively; further, $\mathrm{ad}_{\mathfrak{g}}(x_s)$ and $\mathrm{ad}_{\mathfrak{g}}(x_n)$ are the semisimple and nilpotent parts of $\mathrm{ad}_{\mathfrak{g}}(x) \in \mathrm{End}_k(\mathfrak{g})$, and can be written as polynomials in $\mathrm{ad}_{\mathfrak{g}}(x)$ without constant term. For all this, see [**Bo2**, I.6.3] and [**Bo1**, VII.5, especially nos. 7–9]. ⁽¹⁾ A subalgebra is called *nil* if it consists of nilpotent elements.

^{1.} For semisimple and nilpotent components of a vector space endomorphism, the Bourbaki reference in earlier (Hermann) editions was *Algèbre* VIII.9.

We also mention the base change routines that we will often use:

DEFINITION 2.1.3. — Let K|k be a field extension.

- i. If \mathfrak{G} is a Lie algebra over K, then in the obvious way it can be considered as a Lie algebra over k. Call this Lie algebra $R_{K|k}\mathfrak{G}$, the scalar restriction of \mathfrak{G} to k.
- ii. If \mathfrak{g} is a Lie algebra over k, then the tensor product $K \otimes_k \mathfrak{g}$ becomes a Lie algebra over K when as Lie bracket $[.,.]_K$ we take the unique K-bilinear map satisfying

$$[a \otimes x, b \otimes y]_K = ab \otimes [x, y]$$

for $a, b \in K, x, y \in \mathfrak{g}$. Call this Lie algebra \mathfrak{g}_K , the scalar extension of \mathfrak{g} to K.

Note that in scalar restriction, the usual assumption of finite dimension of the Lie algebra only remains intact for a *finite* extension K|k (or if the algebra is zero).

If $\mathfrak{h} \subset \mathfrak{g}$ is a subalgebra, we can in an obvious way, and will without further notice, identify \mathfrak{h}_K with a (K-)subalgebra of \mathfrak{g}_K . Also, we will occasionally identify \mathfrak{g} with the (k-)subalgebra of $R_{K|k}\mathfrak{g}_K$ given by $1 \otimes \mathfrak{g} \subset \mathfrak{g}_K$.

We will use the notion of *Cartan subalgebras* (cf. [**Bo2**, VII.2]) which in our case are better viewed as the the maximal toral subalgebras, see Chapter 3 and especially 3.1.5. With these, the notions of *Borel subalgebras* and parabolic subalgebras are defined, first in the split case (cf. [**Bo2**, VIII.3.3–4]) and then in the general case as those that become Borel or parabolic after scalar extension to an algebraic closure (cf. [**Bo2**, VIII.3.5]). In particular, Borel subalgebras do not necessarily exist. In fact, the question of existence of Borel subalgebras will play an important role in our classification, and we will make use of the results of [**Bo2**, VIII.10] which very roughly say that "many nilpotent elements" is equivalent to "smaller parabolic subalgebras".

The rank $\operatorname{rk}(\mathfrak{g})$ of a Lie algebra \mathfrak{g} and the notion of a regular element are defined in [**Bo2**, VII.2]. If $\operatorname{char}(k) = 0$, all Cartan subalgebras of \mathfrak{g} have dimension $\operatorname{rk}(\mathfrak{g})$, and they are precisely the $\mathfrak{g}^0(x)$ for regular elements x. Here,

$$\mathfrak{g}^0(x) := \{ y \in \mathfrak{g} : (\mathrm{ad}_\mathfrak{g}(x)^n)(y) = 0 \text{ for some } n \ge 1 \}.$$

For semisimple x, we have $\mathfrak{g}^0(x) = \ker(\mathrm{ad}_\mathfrak{g}(x))$, and the dimension of this space is $\mathrm{rk}(\mathfrak{g})$ if and only if x is regular; else it is strictly larger.

REMARK 2.1.4 (Stabilities under scalar extension). — Let K|k be an extension of fields of characteristic zero, and let \mathfrak{g} be a Lie algebra over k.

- i. We have $(\mathfrak{z}(\mathfrak{g}))_K = \mathfrak{z}(\mathfrak{g}_K)$; and for any subalgebra $\mathfrak{h} \subseteq \mathfrak{g}$, we have $(\mathfrak{n}_{\mathfrak{g}}(\mathfrak{h}))_K = \mathfrak{n}_{\mathfrak{g}_K}(\mathfrak{h}_K)$, see [**Bo2**, I.3.8]. (The analogous statement for the centraliser is wrong in general.)
- ii. If $\kappa(.,.)$ is the Killing form of \mathfrak{g} , the Killing form of \mathfrak{g}_K is $\kappa_K(a \otimes x, b \otimes y) = ab \kappa(x, y)$, see [**Bo2**, I.3.8].
- iii. The following properties are satisfied by \mathfrak{g} if and only if they are satisfied by \mathfrak{g}_K :

CHAPTER 2. PREREQUISITES

- (a) Being semisimple, see [**Bo2**, I.6.10].
- (b) Being reductive, see [Bo2, I.6.10].
- (c) Being nilpotent, see [Bo2, I.4.5].
- (d) Being solvable, see [Bo2, I.5.6].
- (e) Being a Cartan subalgebra (of another Lie algebra \mathfrak{g}' resp. \mathfrak{g}'_K), see [**Bo2**, VII.2.2, Prop. 3].
- iv. If \mathfrak{g}_K is simple, \mathfrak{g} is simple, see [**Bo2**, I.6.10]. The converse is not true in general, cf. Section 4.1; at least by the above, \mathfrak{g}_K is semisimple for simple \mathfrak{g} .

REMARK 2.1.5 (Stabilities under scalar restriction). — Let K|k be a *finite* extension of fields of characteristic zero, and let \mathfrak{G} be a Lie algebra over K. Set $\mathfrak{g} := R_{K|k}\mathfrak{G}$. We have $\mathfrak{z}(\mathfrak{G}) = \mathfrak{z}(\mathfrak{g})$. The following properties are satisfied by \mathfrak{G} if and only if they are satisfied by \mathfrak{g} :

- i. Being simple, see [**Bo2**, I.6.10].
- ii. Being semisimple, see [Bo2, I.6.10].
- iii. Being nilpotent: follows from [Bo2, I.1.9].
- iv. Being solvable: follows from [Bo2, I.1.9].

We will make use of the *theorems of Engel* (cf. [**Bo2**, I.4.2]), *Levi-Malcev* (cf. [**Bo2**, I.6.8]) and *Jacobson-Morozov* (cf. [**Bo2**, VIII.11]) for which, however, we use a slightly different normalisation:

DEFINITION 2.1.6. — An \mathfrak{sl}_2 -triple in a Lie algebra \mathfrak{g} is a triple (x, h, y) of elements of \mathfrak{g} distinct from (0, 0, 0) and such that

$$[h, x] = 2x, \quad [h, y] = -2y, \quad [x, y] = h.$$

(Bourbaki's y is our -y.)

Accordingly, for us the Lie algebra $\mathfrak{sl}_2(k)$ has the standard basis:

$$X = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}, \quad Y = \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix}, \quad H = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$$

The Jacobson-Morozov theorem states that every nilpotent element $x \neq 0$ in a semisimple Lie algebra \mathfrak{g} can be extended to an \mathfrak{sl}_2 -triple (x, h, y). An \mathfrak{sl}_2 -triple in \mathfrak{g} is the same as a (necessarily injective) homomorphism $\mathfrak{sl}_2(k) \hookrightarrow \mathfrak{g}$ and thus a way to view \mathfrak{g} as a representation of $\mathfrak{sl}_2(k)$.

From this representation theory (cf. [**Bo2**, VIII.1]) we use the following. Let E be a representation of $\mathfrak{sl}_2(k)$, and for an eigenvalue λ of H in E (a "weight"), denote the corresponding eigenspace by E_{λ} . The following facts are stated in or follow immediately from *loc. cit.* Proposition 2 and its Corollaire (in particular, part iii is Corollaire (ii) with $1 = i \ge p = 2 - \lambda$).

LEMMA 2.1.7. — Let E be a finite-dimensional representation of $\mathfrak{sl}_2(k)$.

- i. If λ is a weight, then it is an integer, and |λ|, |λ| − 2,..., −|λ| are weights too. Consequently, the largest and smallest weights are integers symmetric around 0, i.e. they are of the form λ and −λ for λ ∈ Z_{>0}.
- ii. E is the direct sum of the E_{λ} .
- iii. If $\lambda > 0$ is an eigenvalue of H in E, then X maps $E_{\lambda-2}$ onto E_{λ} .

Another nice application is:

LEMMA 2.1.8. — Let $\mathfrak{h} \subseteq \mathfrak{g}$ be an inclusion of semisimple Lie algebras. Then every $x \in \mathfrak{h}$ which is nilpotent in \mathfrak{h} is nilpotent in \mathfrak{g} .

Namely, by Jacobson-Morozov, there is a homomorphism $\mathfrak{sl}_2 \hookrightarrow \mathfrak{h}$ sending X to x, and viewing \mathfrak{g} as \mathfrak{sl}_2 -representation with respect to this gives that $\mathrm{ad}_{\mathfrak{g}}(x) = \mathrm{ad}(X)$ is nilpotent.

For *automorphisms of (semisimple) Lie algebras*, we follow [**Bo2**, VIII.5]. In particular:

DEFINITION 2.1.9. — We have the chain of subgroups $\operatorname{Aut}_e(\mathfrak{g}) \subset \operatorname{Aut}_0(\mathfrak{g}) \subset \operatorname{Aut}(\mathfrak{g})$, with strict inclusions in general, where $\operatorname{Aut}_e(\mathfrak{g})$ are the *elementary automorphisms*, that is, finite products of

$$\exp(\operatorname{ad} n) := x \longmapsto \sum_{k \ge 0} \frac{(\operatorname{ad}_{\mathfrak{g}}(n))^k}{k!} (x)$$

for nilpotent $n \in \mathfrak{g}$; and $\operatorname{Aut}_0(\mathfrak{g})$ are those automorphisms that become elementary after scalar extension to an algebraic closure. For a subset $H \subseteq \mathfrak{g}$ we set

 $\operatorname{Aut}(\mathfrak{g}, H) = \{ f \in \operatorname{Aut}(\mathfrak{g}) : f(H) = H, \quad \operatorname{Aut}_*(\mathfrak{g}, H) = \operatorname{Aut}_*(\mathfrak{g}) \cap \operatorname{Aut}(\mathfrak{g}, H) \}$

for $* \in \{0, e\}$.

We presuppose the classification of *split Lie algebras* via root systems as in [**Bo2**, VIII.4]; also, the matrix description of the "classical types" *A-D* in [**Bo2**, VIII.13] or [**Jac6**, IV. 6]. For *root systems*, the general reference is [**Bo2**, VI]. Let us state some basics in the following.

Let V be vector space over a field k with $\operatorname{char}(k) = 0$, and let V^* be its dual. For $\alpha \in V, \alpha^* \in V^*$, we define an element $s_{\alpha,\alpha^*} \in \operatorname{End}_k(V)$ by:

(1)
$$s_{\alpha,\alpha^*}(x) := x - \alpha^*(x)\alpha$$

If $\alpha^*(\alpha) = 2$, this is a *reflection* in the sense of [**Bo2**, V.2], in particular $s_{\alpha,\alpha^*}^2 = \mathrm{id}_V$.

Definition 2.1.10

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

i. A root system (in V) is a subset $R \subset V$ that satisfies:

- (RS1) R is finite, generates V, and $0 \notin R$.
- (RS2) For all $\alpha \in R$, there exists $\check{\alpha} \in V^*$ with $\check{\alpha}(\alpha) = 2$ and $s_{\alpha,\check{\alpha}}(R) \subseteq R$.
- (RS3) For all $\alpha \in R$, $\check{\alpha}(R) \subseteq \mathbb{Z}$.

As explained in [**Bo2**, VI.1.1], $\check{\alpha}$ in (RS2) is uniquely determined by α (and thus (RS3) is to be understood). For brevity, set

 $s_{\alpha} := s_{\alpha,\check{\alpha}}.$

- ii. For a root system R in V, we call
 - \triangleright rank $(R) := \dim_k V$ the rank of R,
 - $\triangleright A(R) := \{ \sigma \in \operatorname{Aut}_k(V) : \sigma(R) = R \}$ the automorphism group of R, and
 - $\triangleright W(R) :=$ subgroup of A(R) generated by the s_{α} for $\alpha \in R$ the Weyl group of R.
- iii. The root system R is called *reduced* if whenever $\alpha = c \cdot \beta$ for $\alpha, \beta \in R$, we have $c \in \{1, -1\}$.

We refer to [**Bo2**, VI] for the plentitude of properties that follow from these definitions. In particular, we will freely use the translation of the theory to and from the ground fields \mathbb{Q} and \mathbb{R} , the relations and proportions between roots in a root system, and the decomposition of a root system into *irreducible* root systems.

REMARK 2.1.11. — Let R be a root system (in the real vector space V). For $\sigma \in A(R)$, we have

$$(\sigma(\alpha)))(\sigma(\beta)) = \check{\alpha}(\beta)$$

for all $\alpha, \beta \in R$. Namely, there is an A(R)-invariant scalar product $\langle ., . \rangle$ on V such that

$$\check{\alpha}(.) = 2 \frac{\langle \alpha, . \rangle}{\langle \alpha, \alpha \rangle} \cdot$$

The notions of positive roots and bases of a root system are used. Recall in particular that W(R) acts simply transitively on the set of bases (cf. [**Bo2**, VI.1.5, Thm 2 and Rem. 4]), and that W(R) is normal in A(R) with $A(R)/W(R) \simeq \operatorname{Aut}(R, \Delta)$ for any basis Δ of R (cf. [**Bo2**, VI.1.5, Prop. 16 and 4.2, Cor.]); further, that this group identifies with the graph automorphisms of the *Dynkin diagram*. We have the usual list of irreducible reduced root systems/connected Dynkin diagrams A_n , B_n , C_n , D_n ($n \ge 2$), E_6, E_7, E_8, F_4, G_2 (with the repetitions $A_1 = B_1 = C_1$; $B_2 = C_2$; $D_2 = A_1 \times A_1$; $D_3 = A_3$), as well as the non-reduced BC_n , and we use the plates at the end of [**Bo2**, VI] for their properties. In particular we adhere to Bourbaki's labelling of the simple roots/vertices, although they look strange for the E types, where the "lower" vertex is α_2 :



A(R)/W(R) is the symmetric group S_3 for D_4 (an exceptional phenomenon which will occur as *triality* in the course of this work), of order two for $A_n (n \ge 2), D_n (n \ne 4)$ and E_6 , and trivial in all other cases.

2.2. p-adic fields, C1 fields, quaternion algebras

We have the field of *p*-adic numbers \mathbb{Q}_p with ring of integers \mathbb{Z}_p , and we call \mathfrak{p} -adic field a finite extension of \mathbb{Q}_p . Any finite extension of a \mathfrak{p} -adic field is again a \mathfrak{p} -adic field. A \mathfrak{p} -adic field k is a *local field* of characteristic zero. It is complete with respect to a discrete valuation $v_k : k \to \mathbb{Z} \cup \{\infty\}$ and has some \mathbb{F}_q as residue field. If k is a local field, we denote by \mathcal{O}_k its ring of integers, by \mathfrak{m}_k the maximal ideal of \mathcal{O}_k and by π_k a uniformiser, i.e. generator of \mathfrak{m}_k . The general reference for local fields is [**Ser2**]. Specifically, the concept of (un)ramified extensions will be important. Any local field k has a maximal unramified extension which is unique up to isomorphism (or unique inside a fixed algebraic closure \overline{k}) and will be denoted by k^{nr} ; see [**Ser2**, III.5].

DEFINITION 2.2.1 (Property C1). — We say that a field k has the C1 property or is C1 if the following holds: every homogeneous polynomial in n variables $P \in k[X_1, \ldots, X_n]$ of degree $d \in \{1, \ldots, n-1\}$ has a non-trivial zero, meaning that there is $(0, \ldots, 0) \neq (a_1, \ldots, a_n) \in k^n$ with $P(a_1, \ldots, a_n) = 0$.

At one point we use the

THEOREM 2.2.2 (cf. [Lang]). — If k is a local field, k^{nr} is C1.

Quaternion algebras will be of some importance in the classification. See [**Pie**, 1.6–1.7] and [**Inv**, 2.C] for the following basics: a quaternion algebra Q over a field k with $\operatorname{char}(k) \neq 2$ is described (not one-to-one) by two parameters $a, b \in k^*$:

$$Q := \left(\frac{a, b}{k}\right)$$

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

which is the 4-dimensional k-algebra with basis 1, i, j, ij and multiplication given by $i^2 = a, j^2 = b$ and ij = -ji which implies $(ij)^2 = -ab$. The three-dimensional space spanned by i, j and ij is that of *pure quaternions*. Over a p-adic field k, just as over \mathbb{R} , up to isomorphism there is only one quaternion division algebra (i.e. quaternion algebra which is a skew field). The one over \mathbb{R} , the Hamilton quaternions, is usually denoted as

$$\mathbb{H} = \left(\frac{-1, -1}{\mathbb{R}}\right)$$

while for those over a p-adic k, a standard description (for $p \neq 2$) is

$$\left(\frac{u,\pi_k}{k}\right)$$

where $u \in \mathcal{O}_k^* \setminus \mathcal{O}_k^{*2}$.

On a quaternion algebra $Q := \left(\frac{a,b}{k}\right)$, there is the so-called *standard involution*

 $\gamma: x1 + yi + zj + wij \mapsto x1 - yi - zj - wij,$

i.e. the k-linear map with eigenspaces k and the pure quaternions, of respective eigenvalues 1 and -1. It satisfies $\gamma^2 = \mathrm{id}_Q$ and $\gamma(st) = \gamma(t)\gamma(s)$ [!] for all $s, t \in Q$. We will need this in the discussion of the classical types of Lie algebras in Section 4.5.

2.3. Quadratic, symplectic and hermitian forms

Our general references for this theory are [**Bo1**, IX] and [**Lam**]. Since we are generally in characteristic $\neq 2$, we have the well-known equivalences between quadratic forms and symmetric bilinear forms. For symplectic (= alternate bilinear) forms, we will use the standard form in [**Bo1**, IX.5.1]. In left/right conventions, we follow Bourbaki and make the following

DEFINITION 2.3.1. — Let D be a skew field and $\sigma : D \to D$ an *involutorial antiautomorphism*, i.e. an additive map satisfying

$$\sigma(ab) = \sigma(b)\sigma(a)$$
 for all $a, b \in D$ and $\sigma^2 = \mathrm{id}_D$.

Let V be a left D-vector space. A bi-additive map $h: V \times V \to D$ satisfying

$$h(ax, by) = ah(x, y)\sigma(b)$$

for all $x, y \in V$ and $a, b \in D$ is called a σ -hermitian form (on V) if

$$h(y, x) = \sigma(h(x, y))$$

for all $x, y \in V$, and is called a σ -skew-hermitian form (on V) if

$$h(y,x) = -\sigma(h(x,y))$$

for all $x, y \in V$. Such an h is non-degenerate if h(x, V) = 0 implies x = 0.

Skew-hermitian forms are called "anti-hermitian" by Bourbaki; in any case, they are the ϵ -hermitian forms with $\epsilon = -1$. The notions of *(totally) isotropic* vectors or subspaces are defined in [**Bo1**, IX.4]. There we also find the fundamental

THEOREM 2.3.2 (Witt's decomposition theorem [Wit]). — Let k be a field with $char(k) \neq 2$, and let (V,q) be a quadratic space over k. Then there is a decomposition

$$(V,q) \simeq (V_0,0) \oplus (V_h,q_h) \oplus (V_a,q_a)$$

where $V_0 = \ker(q)$ is the radical, (V_h, q_h) is hyperbolic and (V_a, q_a) is anisotropic, and all three are uniquely determined up to isomorphism. If $V_0 = 0$, the integer $r := \frac{1}{2} \dim_k V_h$ is the maximal dimension of a totally isotropic subspace of (V, q), and is called the Witt index of (V, q).

Loc. cit. also contains its analogue for hermitian and skew-hermitian forms, which also have a well-defined Witt index if they satisfy the "condition T": For every $x \in V$, there is $a \in D$ such that

$$h(x, x) = a + \epsilon \sigma(a)$$

where $\epsilon = 1$ or -1 according to whether *h* is hermitian or skew-hermitian. We do not have to worry about this, because we are in characteristic $\neq 2$ and can set $a = \frac{1}{2}h(x, x)$.

If b is a non-degenerate symmetric bilinear form on an n-dimensional k-vector space, the discriminant d(b) is the well-defined residue in k^*/k^{*2} of $(-1)^{\lfloor \frac{1}{2}n \rfloor}$ times the determinant of a representing matrix of b; the sign ensures that hyperbolic spaces have discriminant 1.

CHAPTER 3

STRUCTURE THEORY AND THE ISOMORPHISM THEOREM

Let k be a field with $\operatorname{char}(k) = 0$. Unless mentioned otherwise, all Lie algebras are finite dimensional, and all field extensions K|k will be inside a fixed algebraic closure $\bar{k}|k$ with absolute Galois group $\mathcal{G} = \operatorname{Gal}(\bar{k}|k)$. One can check that nothing depends essentially on the choice of \bar{k} .

General references for the algebraic group analogue of this chapter are [**BT1**], [**Sat3**], [**Spr2**, 2..5–6] and [**Spr3**, Chap. 15–17].

Throughout this whole chapter, let \mathfrak{g} be a semisimple Lie algebra over k. Let $\kappa(.,.)$ be its Killing form.

3.1. Definition of the invariants

3.1.1. Toral subalgebras and rational root decompositions

DEFINITION 3.1.1 (Weights and roots). — Let \mathfrak{h} be any subalgebra of \mathfrak{g} . For any linear form $\alpha \in \mathfrak{h}^* := \operatorname{Hom}_k(\mathfrak{h}, k)$, we set

 $\mathfrak{g}_{\alpha} := \mathfrak{g}_{\alpha}(\mathfrak{h}) := \big\{ x \in \mathfrak{g} : [h, x] = \alpha(h) x \text{ for all } h \in \mathfrak{h} \big\}.$

If this subspace is non-zero, we call it the *weight space* of the *weight* α (of \mathfrak{h} in \mathfrak{g}). The non-zero weights are called the *roots* of \mathfrak{h} in \mathfrak{g} , the – obviously finite – set of these is denoted by $R(\mathfrak{g}, \mathfrak{h})$.

Note that in the above situation, $\mathfrak{g}_0 = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{h})$. Also, an immediate calculation with the Lie algebra axioms gives

(2)
$$[\mathfrak{g}_{\alpha},\mathfrak{g}_{\beta}]\subseteq\mathfrak{g}_{\alpha+\beta}$$

for all $\alpha, \beta \in \mathfrak{h}^*$.

For later use, we note:

LEMMA 3.1.2. — For $\alpha \in R(\mathfrak{g}, \mathfrak{h})$, every element of \mathfrak{g}_{α} is nilpotent in \mathfrak{g} .

Proof. — Let $x \in \mathfrak{g}_{\alpha}$. It suffices to show that $X := \operatorname{ad}_{\mathfrak{g}} x$ is nilpotent in the associative algebra $\operatorname{End}_{k}(\mathfrak{g})$. We will show that the trace $\operatorname{Tr}(X^{n})$ is 0 for all $n \geq 1$, which proves the assertion by [**Bo1**, VII.5.5, Cor. 4].⁽¹⁾

Choose $h \in \mathfrak{h}$ with $r := \alpha(h) \neq 0$, and set $H := \operatorname{ad}_{\mathfrak{g}}(h), Y := \frac{1}{r}X$. Then we have X = HY - YH, and because Y commutes with X^{n-1} (where $X^0 := 1$), we can write X^n as commutator

$$X^{n} = X^{n-1}(HY - YH) = (X^{n-1}H)Y - Y(X^{n-1}H)$$

which has trace 0 because Tr(AB) = Tr(BA) for any endomorphisms A, B. (The argument is a special case of [**Jac6**, II.5, Lemma 4]; for a proof explicitly using scalar extension, see [**Bo2**, VII.1.3, Prop. 10.(iv)].)

DEFINITION 3.1.3 (Toral and split toral subalgebras)

- i. A subalgebra \mathfrak{a} of \mathfrak{g} is *toral* if it is abelian and all its elements are semisimple.
- ii. A toral subalgebra \mathfrak{a} is called *split* (over k) if all its elements are ad-diagonalisable over k; that is, for every $x \in \mathfrak{a}$, there is a k-basis of \mathfrak{g} consisting of eigenvectors for $\mathrm{ad}_{\mathfrak{g}}(x)$ with all eigenvalues in k.
- iii. A maximal (split) toral subalgebra is one which is not strictly contained in another (split) toral subalgebra.

Remark 3.1.4

20

- i. Subalgebras of (split) toral subalgebras are (split) toral.
- ii. For an isomorphism $h : \mathfrak{g} \simeq \mathfrak{g}'$, the subalgebra $\mathfrak{a} \subseteq \mathfrak{g}$ is (maximal/split/maximal split) toral if and only if so is $h(\mathfrak{a}) \subseteq \mathfrak{g}'$.
- iii. If k is algebraically closed, every toral subalgebra is split.
- iv. The argument in [Hum, 8.1] shows that a subalgebra which consists of addiagonalisable elements is automatically abelian. Therefore, one could define a split toral subalgebra as a subalgebra consisting of ad-diagonalisable elements. On the other hand, the "abelian" condition is *not* redundant in the definition of toral subalgebras; in fact, we will encounter non-zero semisimple Lie algebras which consist entirely of semisimple elements, see 3.2.1 with example 3.2.5.
- v. Maximal (split) toral subalgebras always exist because our Lie algebras have finite dimension and we have the split toral subalgebra {0}.
- vi. The usual *Caveat* about the order of words is in order: whereas a split maximal toral subalgebra is also a maximal split toral subalgebra, not every maximal split toral subalgebra is a split maximal toral subalgebra (it is if and only if it is a maximal toral subalgebra).

PROPOSITION 3.1.5. — In our setting, the maximal toral subalgebras \mathfrak{t} are precisely the Cartan subalgebras of \mathfrak{g} . In particular, they equal their own centraliser and

^{1.} In earlier editions, this was Corollaire 4 in Algèbre VII.3.5.

normaliser, all maximal toral subalgebras of \mathfrak{g} have the same dimension (cf. [Bo2, VII.3.3]) and a split maximal toral subalgebra is a splitting Cartan subalgebra in the sense of [Bo2, VIII.2.1].

Proof. — In [**Bo2**, VII], this is left as exercise 3 to §2. That a Cartan subalgebra of \mathfrak{g} is maximal toral follows easily from [**Bo2**, VII.2.4]. The following proof of the converse, for maximal toral \mathfrak{t} , is a combination of the proof of [**Sel2**, §1, Prop. 1] with ideas of [**Hum**, 8.1–8.2] and [**Xue**].

Step 1. — For any toral subalgebra $\mathfrak{a} \subseteq \mathfrak{g}$, we have $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}) = \mathfrak{n}_{\mathfrak{g}}(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}))$. Namely, $\mathrm{ad}_{\mathfrak{g}}(a)$ is semisimple for $a \in \mathfrak{a}$, so $\ker(\mathrm{ad}_{\mathfrak{g}}(a)) = \ker((\mathrm{ad}_{\mathfrak{g}}(a))^2)$. Hence

$$n \in \mathfrak{n}_{\mathfrak{g}}(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})) \Longrightarrow [a, n] \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}) \text{ for all } a \in \mathfrak{a} \Rightarrow [a, [a, n]] = 0 \text{ for all } a \in \mathfrak{a}$$
$$\iff n \in \ker((\mathrm{ad}_{\mathfrak{g}}(a))^2) \text{ for all } a \in \mathfrak{a}$$
$$\iff n \in \ker(\mathrm{ad}_{\mathfrak{g}}(a)) \text{ for all } a \in \mathfrak{a} \iff n \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}),$$

and the other inclusion is clear. (Similarly one can show $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}) = \mathfrak{n}_{\mathfrak{g}}(\mathfrak{a})$.)

Step 2. — Keeping the notation of step 1, $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})$ contains the semisimple and nilpotent components (in \mathfrak{g}) of its elements. Namely, for $x \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})$, write $x = x_s + x_n$; then $\mathrm{ad}_{\mathfrak{g}}(x_s) = (\mathrm{ad}_{\mathfrak{g}}(x))_s$ and $\mathrm{ad}_{\mathfrak{g}}(x_n) = (\mathrm{ad}_{\mathfrak{g}}(x))_n$ can be written as polynomials in $\mathrm{ad}_{\mathfrak{g}}(x)$ without constant term, so that both restrict to the zero map on \mathfrak{a} , hence x_s and x_n centralise \mathfrak{a} .

Step 3. — Any semisimple element x_s of $\mathfrak{z} := \mathfrak{z}_{\mathfrak{g}}(\mathfrak{t})$ is contained in \mathfrak{t} . Namely, if it were not, we would have the toral subalgebra $\mathfrak{t} + kx_s \supseteq \mathfrak{t}$.

Step 4. — \mathfrak{z} is nilpotent. Namely, decomposing $x = x_s + x_n$ for $x \in \mathfrak{z}$, by step 3 we have $\mathrm{ad}_{\mathfrak{z}}(x) = \mathrm{ad}_{\mathfrak{z}}(x_s) + \mathrm{ad}_{\mathfrak{z}}(x_n) = \mathrm{ad}_{\mathfrak{z}}(x_n)$ and this, being the restriction of the nilpotent endomorphism $\mathrm{ad}_{\mathfrak{g}}(x_n)$ to \mathfrak{z} , is a nilpotent endomorphism, so that the assertion follows from Engel's theorem.

By steps 1 and 4, \mathfrak{z} is a Cartan subalgebra of \mathfrak{g} . In particular, all its elements are semisimple (cf. [**Bo2**, VII.2.4, Thm 2]), which implies $x = x_s$ in the above decompositions. So by step 3, $\mathfrak{t} = \mathfrak{z}$. (For this last step, some of the references use the non-degeneracy of the Killing form on $\mathfrak{z} \times \mathfrak{z}$, which in turn is shown via scalar extension. Implicitly, this is also used in the proof of our Bourbaki citation.)

REMARK 3.1.6. — From $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{t}) = \mathfrak{t}$ for a maximal toral \mathfrak{t} , it also follows that \mathfrak{t} is a maximal abelian subalgebra of any intermediate algebra $\mathfrak{t} \subseteq \mathfrak{h} \subseteq \mathfrak{g}$ (and maximal toral in \mathfrak{h} for semisimple \mathfrak{h} – only in this case we have defined the concept). For any toral subalgebra \mathfrak{a} , it follows from $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}) = \mathfrak{n}_{\mathfrak{g}}(\mathfrak{a})$ that the centraliser commutes with scalar extension: $(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}))_K = \mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{a}_K)$.

 $\mathbf{21}$

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

Lemma 3.1.7. — If \mathfrak{a} is a

- i. toral subalgebra
- ii. split toral subalgebra
- iii. maximal toral subalgebra

of \mathfrak{g} , then for every extension K|k, \mathfrak{a}_K has the respective property in \mathfrak{g}_K .

Proof

- i. \mathfrak{a}_K is abelian and generated additively by the elements $c \otimes a$ with $c \in K, a \in \mathfrak{a}$. These are obviously semisimple, and the sum of commuting semisimple elements is again semisimple (cf. [**Bo1**, VII.5.8, Cor.]).
- ii. Analogously, scalar multiples and sums of commuting diagonalisable elements are diagonalisable (cf. [Bo1, VII.5.7, Cor.]).
- iii. We have $\mathfrak{n}_{\mathfrak{g}_K}(\mathfrak{a}_K) = (\mathfrak{n}_{\mathfrak{g}}(\mathfrak{a}))_K = \mathfrak{a}_K$ by [**Bo2**, I.3.8] and 3.1.5. Thus there is not even any abelian (K-)Lie algebra \mathfrak{b} with $\mathfrak{a}_K \subsetneq \mathfrak{b} \subseteq \mathfrak{g}_K$.

With that proposition, this (and the converse " \mathfrak{a}_K maximal toral in $\mathfrak{g}_K \Rightarrow \mathfrak{a}$ maximal toral in \mathfrak{g} ") also follows from the corresponding assertion for Cartan subalgebras, [**Bo2**, VII.2.1, Prop. 3].

If \mathfrak{a} is split toral, then by definition, all $\mathrm{ad}_{\mathfrak{g}}(a) \in \mathrm{End}_k(\mathfrak{g})$ for $a \in \mathfrak{a}$ are diagonalisable and commute with each other. It is then a well-known fact of linear algebra that, with the notation of 3.1.1, we have a decomposition of k-vector spaces:

$$\mathfrak{g} = \mathfrak{g}_0 \oplus \bigoplus_{\alpha \in R(\mathfrak{g}, \mathfrak{a})} \mathfrak{g}_{\alpha} \,.$$

We will now collect decisive facts about this weight decomposition that generalise standard facts (cf. [**Bo2**, VIII.2]) about splitting Cartan subalgebras – which, as per remark 3.1.5, are the split maximal toral subalgebras in our setting. In particular we will see that for a maximal split toral \mathfrak{s} , $R(\mathfrak{g}, \mathfrak{s})$ is a (possibly non-reduced) root system. One should beware that the case $\mathfrak{s} = 0$ or equivalently, $R(\mathfrak{g}, \mathfrak{s}) = \emptyset$, can occur, see example 3.2.5. Most of the following discussion (prop. 2.1.8 to 2.1.13) is a reformulation of [**Sel2**, I.1].

PROPOSITION 3.1.8. — Let \mathfrak{a} be a split toral subalgebra.

- i. $R(\mathfrak{g}, \mathfrak{a})$ spans \mathfrak{a}^* as k-vector space.
- ii. For weights α and β with $\alpha + \beta \neq 0$, \mathfrak{g}_{α} and \mathfrak{g}_{β} are orthogonal with respect to κ .
- iii. For every weight α, the restriction of κ to g_α × g_{-α} is non-degenerate. In particular, if α ∈ R(g, a), then -α ∈ R(g, a), and g_α and g_{-α} are dual to each other with respect to κ.
- iv. For all $s, s' \in \mathfrak{a}$ we have $\kappa(s, s') = \sum_{\beta \in R(\mathfrak{g}, \mathfrak{g})} \dim_k(\mathfrak{g}_\beta)\beta(s)\beta(s')$.

MÉMOIRES DE LA SMF 151

 $\mathbf{22}$

Proof

- i. If the k-span of $R(\mathfrak{g},\mathfrak{a})$ were strictly smaller, there would exist an $a \in \mathfrak{a} \setminus \{0\}$ with $\alpha(a) = 0$ for all $\alpha \in R(\mathfrak{g},\mathfrak{a})$. Then the weight decomposition would imply [a, x] = 0 for all $x \in \mathfrak{g}$, that is, $a \in \mathfrak{z}(\mathfrak{g})$; but $\mathfrak{z}(\mathfrak{g})$ is zero because \mathfrak{g} is semisimple.
- ii. If $x \in \mathfrak{g}_{\alpha}, y \in \mathfrak{g}_{\beta}$, applying (2) twice shows that for any weight γ , $\mathrm{ad}_{\mathfrak{g}}(x) \circ \mathrm{ad}_{\mathfrak{g}}(y)$ maps \mathfrak{g}_{γ} to $\mathfrak{g}_{\gamma+\alpha+\beta}$, which implies the assertion by the weight decomposition.
- iii. The first assertion follows from ii. and the non-degeneracy of κ , the others are immediate consequences.
- iv. This is immediate from the weight decomposition.

With the following crucial proposition we get our hands on the coroots.

PROPOSITION 3.1.9. — Let \mathfrak{s} be a maximal split toral subalgebra. For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$, there is a unique element $H_{\alpha} \in \mathfrak{s}$ such that

$$\alpha(H_{\alpha}) = 2$$
 and $H_{\alpha} = [U, V]$

for certain $U \in \mathfrak{g}_{\alpha}, V \in \mathfrak{g}_{-\alpha}$. Moreover, for any $0 \neq X \in \mathfrak{g}_{\alpha}$ there is $0 \neq Y \in \mathfrak{g}_{-\alpha}$ such that $[X,Y] = H_{\alpha}$ (and then necessarily, $[H_{\alpha},X] = 2X$ and $[H_{\alpha},Y] = -2Y$).

Proof. — Let $0 \neq X \in \mathfrak{g}_{\alpha}$. By 3.1.2, X is nilpotent. By the Jacobson-Morozov theorem, there are $y, z \in \mathfrak{g}$ such that [z, X] = 2X, [z, y] = -2y and [X, y] = z. Decomposing

$$y = \sum_{eta \in R(\mathfrak{g}, \mathfrak{a}) \cup \{0\}} y_{eta} \quad ext{and} \quad z = \sum_{eta \in R(\mathfrak{g}, \mathfrak{a}) \cup \{0\}} z_{eta}$$

with respect to the weight decomposition, we see from (2) and $X \in \mathfrak{g}_{\alpha}$ that $[z_0, X] = 2X$ (in particular, $z_0 \neq 0$), that $[z_{\beta}, X] = 0$ for $\beta \neq 0$, and that $[X, y_{-\alpha}] = z_0$. Applying [**Bo2**, VIII.11.2, Lemma 6] to $h = z_0$ and x = X gives a "better" $\tilde{y} \in \mathfrak{g}$ with $[z_0, \tilde{y}] = -2\tilde{y}$ and $[X, \tilde{y}] = z_0$. Again decomposing $\tilde{y} = \sum_{\beta \in R(\mathfrak{g}, \mathfrak{g}) \cup \{0\}} \tilde{y}_{\beta}$ and using (2) we get $[X, \tilde{y}_{-\alpha}] = z_0$ (in particular, $\tilde{y}_{-\alpha} \neq 0$), and $[z_0, \tilde{y}_{\beta}] = -2\tilde{y}_{\beta}$ for all weights β , specifically for $\beta = -\alpha$.

We now claim that $H_{\alpha} := z_0$ has the properties asserted in the second sentence. (Our construction then shows the assertion of the third sentence for the above X with $Y := \tilde{y}_{-\alpha}$.)

The above construction already shows $\alpha(H_{\alpha}) = 2$, and that H_{α} is the Lie product of an element of \mathfrak{g}_{α} with one of $\mathfrak{g}_{-\alpha}$. Further, H_{α} is contained in \mathfrak{g}_{0} which means that it commutes with all elements of \mathfrak{s} . Finally, it being the semisimple part of an \mathfrak{sl}_{2} -triple and the representation theory of \mathfrak{sl}_{2} shows that all eigenvalues of $\mathfrak{ad}_{\mathfrak{g}}(H_{\alpha})$ are integers, so it certainly is diagonalisable. Hence with \mathfrak{s} also $\mathfrak{s} + kH_{\alpha}$ is a split toral subalgebra of \mathfrak{g} , and by maximality, we must have $H_{\alpha} \in \mathfrak{s}$. The uniqueness will follow from the following lemma by noting that for $H, H' \in \mathfrak{a}$ with $\beta(H) = \beta(H')$ for all roots β , we have that H - H' is central in \mathfrak{g} , hence 0. LEMMA 3.1.10. — For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$, let $U \in \mathfrak{g}_{\alpha}, V \in \mathfrak{g}_{-\alpha}$ be such that H := [U, V] is in \mathfrak{s} and satisfies $\alpha(H) = 2$. Let β be any weight of \mathfrak{s} in \mathfrak{g} . Then there are $q, r \in \mathbb{Z}_{\geq 0}$ such that for $i \in \mathbb{Z}$, we have $\mathfrak{g}_{\beta+i\alpha} \neq \{0\}$ if and only if $-r \leq i \leq q$; further, $\beta(H) = r - q$.

Proof. — U, H, V are an \mathfrak{sl}_2 -triple in \mathfrak{g} , with respect to which we consider the \mathfrak{sl}_2 -representation $E := \sum_{i \in \mathbb{Z}} \mathfrak{g}_{\beta+i\alpha}$. Let q (resp. r) be maximal in \mathbb{Z} with the property $\mathfrak{g}_{\beta+q\alpha} \neq \{0\}$ (resp. $\mathfrak{g}_{\beta-r\alpha} \neq \{0\}$). The eigenspaces of H in E are the non-zero $E_{\beta(H)+2i} = \mathfrak{g}_{\beta+i\alpha}$ for $i \in \mathbb{Z}$, so by 2.1.7.ii we have

$$\beta(H) + 2q = -(\beta(H) - 2r),$$

hence $\beta(H) = r - q$; then by part i of the same lemma, the possible eigenspaces above are indeed non-zero for $-r \leq i \leq q$.

These H_{α} allow us to derive further results:

PROPOSITION 3.1.11. — Let \mathfrak{s} be a maximal split toral subalgebra, and for $\alpha \in R(\mathfrak{g}, \mathfrak{s})$, let H_{α} be as above.

- i. For $\alpha, \beta \in R(\mathfrak{g}, \mathfrak{s})$, we have $\beta(H_{\alpha}) \in \mathbb{Z}$.
- ii. For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$, we have $\kappa(H_{\alpha}, H_{\alpha}) \in \mathbb{Z}_{>0}$.
- iii. For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$ and $s \in \mathfrak{s}$, we have:

$$\alpha(s) = 2 \frac{\kappa(H_{\alpha}, s)}{\kappa(H_{\alpha}, H_{\alpha})}$$

- iv. If $\alpha_1, \ldots, \alpha_n \in R(\mathfrak{g}, \mathfrak{s}) \subset \mathfrak{s}^*$ are k-linearly independent, then $H_{\alpha_1}, \ldots, H_{\alpha_r} \in \mathfrak{s}$ are k-linearly independent. The H_{α} for $\alpha \in R(\mathfrak{g}, \mathfrak{s})$ span \mathfrak{s} over k.
- v. The restriction of κ to $\mathfrak{s} \times \mathfrak{s}$ is non-degenerate.

Proof

- i. Restatement of part of the previous lemma.
- ii. We have $\kappa(H_{\alpha}, H_{\alpha}) = \sum_{\beta \in R(\mathfrak{g}, \mathfrak{s})} \dim_k(\mathfrak{g}_{\beta})(\beta(H_{\alpha}))^2$, so the assertion follows from i. and $\alpha(H_{\alpha}) = 2 > 0$.
- iii. Let $\alpha \in R(\mathfrak{g}, \mathfrak{s})$. Setting $s' := s \frac{1}{2}\alpha(s)H_{\alpha} \in \mathfrak{s}$ (and using ii.), the claim is equivalent to: $\kappa(H_{\alpha}, s') = 0$.

To show this, choose an \mathfrak{sl}_2 -triple X, H_α, Y as in 3.1.9. There is a subset $B \subseteq R(\mathfrak{g}, \mathfrak{s})$ such that the roots can be partitioned as

$$R(\mathfrak{g},\mathfrak{s})=\coprod_{\beta\in B}\{\beta+i\alpha:i\in\mathbb{Z},\beta+i\alpha\in R(\mathfrak{g},\mathfrak{s})\};$$

the spaces $\mathfrak{g}(\beta, \alpha) := \sum_{i \in \mathbb{Z}, \beta+i\alpha \in R(\mathfrak{g}, \mathfrak{s})} \mathfrak{g}_{\beta+i\alpha}$ are stable under $\mathrm{ad}(s')$ as well as under the adjoint action of the \mathfrak{sl}_2 -triple. Because of the partition, $\kappa(H_\alpha, s')$ is the sum of the traces of $\mathrm{ad}_{\mathfrak{g}(\beta,\alpha)}(H_\alpha) \circ \mathrm{ad}_{\mathfrak{g}(\beta,\alpha)}(s')$ for $\beta \in B$. But because $\alpha(s') = 0$, such a trace is just $\beta(s')$ multiplied with the trace of $\mathrm{ad}_{\mathfrak{g}(\beta,\alpha)}(H_\alpha)$. Finally, this

last trace is zero because the endomorphism $\operatorname{ad}_{\mathfrak{g}(\beta,\alpha)}(H_{\alpha})$ is the commutator of the endomorphisms $\operatorname{ad}_{\mathfrak{g}(\beta,\alpha)}(X)$ and $\operatorname{ad}_{\mathfrak{g}(\beta,\alpha)}(Y)$.

iv. Assume we have $\lambda_1, \ldots, \lambda_n \in k$ with $\sum_{i=1}^n \lambda_i H_{\alpha_i} = 0$. Then for all $s \in \mathfrak{s}$, we have

$$0 = \kappa \left(\sum_{i=1}^{n} \lambda_i H_{\alpha_i}, s\right) \stackrel{iii.}{=} \frac{1}{2} \sum_{i=1}^{n} \lambda_i \cdot \kappa(H_{\alpha_i}, H_{\alpha_i}) \cdot \alpha_i(s),$$

so by assumption and part ii, all λ_i must be zero. The second assertion follows from the first one and 3.1.8.i.

v. This follows from part iii, as $\alpha(s) = 0$ for all $\alpha \in R(\mathfrak{g}, \mathfrak{s})$ implies s = 0 for $s \in \mathfrak{s}$.

Using the canonical identification of \mathfrak{s} with its bidual $(\mathfrak{s}^*)^*$, we set $\check{\alpha} := H_{\alpha}$ and see with the remark after (1) that $s_{\alpha,\check{\alpha}}$ is a reflection.

THEOREM 3.1.12. — Let \mathfrak{s} be a maximal split toral subalgebra of \mathfrak{g} . Then $R(\mathfrak{g}, \mathfrak{s})$ is a root system (in \mathfrak{s}^*).

Proof. — Finiteness and $0 \notin R$ are obvious, so (RS1) follows from 3.1.8.i. For α, β in $R(\mathfrak{g}, \mathfrak{s})$, Lemma 3.1.10 gives us non-negative q, r such that

$$s_{\alpha,\check{\alpha}}(\beta) = \beta - \beta(H_{\alpha})\alpha = \beta + (q-r)\alpha,$$

and that this is a weight. It is non-zero because as a reflection, $s_{\alpha,\check{\alpha}}$ is injective, so (RS2) follows. Finally, (RS3) is given by 3.1.11.i.

Let us point out three differences to the "split" situation in [**Bo2**, VIII.2.2] (for all of them to occur, see example 3.2.9):

- $\triangleright R(\mathfrak{g},\mathfrak{s})$ is not reduced in general: there might be $\alpha \in R(\mathfrak{g},\mathfrak{s})$ with $2\alpha \in R(\mathfrak{g},\mathfrak{s})$.
- \triangleright The root spaces \mathfrak{g}_{α} can have dimension > 1.

▷ The space $[\mathfrak{g}_{\alpha}, \mathfrak{g}_{-\alpha}]$ can also have dimension > 1, in particular might be not generated by the element H_{α} (but its intersection with \mathfrak{s} is, as follows from later results).

Obviously, the third point can only occur if the second one does. The next proposition shows that the first two points can only occur together, and "why" they do not in the case of a split maximal toral \mathfrak{s} .

Proposition 3.1.13

- i. For $\alpha, \beta \in R(\mathfrak{g}, \mathfrak{s})$ with $\beta(H_{\alpha}) > 0$ and for every $0 \neq X_{\alpha} \in \mathfrak{g}_{\alpha}$, we have $[X_{\alpha}, \mathfrak{g}_{\beta-\alpha}] = \mathfrak{g}_{\beta}$.
- ii. For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$ we have $\dim_k \mathfrak{g}_{2\alpha} < \dim_k \mathfrak{g}_{\alpha}$.
- iii. If $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{s}$ (equivalently, \mathfrak{s} is maximal toral), $\dim_k \mathfrak{g}_{\alpha} = 1$ for all $\alpha \in R(\mathfrak{g}, \mathfrak{s})$. In particular, $[\mathfrak{g}_{\alpha}, \mathfrak{g}_{-\alpha}] = k \cdot H_{\alpha}$, and $R(\mathfrak{g}, \mathfrak{s})$ is reduced.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

Proof

26

- i. By 3.1.9, X_{α} is part of an \mathfrak{sl}_2 -triple $X_{\alpha}, H_{\alpha}, Y_{\alpha}$. Set $\lambda := \beta(H_{\alpha})$. In the finite dimensional \mathfrak{sl}_2 -representation $E := \sum_{i \in \mathbb{Z}} \mathfrak{g}_{\beta+i\alpha}$ again we have $E_{\lambda+2i} = \mathfrak{g}_{\beta+i\alpha}$ for $i \in \mathbb{Z}$; now apply 2.1.7.iii.
- ii. Trivial if $2\alpha \notin R(\mathfrak{g}, \mathfrak{s})$. Else, part i with $\beta = 2\alpha$ shows that for $0 \neq X_{\alpha} \in \mathfrak{g}_{\alpha}$, $\operatorname{ad}(X_{\alpha})$ induces a k-linear surjection $\mathfrak{g}_{\alpha} \twoheadrightarrow \mathfrak{g}_{2\alpha}$, whose kernel contains $0 \neq X_{\alpha}$.

iii. Setting
$$\beta = \alpha$$
 in part i, we get $\mathfrak{g}_{\alpha} = [\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}), X_{\alpha}] = [\mathfrak{s}, X_{\alpha}] = \alpha(\mathfrak{s})X_{\alpha}$.

The root system $R(\mathfrak{g}, \mathfrak{s})$ abstractly comes equipped with its Weyl group, which is identified with the Weyl group of its inverse root system formed by the $H_{\alpha} = \check{\alpha}$ (see the remarks after Proposition 2 in [**Bo2**, VI.1.1]). One can check with 3.1.11 that on the inverse root system, the Killing form κ is non-degenerate and invariant under the action of the Weyl group. We will find a realisation of the Weyl group as a subquotient of $\operatorname{Aut}_e(\mathfrak{g},\mathfrak{s})$ (see definition 2.1.9) in the spirit of [**Bo2**, VIII.5.2]. To start with, there is a well-defined group homomorphism, the contragredient of the restriction to \mathfrak{s} ,

$$\varepsilon_{\mathfrak{s}} : \operatorname{Aut}_{e}(\mathfrak{g}, \mathfrak{s}) \longrightarrow A(R(\mathfrak{g}, \mathfrak{s})), \quad f \longmapsto [\alpha \mapsto \alpha \circ f_{|\mathfrak{s}}^{-1} \text{ for } \alpha \in R(\mathfrak{g}, \mathfrak{s})]$$

whose kernel obviously is $\{f \in \operatorname{Aut}_e(\mathfrak{g}, \mathfrak{s}) : f_{|\mathfrak{s}} = \operatorname{id}_{\mathfrak{s}}\}.$

PROPOSITION 3.1.14. — For $\alpha \in R(\mathfrak{g}, \mathfrak{s})$, choose an \mathfrak{sl}_2 -triple (X, H_α, Y) as in 3.1.9 and set

$$g_{\alpha} := \exp(\operatorname{ad} Y) \circ \exp(\operatorname{ad}(-X)) \circ \exp(\operatorname{ad} Y) \in \operatorname{Aut}_{e}(\mathfrak{g}).$$

Then $g_{\alpha}(H_{\alpha}) = -H_{\alpha}$, and $g_{\alpha} = \text{id}$ on the orthogonal complement of H_{α} in \mathfrak{s} (with respect to κ).

Proof. — For the second assertion, $s \perp H_{\alpha}$ for $s \in \mathfrak{s}$ by 3.1.11.iii means $\alpha(s) = 0$, hence $g_{\alpha}(s) = s$ because $X \in \mathfrak{g}_{\alpha}, Y \in \mathfrak{g}_{-\alpha}$. For the first assertion, calculate:

$$(\exp(\operatorname{ad} Y) \circ \exp(\operatorname{ad}(-X)) \circ \exp(\operatorname{ad} Y))(H_{\alpha})$$

= $(\exp(\operatorname{ad} Y) \circ \exp(\operatorname{ad}(-X)))(H_{\alpha} + 2Y)$
= $\exp(\operatorname{ad} Y)(-H_{\alpha} + 2Y + 2X - 2H_{\alpha} + \frac{1}{2}(-4X))$
= $\exp(\operatorname{ad} Y)(-H_{\alpha} + 2Y) = -H_{\alpha} + 2Y + (-2Y) = -H_{\alpha}$

COROLLARY 3.1.15. — For choices as above, let G be the subgroup of $\operatorname{Aut}_e(\mathfrak{g},\mathfrak{s})$ generated by the g_{α} . Then $\varepsilon_{\mathfrak{s}}(G) = W(R(\mathfrak{g},\mathfrak{s}))$.

We will later improve this result, see 3.1.47 and the subsection around it.

We end this subsection with a conjugacy result on maximal split toral subalgebras in \mathfrak{g} . It will be needed "only" to show that the invariants we will define are independent from various choices. The analogous conjugacy statement for maximal k-split tori in reductive groups is proven in [**BT1**, 4.21].

THEOREM 3.1.16. — Let $\mathfrak{s}, \mathfrak{s}'$ be maximal split toral subalgebras of \mathfrak{g} . Then there is an elementary automorphism $g \in \operatorname{Aut}_e(\mathfrak{g})$ such that $g(\mathfrak{s}') = \mathfrak{s}$.

Proof. — The only statement and proof of (a slightly more general version of) this in the literature seems to be [Sel2, I.3]. We give a sketch of this proof. Fix a system of positive roots P in $R(\mathfrak{g}, \mathfrak{s})$ and set

$$\mathfrak{N}(\mathfrak{s}) := igoplus_{\lambda \in P} \mathfrak{g}_{\lambda}, \qquad \mathfrak{P}(\mathfrak{s}) := \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \oplus \mathfrak{N}(\mathfrak{s}).$$

Step 1. — For every semisimple $x_s \in \mathfrak{P}(\mathfrak{s})$, there is an $h \in \operatorname{Aut}_e(\mathfrak{g})$ with $h(x_s) \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$.

Namely, write a general element $x \in \mathfrak{P}(\mathfrak{s})$ as $u+n_0$ with $u \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}), n_0 \in \mathfrak{N}(\mathfrak{s})$. Using that the endomorphism $\mathrm{ad}_{\mathfrak{N}(\mathfrak{s})}(u)$ is semisimple (by 3.1.29), one defines n_i and $n_0^{(i)}$ in $\mathfrak{N}(\mathfrak{s})$ inductively so that

$$\exp(\operatorname{ad} n_i) \circ \cdots \circ \exp(\operatorname{ad} n_1)(x) = u + n_0^{(i)} \mod \mathcal{C}^i \mathfrak{N}(\mathfrak{s}),$$

where the $n_0^{(i)}$ are in the kernel of $\operatorname{ad}_{\mathfrak{N}(\mathfrak{s})}(u)$. Using nilpotence of $\mathfrak{N}(\mathfrak{s})$, after finitely many steps one has u and $n_0^{(r)}$ being the semisimple and nilpotent part of $\exp(\operatorname{ad} n_r) \circ \cdots \circ \exp(\operatorname{ad} n_1)(x)$; in particular, if $x = x_s$ is semisimple, it is mapped to u.

Step 2. — If $\mathfrak{s}' \subset \mathfrak{P}(\mathfrak{s})$, there is $h \in \operatorname{Aut}_e(\mathfrak{g})$ with $h(\mathfrak{s}') = \mathfrak{s}$.⁽³⁾

Namely, pick a "generic" element $x \in \mathfrak{s}'$, meaning $\mathfrak{z}_{\mathfrak{g}}(x) = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}')$, and apply step 1. One gets $\mathfrak{s} \subseteq \mathfrak{z}_{\mathfrak{g}}(h(x)) = \mathfrak{z}_{\mathfrak{g}}(h(\mathfrak{s}'))$. Using maximalities, one infers that $\mathfrak{s} = h(\mathfrak{s}')$.

Step 3. — One checks that $\mathfrak{N}(\mathfrak{s})$ is a maximal nil (= consisting of nilpotent elements) subalgebra of \mathfrak{g} , and that $\mathfrak{P}(\mathfrak{s})$ is the normaliser of $\mathfrak{N}(\mathfrak{s})$ in \mathfrak{g} . One further checks that for any two maximal nil subalgebras $\mathfrak{N}_1, \mathfrak{N}_2$ of \mathfrak{g} with respective normalisers $\mathfrak{P}_1, \mathfrak{P}_2$, one has (i = 1, 2):

$$\mathfrak{P}_i = (\mathfrak{P}_1 \cap \mathfrak{P}_2) + \mathfrak{N}_i$$

Step 4. — Let \mathfrak{N} be a maximal nil subalgebra of \mathfrak{g} with normaliser \mathfrak{P} . Then there is $h \in \operatorname{Aut}_e(\mathfrak{g})$ with $h^{-1}(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})) \subseteq \mathfrak{P}$.

Namely, there is a reductive subalgebra \mathfrak{M} of \mathfrak{g} consisting of semisimple elements and such that $\mathfrak{P} = \mathfrak{M} \oplus \mathfrak{N}$ and $\mathfrak{P}(\mathfrak{s}) = \mathfrak{M} \oplus \mathfrak{N}(\mathfrak{s})$. (To construct such an algebra inside $\mathfrak{P}(\mathfrak{s}) \cap \mathfrak{P}$ with step 3 is the most technical part of the proof, using the embedding $\mathfrak{g} \hookrightarrow \mathfrak{gl}(\mathfrak{g})$ and facts about Lie algebras of endomorphisms.) Now first one conjugates the centre $\mathfrak{z}(\mathfrak{M})$ into $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$, applying the technique of step 2. (Here again there is a

^{2.} This is in fact a minimal parabolic subalgebra of $\mathfrak{g}.$

^{3.} This is an analogue of Corollaire (ii) of [Bo2, VIII.3, Prop. 9] in our situation.

technical digression to construct a suitably "generic" element.) Secondly, one looks at the algebra

 $\mathfrak{C} := \mathfrak{z}_{\mathfrak{P}(\mathfrak{s})}(\mathfrak{z}(\mathfrak{M}))$

and shows that it is reductive with semisimple part $\mathcal{D}(\mathfrak{C} \cap \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})) \subseteq \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$. Since the semisimple $\mathcal{D}\mathfrak{M}$ is contained in \mathfrak{C} , by a part of the Levi-Malcev theorem (cf. [**Bo2**, I.6.8, Cor. 1]) there is an elementary automorphism (of \mathfrak{C} , extending to \mathfrak{g}) hwhich conjugates $\mathcal{D}\mathfrak{M}$ into that semisimple part, and stabilises $\mathfrak{z}(\mathfrak{M})$. So we have conjugated all of \mathfrak{M} into $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$. With a dimension argument, one shows that in fact $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = h(\mathfrak{M}) \subseteq h(\mathfrak{P})$.

We now define $\mathfrak{N}(\mathfrak{s}')$ and $\mathfrak{P}(\mathfrak{s}')$ analogously, with respect to some system of positive roots $P' \subset R(\mathfrak{g}, \mathfrak{s}')$. By steps 3 and 4, we can assume without loss of generality that $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \subseteq \mathfrak{P}(\mathfrak{s}')$.

Step 5. — Under this assumption, the roots

$$P(P, P') := \left\{ \alpha \in R(\mathfrak{g}, \mathfrak{s}) : \mathfrak{g}_{\alpha} \cap \mathfrak{P}(\mathfrak{s}') \neq \{0\} \right\}$$

form a system of positive roots, i.e. there is $w \in W(R(\mathfrak{g}, \mathfrak{s}))$ with P = wP(P, P'), and we have:

$$\mathfrak{P}(\mathfrak{s}') = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \oplus \bigoplus_{lpha \in P(P,P')} \mathfrak{g}_{lpha}$$

Namely, for this one argues cleverly with (2), 3.1.13.i and \mathfrak{sl}_2 -triples.

Step 6. — As in 3.1.14 and its corollary, there is an $f \in Aut_e(\mathfrak{g}, \mathfrak{s})$ that induces the w in step 5. It follows:

$$f(\mathfrak{P}(\mathfrak{s}')) = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \oplus \bigoplus_{lpha \in P} \mathfrak{g}_{lpha} = \mathfrak{P}(\mathfrak{s})$$

So without loss of generality, we are in the situation of step 2, and therefore finished. $\hfill\square$

COROLLARY 3.1.17. — The integer rank $(R(\mathfrak{g},\mathfrak{s})) := \dim_k(\mathfrak{s})$ is independent from the choice of \mathfrak{s} . We call it the k-rank of \mathfrak{g} .

3.1.2. Galois actions and the anisotropic kernel. — Let \mathfrak{a} be a toral subalgebra of \mathfrak{g} . Choose a k-basis a_1, \ldots, a_n of \mathfrak{a} ; by field theory, there exists an extension K|k such that all the characteristic polynomials of the $\mathrm{ad}_{\mathfrak{g}}(a_i)$ split into linear factors over K. This implies that \mathfrak{a}_K is a *split* toral subalgebra of \mathfrak{g}_K .

DEFINITION 3.1.18 (Splitting extensions). — An extension K|k such that \mathfrak{a}_K is split toral in \mathfrak{g}_K is called a *splitting extension*, and K a *splitting field*, for the toral subalgebra \mathfrak{a} .
It is clear from the preceding construction that a splitting extension can be chosen to be finite and Galois over k. It is also clear that every extension of a splitting field is again a splitting field.

After we have come this far using not much scalar extension, we will now unleash the full power of this tool. To begin with, let $\mathfrak{a} \subseteq \mathfrak{t}$ be any inclusion of toral subalgebras of \mathfrak{g} such that \mathfrak{a} is split over k, and let K|k be a splitting extension for \mathfrak{t} . We have the vector space decomposition:

(3)
$$\mathfrak{g} = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}) \oplus \bigoplus_{\lambda \in R(\mathfrak{g},\mathfrak{a})} \mathfrak{g}_{\lambda}$$

Tensoring this gives:

(4)
$$\mathfrak{g}_K = (\mathfrak{z}_\mathfrak{g}(\mathfrak{a}))_K \oplus \bigoplus_{\lambda \in R(\mathfrak{g},\mathfrak{a})} (\mathfrak{g}_\lambda)_K$$

On the other hand, \mathfrak{a}_K is split toral in \mathfrak{g}_K , so:

(5)
$$\mathfrak{g}_{K} = \mathfrak{z}_{\mathfrak{g}_{K}}(\mathfrak{a}_{K}) \oplus \bigoplus_{\lambda \in R(\mathfrak{g}_{K},\mathfrak{a}_{K})} (\mathfrak{g}_{K})_{\lambda}$$

And finally, because \mathfrak{t}_K is split toral in \mathfrak{g}_K :

(6)
$$\mathfrak{g}_K = \mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{t}_K) \oplus \bigoplus_{\alpha \in R(\mathfrak{g}_K, \mathfrak{t}_K)} (\mathfrak{g}_K)_\alpha$$

These decompositions are strongly related. Recall that for any finite-dimensional vector space V over k, we have a canonical inclusion

$$V \hookrightarrow K \otimes_k V, \quad v \longmapsto 1 \otimes v,$$

and a canonical isomorphism

(7)
$$K \otimes_k (V^*) \cong (K \otimes_k V)^*,$$

see [**Bo1**, II.6.4], where by abuse of notation the * denotes k-dual on the left, but K-dual on the right. Using this, the following lemma establishes identity between (4) and (5).

LEMMA 3.1.19. — The composed map

$$\mathfrak{a}^* \hookrightarrow (\mathfrak{a}^*)_K \cong (\mathfrak{a}_K)^*$$

induces a bijection $R(\mathfrak{g},\mathfrak{a}) \simeq R(\mathfrak{g}_K,\mathfrak{a}_K)$. We have $(\mathfrak{g}_\lambda)_K = (\mathfrak{g}_K)_\lambda$ for every λ in $R(\mathfrak{g},\mathfrak{a})$ using this identification. Further, we can identify $(\mathfrak{z}_\mathfrak{g}(\mathfrak{a}))_K$ and $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{a}_K)$ ("case $\lambda = 0$ ").

Proof. — We have already remarked $(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}))_K = \mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{a}_K)$ in 3.1.6. Calling $1 \otimes \lambda$ the image of $\lambda \in R(\mathfrak{g}, \mathfrak{a})$ under the composed map, clearly $(\mathfrak{g}_{\lambda})_K \subseteq (\mathfrak{g}_K)_{1 \otimes \lambda}$. Comparing (4) and (5) and a consideration of dimensions shows that all these inclusions have to be equalities, which also shows that the map is surjective on the roots.

The inclusion $\mathfrak{a}_K \subseteq \mathfrak{t}_K$ induces the restriction map

$$\rho: (\mathfrak{t}_K)^* \longrightarrow (\mathfrak{a}_K)^*.$$

This connects (5) with (6):

30

PROPOSITION 3.1.20. — We have

$$\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})_{K} = \mathfrak{z}_{\mathfrak{g}_{K}}(\mathfrak{t}_{K}) \oplus \bigoplus_{\substack{lpha \in R(\mathfrak{g}_{K},\mathfrak{t}_{K}) \\
ho(lpha) = 0}} (\mathfrak{g}_{K})_{lpha}$$

and for every $\lambda \in R(\mathfrak{g}_K, \mathfrak{a}_K)$,

$$(\mathfrak{g}_{\lambda})_{K} = \bigoplus_{\substack{\alpha \in R(\mathfrak{g}_{K}, \mathfrak{t}_{K})\\\rho(\alpha) = \lambda}} (\mathfrak{g}_{K})_{\alpha}.$$

The map ρ restricts to a surjection

$$\rho: R(\mathfrak{g}_K, \mathfrak{t}_K) \cup \{0\} \longrightarrow R(\mathfrak{g}_K, \mathfrak{a}_K) \cup \{0\}.$$

Proof. — Using the identification of the foregoing lemma, the " \supseteq " in the asserted equations is clear. As before, comparing (5) and (6) and a consideration of dimensions proves all claims.

COROLLARY 3.1.21. — For any toral subalgebra $\mathfrak{a} \subseteq \mathfrak{g}$, the centraliser $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})$ is a reductive Lie algebra.

Proof. — By 2.1.4, being reductive is stable under scalar extension; by 3.1.6, for a toral subalgebra, the centraliser commutes with scalar extension; thus we can w.l.o.g. assume \mathfrak{a} to be split. Choose a maximal toral subalgebra \mathfrak{t} containing \mathfrak{a} , and a splitting field K for \mathfrak{t} . Again it suffices to check that $(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a}))_K$ is reductive. By 3.1.20 and 3.1.5,

$$\mathfrak{z}_{\mathfrak{g}}(\mathfrak{a})_{K} = \mathfrak{t}_{K} \oplus \bigoplus_{\substack{\alpha \in R(\mathfrak{g}_{K}, \mathfrak{t}_{K}) \\ \rho(\alpha) = 0}} (\mathfrak{g}_{K})_{\alpha}$$

and this is a reductive Lie algebra by [Bo2, VIII.3.1, Prop. 2].

We now fix a maximal split toral subalgebra \mathfrak{s} , a maximal toral subalgebra \mathfrak{t} containing \mathfrak{s} (again the existence of such a \mathfrak{t} is clear by finite dimension), and a finite Galois splitting extension K|k for \mathfrak{t} . We set

$$R := R(\mathfrak{g}_K, \mathfrak{t}_K), \quad \overline{R} := R(\mathfrak{g}_K, \mathfrak{s}_K), \quad R_0 := \{\alpha \in R : \rho(\alpha) = 0\}.$$

By 3.1.12 and 3.1.19, R and \overline{R} are root systems. The elements of \overline{R} are sometimes called *restricted roots*, or (k)-rational roots. By 3.1.5, for R we have the classical theory of [**Bo2**, VIII]. Finally, if R_0 is not empty, it also is a root system (in the vector space it generates): we obviously have $-\alpha \in R_0$ and $\alpha + \beta \in R \Rightarrow \alpha + \beta \in R_0$ for $\alpha, \beta \in R_0$

and conclude by [**Bo2**, VI.1.7, Prop. 23]. We will have a closer look at R_0 later: see the Subsection 3.1.2.1, in particular 3.1.31.

In the theory of algebraic and Lie groups, the roots and weights are constructed from some module of characters usually denoted by variants of the letter X. Here we just *define* X to be the additive group (= \mathbb{Z} -module) in (\mathfrak{t}_K)^{*} generated by R. By 3.1.8.i, it is a lattice, i.e. the natural map $K \otimes_{\mathbb{Z}} X \to (\mathfrak{t}_K)^*$ is bijective.⁽⁴⁾

We will now construct actions of the absolute Galois group \mathcal{G} on several of the above objects, in particular on roots. Since they will all be induced by the action of \mathcal{G} on K, we can and will see them as actions of $\Gamma := \operatorname{Gal}(K|k)$. We start with the action of Γ on $\mathfrak{g}_K = K \otimes_k \mathfrak{g}$ (more general, every \mathfrak{a}_K for a subalgebra \mathfrak{a} of \mathfrak{g}) given by operation on the first factor. Then the action of $\sigma \in \Gamma$ is σ -linear, i.e.

$$\sigma(ax+y) = \sigma(a)\sigma(x) + \sigma(y) \text{ for all } a \in K, x, y \in \mathfrak{g}_K,$$

and satisfies:

$$\left[\sigma(x), \sigma(y)\right] = \sigma\left(\left[x, y\right]\right) \text{ for all } x, y \in \mathfrak{g}_{K}$$

In particular, σ acts as automorphism of $R_{K|k} \mathfrak{g}_K$. Further, because σ is an automorphism of K, for a K-subspace U of \mathfrak{g}_K , $\sigma(U)$ is a K-subspace of the same K-dimension. The fixed set $(\mathfrak{g}_K)^{\Gamma}$ is \mathfrak{g} , and we define the following, k-linear, kind of a trace map:

$$\mathfrak{g}_K \longrightarrow \mathfrak{g}, \quad x \longmapsto \widehat{x} := \sum_{\sigma \in \Gamma} \sigma(x)$$

It is surjective because for $x \in \mathfrak{g}$, $\hat{x} = |\Gamma| \cdot x$.

We also have Γ -actions on $(\mathfrak{t}_K)^*$ and $(\mathfrak{s}_K)^*$ per (7), which we will for the moment write in the form $\sigma\lambda$. Choosing e.g. a basis t_1, \ldots, t_n of \mathfrak{t} and the corresponding dual basis u_1, \ldots, u_n of \mathfrak{t}^* , the isomorphism (7) writes $\lambda \in (\mathfrak{t}_K)^*$ as

$$\sum_{i=1}^n \lambda(1 \otimes t_i) \otimes u_i$$

so that, for $\sigma \in \Gamma$, $\sigma \lambda$ is defined to be the K-linear map sending $1 \otimes t_i \in \mathfrak{t}_K$ to $\sigma(\lambda(1 \otimes t_i))$. Abusing notation, this is the map $\sigma \circ \lambda \circ \sigma^{-1}$, where the left σ is the original k-automorphism of K and the right one is the automorphism of $R_{K|k}\mathfrak{t}_K$.

Now for any weight α of \mathfrak{t}_K in \mathfrak{g}_K , $\sigma((\mathfrak{g}_K)_\alpha)$ is a K-subspace of \mathfrak{g}_K , and for $x \in \sigma((\mathfrak{g}_K)_\alpha)$, say $x = \sigma(y)$ with $y \in (\mathfrak{g}_K)_\alpha$, we have

$$[a \otimes t_i, x] = [\sigma(\sigma^{-1}(a) \otimes t_i), \sigma(y)]$$

= $\sigma([\sigma^{-1}(a) \otimes t_i, y]) = \sigma(\alpha(\sigma^{-1}(a) \otimes t_i) \cdot y)$
= $(\sigma(\alpha)(a \otimes t_i) \cdot x)$

for $a \in K$ and t_i as above. Using that the $\sigma \in \Gamma$ act as automorphisms, we see:

^{4.} This root lattice is often called Q in the group case; it sits inside the weight lattice P. Roughly speaking, with X := Q we define Lie algebras to be "of adjoint type"; less roughly said, their automorphism groups really are.

Lemma 3.1.22

- i. For σ ∈ Γ and a weight α of t_K in g_K, ^σα also is a weight of t_K in g_K. The corresponding weight space is σ((g_K)_α), whose K-dimension is the same as that of (g_K)_α, namely, 1. The Galois action stabilises R, a fortiori restricts to X, turning it into a Γ-module.
- ii. On \overline{R} , the Galois action is trivial.
- iii. The map ρ is equivariant with respect to the Galois actions. That is (by ii.), $\rho(\sigma(\alpha)) = \rho(\alpha)$ for all $\sigma \in \Gamma, \alpha \in R$.

Proof. — Part i sums up the foregoing discussion. For part ii, analogous considerations apply, and additionally we can use that by 3.1.19, for a k-basis s_1, \ldots, s_m of \mathfrak{s} , we have $\lambda(1 \otimes s_i) \in k = K^{\Gamma}$ for all $1 \leq i \leq m$ and $\lambda \in \overline{R}$. Part iii follows from the canonicity of the isomorphism (7).

REMARK 3.1.23. — We will later define another Galois action on the roots as a certain twisting of the above action – see 3.1.41.

From now on, if not noted otherwise, we will consider R as a root system inside the \mathbb{R} -vector space $V := \mathbb{R} \otimes_{\mathbb{Z}} X$. The automorphism group A(R) and the Weyl group W(R) are thus subgroups of $\operatorname{Aut}_{\mathbb{R}}(V)$. The Galois action above is identified with a homomorphism $\Gamma \to A(R)$, and we will write σ_A for the image of $\sigma \in \Gamma$ under this homomorphism (so $\sigma_{\chi} = \sigma_A(\chi)$ for $\chi \in X$). This enables us to define a Galois action on the Weyl group:

LEMMA 3.1.24. — For $w \in W(R)$, set $\sigma w := \sigma_A \circ w \circ \sigma_A^{-1}$, where composition is in A(R). This defines an action of Γ on the Weyl group.

Proof. — One has ${}^{\sigma}w \in W(R)$ because W(R) is normal in A(R). Verification of the rest is straightforward.

To see more structure in X and R, we go back to our Lie algebras over K. The natural pairing between \mathfrak{t}_K^* and \mathfrak{t}_K – for which we have $({}^{\sigma}\ell)(\sigma(x)) = \sigma(\ell(x))$ for $\ell \in \mathfrak{t}_K^*$, $x \in \mathfrak{t}_K$ and $\sigma \in \Gamma$ – induces an inclusion-reversing bijection

{
$$\Gamma$$
-stable K-subspaces of \mathfrak{t}_K } \rightleftharpoons { Γ -stable K-subspaces of $(\mathfrak{t}_K)^*$ },
 $\mathfrak{a} \longrightarrow \mathfrak{a}^\perp := \{\ell \in (\mathfrak{t}_K)^* : \ell(\mathfrak{a}) = 0\},$

where $^{\perp}L := \{t \in \mathfrak{t}_K : L(t) = 0\} \leftarrow L.$

Note that since \mathfrak{t}_K is abelian, we can write "subalgebras" for "subspaces" on the left. Also, the projection $(\mathfrak{t}_K)^* \twoheadrightarrow (\mathfrak{t}_K)^*/\mathfrak{a}^{\perp}$ can be identified with the restriction map $(\mathfrak{t}_K)^* \twoheadrightarrow \mathfrak{a}^*, \ \ell \mapsto \ell_{|\mathfrak{a}}$. Since X spans $(\mathfrak{t}_K)^*$ by 3.1.8.i, the above induces an

MÉMOIRES DE LA SMF 151

 $\mathbf{32}$

inclusion-reversing bijection

$$\{\Gamma\text{-stable subalgebras of }\mathfrak{t}_K\} \rightleftharpoons \{\Gamma\text{-stable saturated submodules of } X\},\\ \mathfrak{a} \longrightarrow \ _\mathfrak{a} X := \{\chi \in X : \chi(\mathfrak{a}) = 0\},\\ \mathfrak{a}_Y := \{t \in \mathfrak{t}_K : Y(t) = 0\} \longleftarrow Y$$

where a submodule $Y \subseteq X$ is called *saturated* if X/Y has no \mathbb{Z} -torsion. Using 3.1.8.i again, we can identify $X/_{\mathfrak{a}}X$ with a lattice in \mathfrak{a}^* so that the projection $X \twoheadrightarrow X/_{\mathfrak{a}}X$ again corresponds to the restriction $\chi \mapsto \chi_{|\mathfrak{a}}$.

Finally, note that the set on the left is in bijection with the set of k-subalgebras of t: Every Γ -stable K-subalgebra (= K-subspace) \mathfrak{a} of \mathfrak{t}_K arises by scalar extension from a k-subalgebra (= k-subspace) of \mathfrak{t} , namely, from \mathfrak{a}^{Γ} (see e.g. [Spr3, Prop. 11.1.4]).

DEFINITION 3.1.25. — Define the Γ -stable saturated submodule

$$X_0 := \Big\{ \chi \in X : \sum_{\sigma \in \Gamma} {}^{\sigma} \chi = 0 \Big\}.$$

In general we have

(8)
$$|\Gamma| \cdot X_0 \subsetneq \left\langle \left\{ {}^{\sigma} \chi - \chi : \chi \in X \right\} \right\rangle_{\mathbb{Z}} \subsetneq X_0$$

but the following shows that X/X_0 is well-suited as Γ -coinvariants of X. Let $\mathfrak{r} \subseteq \mathfrak{t}$ be a subalgebra. $\mathfrak{a} := \mathfrak{r}_K$ is a Γ -stable subalgebra of \mathfrak{t}_K . We have an induced action of Γ on $X/\mathfrak{a}X$.

PROPOSITION 3.1.26. — With the above notation, the following are equivalent:

- i. $X_0 \subseteq {}_{\mathfrak{a}}X$.
- ii. The Γ -action on $X/_{\mathfrak{a}}X$ is trivial.
- iii. $\operatorname{im}(\alpha_{|\mathfrak{r}}) \subseteq k$ for all $\alpha \in R$.
- iv. \mathfrak{r} is a split toral subalgebra of \mathfrak{g} .

Proof. — i. \Rightarrow ii. Obviously, $\chi - {}^{\sigma}\chi \in X_0$ for all $\chi \in X$, $\sigma \in \Gamma$. But condition ii. is equivalent to $\chi - {}^{\sigma}\chi \in {}_{\mathfrak{a}}X$ for all $\chi \in X, \sigma \in \Gamma$.

ii. \Rightarrow i. Let $\chi \in X_0$. As remarked, ii. says that $\chi - {}^{\sigma}\chi \in {}_{\mathfrak{a}}X$ for all $\sigma \in \Gamma$, so also

$$\sum_{\sigma \in \Gamma} (\chi - {}^{\sigma}\chi) = |\Gamma| \cdot \chi - \sum_{\substack{\sigma \in \Gamma \\ = 0}} {}^{\sigma}\chi \in {}_{\mathfrak{a}}X$$

and therefore $\chi \in {}_{\mathfrak{a}}X$ since ${}_{\mathfrak{a}}X$ is saturated.

ii. \Leftrightarrow iii. Choose a k-basis t_1, \ldots, t_n of t such that t_1, \ldots, t_r $(r \leq n)$ form a basis of \mathfrak{r} . Then using our identifications and the description of the Galois action in 3.1.22, we see that ii. is equivalent to $\alpha(t_i) \in K^{\Gamma} = k$ for $1 \leq i \leq r$ and all $\alpha \in R$, which is iii.

iii. \Rightarrow iv. Remember the trace $\hat{x} := \sum_{\sigma \in \Gamma} \sigma(x) \in \mathfrak{g}$ for $x \in \mathfrak{g}_K$. For $\alpha \in R$, $x \in (\mathfrak{g}_K)_{\alpha}$ and $r \in \mathfrak{r}$, condition iii. implies $[1 \otimes r, \hat{x}] = \alpha(1 \otimes r)\hat{x}$; that is, \hat{x} is an element of the weight space to the weight $\alpha_{|\mathfrak{r}|} \circ \mathfrak{f} \mathfrak{r}$ in \mathfrak{g} . Now decomposing

$$x = \sum_{\alpha \in R \cup \{0\}} x_{\alpha}$$

we have

$$\widehat{x} = \sum_{\alpha \in R \cup \{0\}} \widehat{x}_{\alpha}$$

and by k-linearity and surjectivity of the trace, it follows that \mathfrak{g} is the sum of the weight spaces of \mathfrak{r} in \mathfrak{g} . Hence each $\mathrm{ad}_{\mathfrak{g}}(r)$ for $r \in \mathfrak{r}$ is diagonalisable over k.

iv. \Rightarrow iii. This follows from the relations between the weight decompositions with respect to $\mathfrak{r}, \mathfrak{t}, \mathfrak{r}_K$ and \mathfrak{t}_K in 3.1.19 and 3.1.20.

Corollary 3.1.27

i. X₀ is the annihilator of s_K in X. Every subalgebra of t which is split toral in g is contained in s: in other words, s is the unique maximal split toral subalgebra of g contained in t. The map ρ is induced by the projection X → X/X₀ which we will also call ρ.

ii.
$$R_0 = R \cap X_0$$
.

Proof

- i. The proposition and the foregoing discussions imply that the annihilator \mathfrak{a}_{X_0} of X_0 is of the form \mathfrak{r}_K , where \mathfrak{r} is a split toral subalgebra of \mathfrak{g} that contains all other subalgebras of \mathfrak{t} which are split toral in \mathfrak{g} . In particular, it contains \mathfrak{s} , and by maximality of \mathfrak{s} , must be equal to it.
- ii. The assertion follows from the first statement in i.

Note that in general, X_0 is *not* generated by R_0 , see example 3.2.9. The submodule generated by R_0 will turn up when we now have a closer look at:

3.1.2.1. The anisotropic kernel. — By 3.1.21, $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ is reductive. More precisely, the derived algebra of $(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}))_K$ is the semisimple

$$\underbrace{\left(\sum_{\alpha\in R_0}K\cdot H_{\alpha}\right)}_{\subseteq\mathfrak{t}_K}\oplus\bigoplus_{\alpha\in R_0}(\mathfrak{g}_K)_{\alpha}$$

and its centre is $\bigcap_{\alpha \in R_0} \ker(\alpha) \subseteq \mathfrak{t}_K$. Write $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{Z} \times \mathcal{D}\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ where \mathfrak{Z} is the centre of $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$, and the derived algebra $\mathcal{D}\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ is semisimple.

REMARK 3.1.28. — We have the chain of inclusions

$$\mathfrak{s} \subseteq \mathfrak{Z} \subseteq \mathfrak{t} \subseteq \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$$

(For the second one, remark that \mathfrak{t} is its own centraliser in \mathfrak{g} and a fortiori in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$.) As \mathfrak{t} is abelian, the second inclusion is an equality if the third one is, and we will see the converse shortly. In example 3.2.9, the first inclusion will be proper; in example 3.2.5, the second and third one will be.

REMARK 3.1.29. — All elements of $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ are semisimple (in \mathfrak{g}). All elements of the semisimple Lie algebra $\mathfrak{g}_a := \mathcal{D}\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ are semisimple (in \mathfrak{g}_a).

Proof. — In step 2 of the proof of 3.1.5 we have seen that with an element $x \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$, its semisimple and nilpotent part are also in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$. So for the first assertion it suffices to see there is no $x \neq 0$ in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ which is nilpotent in \mathfrak{g} . If there were, by a Jacobson-Morozov argument improved with weight decomposition like in the proof of 3.1.9, we would find an $h \in \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ with [h, x] = 2x on the one hand, but h ad-diagonalisable in \mathfrak{g} on the other hand; since it commutes with \mathfrak{s} , the latter implies that h is contained in \mathfrak{s} , hence [h, x] = 0, a contradiction. For the second assertion note that for $x \in \mathfrak{g}_a$, $\mathrm{ad}_{\mathfrak{g}_a}(x)$ is the restriction of the semisimple endomorphism $\mathrm{ad}_{\mathfrak{g}}(x)$ to the $\mathrm{ad}_{\mathfrak{g}}(x)$ -stable subspace \mathfrak{g}_a , and therefore itself semisimple.

DEFINITION 3.1.30. — The semisimple Lie algebra

$$\mathfrak{g}_a := \mathcal{D}\mathfrak{z}_\mathfrak{g}(\mathfrak{s}) \qquad (\cong \mathfrak{z}_\mathfrak{g}(\mathfrak{s})/\mathfrak{Z})$$

is called the *anisotropic kernel* of \mathfrak{g} (with respect to \mathfrak{s}). We call X_a (resp. V_a) the submodule of X (resp. subspace of V) generated by R_0 .⁽⁵⁾ As noted before, we have $X_a \subseteq X_0$ without equality in general. Remark that in the duality setup before definition 3.1.25, X_a is the annihilator of \mathfrak{Z}_K in X.

The anisotropic kernels are the bad guys because they generally resist classification. In later chapters we will see that they disappear over C1 fields and are under strict control over \mathfrak{p} -adic fields. The opposite cases $\mathfrak{g}_a = \mathfrak{g}$ and $\mathfrak{g}_a = \{0\}$ will be treated in Section 4.2, where we also say something about the terminology.

PROPOSITION 3.1.31. — The set $\mathfrak{t}_a := \mathfrak{t} \cap \mathfrak{g}_a$ is a maximal toral subalgebra of \mathfrak{g}_a , and

$$(\mathfrak{t}_a)_K = \sum_{\alpha \in R_0} K \cdot H_\alpha$$

is split maximal toral in

$$(\mathfrak{g}_a)_K = (\mathfrak{t}_a)_K \oplus \bigoplus_{\alpha \in R_0} (\mathfrak{g}_K)_{\alpha}.$$

Restriction $\alpha \mapsto \alpha_{|(\mathfrak{t}_a)_K}$ induces a bijection $R_0 \simeq R((\mathfrak{g}_a)_K, (\mathfrak{t}_a)_K)$.

^{5. &}quot;*a*" for anisotropic. We found notations like X^0 (cf. **[Sat2]**, **[Sat3]**) or X_1 (cf. **[Spr3]**) more confusing.

Proof. — It is clear that \mathfrak{t}_a is a toral subalgebra of \mathfrak{g}_a . Because $\mathfrak{Z} \subseteq \mathfrak{t}$, an abelian subalgebra $\mathfrak{a} \supseteq \mathfrak{t}_a$ in \mathfrak{g}_a would give rise to an abelian subalgebra $\mathfrak{Z} + \mathfrak{a} \supseteq \mathfrak{t}$ in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$, but we had remarked in 3.1.6 that \mathfrak{t} is maximal abelian in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$. So \mathfrak{t}_a is maximal toral in \mathfrak{g}_a and consequently $(\mathfrak{t}_a)_K$ is split maximal toral in $(\mathfrak{g}_a)_K$. Finally, we have $(\mathfrak{g}_a)_K = \mathcal{D}((\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}))_K)$ (cf. [**Bo2**, I.1.9]), and according to earlier remarks, $\mathfrak{t}_K \cap (\mathfrak{g}_a)_K = \sum_{\alpha \in R_0} K \cdot H_{\alpha}$ also is a maximal toral subalgebra of $(\mathfrak{g}_a)_K$, so the evident inclusion $(\mathfrak{t}_a)_K \subseteq \mathfrak{t}_K \cap (\mathfrak{g}_a)_K$ is an equality. The last assertion is an immediate consequence. \Box

COROLLARY 3.1.32. — One has the equivalences

36

 $\mathfrak{Z} = \mathfrak{t} \iff \mathfrak{t}_a = \{0\} \iff \mathfrak{g}_a = \{0\} \iff \mathfrak{Z} = \mathfrak{z}_\mathfrak{g}(\mathfrak{s}) \iff \mathfrak{t} = \mathfrak{z}_\mathfrak{g}(\mathfrak{s}) \iff X_a = \{0\},$

3.1.3. Relative Weyl groups, Γ -linear orders and the twisted Galois action For later use, we introduce certain subgroups of the Weyl group W(R).

LEMMA 3.1.33. — Mapping $s_{\alpha} \in \operatorname{GL}(V_a)$ to $s_{\alpha} \in \operatorname{GL}(V)$ defines a group isomorphism from $W(R_0)$ to the subgroup of W(R) generated by $\{s_{\alpha} : \alpha \in R_0\}$; call this group W_0 . It is invariant under the Galois action on W(R) from 3.1.24.

Proof. — The first assertion is true in the general situation of a subspace $U \subseteq V$ generated by a symmetric and closed subsystem of roots as considered in [**Bo2**, VI.7, Prop. 23]). Namely, there is a W(R)-invariant scalar product $\langle ., . \rangle$ on V (cf. [**Bo2**, VI.1.1, Prop. 3]) and hence (cf. [**Bo2**, V.2.3]), $s_{\alpha} \in \text{GL}(V)$ is given by

$$s_{\alpha}(v) = v - 2 \frac{\langle v, \alpha \rangle}{\langle \alpha, \alpha \rangle} \alpha.$$

Writing U^{\perp} for the orthogonal complement of U in V with respect to $\langle ., . \rangle$, we see that for $\alpha \in U$, the spaces U and U^{\perp} are s_{α} -invariant; moreover, the restriction $s_{\alpha|U}$ is $s_{\alpha} \in \mathrm{GL}(U)$ (by the identification [**Bo2**, VI.1.7, Prop. 23]), whereas $s_{\alpha|U^{\perp}} = \mathrm{id}_{U^{\perp}}$.

The invariance under the Galois action follows from the invariance of R_0 under the Galois action.

REMARK 3.1.34. — The proof also shows: The induced actions of W_0 on X/X_a and V/V_a , a fortiori on X/X_0 and V/V_0 , are trivial.

One of our aims is to find the Weyl group of \overline{R} as a subquotient in W(R). In this direction, up to here we can prove:

LEMMA 3.1.35. — Set $W_{\Gamma} := \{ w \in W(R) : w(X_0) = X_0 \}.$

- i. W_0 is a normal subgroup of W_{Γ} .
- ii. We have a homomorphism $\pi: W_{\Gamma} \to A(\overline{R})$ with $W_0 \subseteq \ker(\pi)$.

Proof. — Part i follows from $ws_{\alpha}w^{-1} = s_{w(\alpha)}$ for $\alpha \in R_0, w \in W$. For part ii, we had seen earlier that the morphism of Γ -modules

$$\rho: X \longrightarrow X/X_0$$

corresponds to the restriction $\chi \mapsto \chi_{|\mathfrak{s}_K}$ (identifying X/X_0 with a lattice in \mathfrak{s}_K^*), and induces a surjection $R \twoheadrightarrow \overline{R}$. For $w \in W_{\Gamma}$, define $\pi(w)(\rho(\chi)) := \rho(w(\chi))$ for any $\chi \in X$. By the definition of W_{Γ} , $\pi(w)$ is a well-defined automorphism of X/X_0 that stabilises \overline{R} . Equality $\pi(W_0) = \{id\}$ follows from the previous remark. \Box

We will see in 3.1.48 that in fact $\ker(\pi) = W_0$ and $\operatorname{im}(\pi) = W(\overline{R})$. But for this, we need the notion of a Γ -*linear order*, introduced by Satake [Sat2].

3.1.3.1. Γ -linear orders. — Let \leq be a linear group order on X, i.e. a total order with $x \leq y \Rightarrow x+z \leq y+z$ for all $x, y, z \in X$. As usual, x > y means the negation of $x \leq y$, and the meaning of < and \geq is clear. Defining one such linear order is equivalent to choosing an additively closed subset $X^+ \subset X$ such that $X = X^+ \sqcup \{0\} \sqcup -X^+$, the correspondence being given by $X^+ = \{\chi \in X : \chi > 0\}$.

LEMMA 3.1.36. — The following are equivalent:

- i. For $\chi \in X \setminus X_0$ with $\chi > 0$, we have $\chi' > 0$ for all $\chi' \equiv \chi \mod X_0$.
- ii. For $\chi \in X \setminus X_0$ with $\chi > 0$, we have ${}^{\sigma}\chi > 0$ for all $\sigma \in \Gamma$.

Proof. — i. \Rightarrow ii. is clear from ${}^{\sigma}\chi - \chi \in X_0$. For the converse, remark first that ii. implies $\chi > 0 \Leftrightarrow$ all ${}^{\sigma}\chi > 0 \Leftrightarrow \sum_{\sigma \in \Gamma} {}^{\sigma}\chi > 0$ for any $\chi \in X$. Now if χ as in i. is given and $\chi' = \chi + \chi_0$ with $\chi_0 \in X_0$, then

$$\sum_{\sigma\in\Gamma}{}^{\sigma}\chi' = \sum_{\sigma\in\Gamma}{}^{\sigma}\chi$$

and this is > 0 by ii.

DEFINITION 3.1.37. — A linear group order on X is called a Γ -linear order if it satisfies the equivalent conditions above.

REMARK 3.1.38. — A Γ -linear order on X clearly induces linear group orders on X/X_0 and X_0 . Conversely, given linear group orders on X/X_0 and X_0 , there is a unique Γ -linear order on X inducing them. In particular, Γ -linear orders exist: For a \mathbb{Z} -module isomorphic to some \mathbb{Z}^r (as X_0 and X/X_0 are), the *lexicographic order* on \mathbb{Z}^r :

 $(x_1, \ldots, x_r) > 0 \iff$ there is $s \in \{1, \ldots, r\}$ with $x_1 = \cdots = x_{s-1} = 0$ and $x_s > 0$ transports to a linear group order on it (depending on the chosen isomorphism, i.e. on the choice of basis).

Given any linear group order on X, the roots in R which are positive with respect to this order form a positive system of roots in the sense of [**Bo2**, VI.1.7]; therefore there is a unique basis Δ of R such that the positive roots with respect to this basis are exactly the positive roots with respect to the given linear order. A basis associated in this way with a Γ -linear order is called a Γ -basis.

From now on, fix a Γ -basis Δ of R.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

In general, for a root system R with basis D and closed subsystem R', it is not true that $D \cap R'$ is a basis of R'. In our situation, the Γ -linear order remedies this:

PROPOSITION 3.1.39. — Let Δ_0 be the basis of R_0 with respect to the induced order. Then $\Delta_0 = \Delta \cap R_0 = \Delta \cap X_0$.

Proof. — It suffices to show that every (w.l.o.g.: positive) $\alpha \in R_0$ is a linear combination of the elements of $\Delta \cap R_0$. Let $\delta_1, \ldots, \delta_r$ be the different elements of Δ arranged such that $\Delta \cap R_0 = \{\delta_i : i < s\}$ for a certain $0 \le s \le r+1$. Write $\alpha = \sum_{i=1}^r n_i \delta_i$ with all $n_i \in \mathbb{Z}_{\geq 0}$. If we had $n_k \ge 1$ for some $k \ge s$, then

$$\delta_k \equiv \delta_k - \alpha = \left(\sum_{\substack{1 \le i \le r \\ i \ne k}} -n_i \delta_i\right) - (n_k - 1)\delta_k \mod X_0,$$

but $\delta_k \notin X_0$ is positive and the right hand side is not, contradicting 3.1.36.

Let $\overline{\Delta} := \rho(\Delta \setminus \Delta_0) = \rho(\Delta) \setminus \{0\}$. We will see in 3.1.45 that this is a basis of \overline{R} .

LEMMA 3.1.40. — Let Δ' be another Γ -basis.

- i. $\Delta = \Delta'$ if and only if $\Delta_0 = \Delta'_0$ and $\overline{\Delta} = \overline{\Delta'}$.
- ii. If $\overline{\Delta} = \overline{\Delta'}$, then there is a $w_0 \in W_0$ with $w_0(\Delta) = \Delta'$ (and unique in W(R) with this property).

Proof. — For the non-trivial direction in i, the conditions imply that all $\alpha \in \Delta'$ are positive with respect to a Γ -linear order defining Δ , hence with respect to Δ , so the bases must coincide. For part ii, by the proposition and 3.1.33 there is $w_0 \in W_0$ with $w_0(\Delta_0) = \Delta'_0 = w_0(\Delta) \cap R_0 = (w_0(\Delta))_0$; on the other hand,

$$\overline{w_0(\Delta)} = \rho(w_0(\Delta) \setminus \Delta'_0) = \pi(w_0)(\rho(\Delta)) \setminus \{0\} = \rho(\Delta) \setminus \{0\} = \overline{\Delta} = \overline{\Delta}',$$

so $w_0(\Delta) = \Delta'$ by part i. The uniqueness assertion follows from W(R) acting simply transitively on bases.

3.1.3.2. The twisted Galois action. — We will now define a new action of Γ on the roots, as follows:

Let $\sigma \in \Gamma$. By 3.1.22 and the characterisation of bases in [**Bo2**, VI.1.7, Cor. 3], σ acts on R in such a way that $\sigma_A(\Delta)$ is another basis (remember σ_A is the element of A(R) through which σ acts). Hence there is a unique element $w_{\sigma} \in W(R)$ such that $w_{\sigma}(\sigma_A(\Delta)) = \Delta$. We will soon show the crucial fact that $w_{\sigma} \in W_0$.

DEFINITION 3.1.41. — For $\chi \in X$, set ⁽⁶⁾

$$t(\sigma)(\chi) := w_{\sigma}(\sigma_A(\chi))$$

MÉMOIRES DE LA SMF 151

38

^{6.} The twisted action is denoted as σ^* in [**Tit1**, 2.3], $\Delta \sigma$ in [**BT1**, 6.2], [σ] in [**Sat3**, II 2.1] (where w_{σ} is our w_{σ}^{-1}), $\varphi(\sigma)$ in [**Spr2**, 2.6.1], $\tau(\sigma)$ in [**Spr3**, 15.5.2].

LEMMA 3.1.42. — This defines a group homomorphism $t : (\mathcal{G} \twoheadrightarrow)\Gamma \to \operatorname{Aut}_{\mathbb{Z}}(X)$ such that each $t(\sigma)$ stabilises R and Δ ; so we can and will also view t as homomorphism to the subgroup $\operatorname{Aut}(R, \Delta) \simeq A(R)/W(R)$ of A(R).

Proof. — For the first assertion, recall the Γ -action on W(R) by conjugation from 3.1.24. The uniqueness statement before the definition implies that $w_{\sigma\tau} = w_{\sigma}{}^{\sigma}(w_{\tau})$ for any $\sigma, \tau \in \Gamma$, meaning that $\sigma \mapsto w_{\sigma}$ is a 1-cocycle $\Gamma \to W(R)$; this gives $t(\sigma\tau) = w_{\sigma\tau}\sigma_A\tau_A = w_{\sigma}\sigma_Aw_{\tau}\tau_A = t(\sigma)t(\tau)$. The rest follows directly from 3.1.22 and the definitions.

PROPOSITION 3.1.43 (Properties of t)

- i. $t(\Gamma)$ stabilises Δ_0 ; a fortiori, also $\Delta \setminus \Delta_0$, R_0 and $R \setminus R_0$.
- ii. For $\chi \in X$ and $\sigma \in \Gamma$, we have

$$t(\sigma)(\chi) \equiv {}^{\sigma}\chi \mod X_a \quad and \quad {}^{\sigma}\chi \equiv \chi \mod X_0.$$

- iii. For $\chi, \psi \in X \setminus X_0$, we have $\rho(\chi) = \rho(\psi)$ if there is $\sigma \in \Gamma$ with $t(\sigma)(\chi) = \psi$.
- iv. Write $\Delta = \{\delta_1, \ldots, \delta_r\}$ such that $\Delta_0 = \{\delta_1, \ldots, \delta_{s-1}\}$ (as in the proof of 3.1.39). Then for $i \geq s$ and $\sigma \in \Gamma$, we have $\sigma_A(\delta_i) = t(\sigma)(\delta_i) + \sum_{j=1}^{s-1} n_j \delta_j$ for certain $n_j \in \mathbb{Z}_{>0}$.
- v. (Converse of iii for simple roots.) With the notations of iv, if $\rho(\delta_i) = \rho(\delta_k)$ for $i, k \geq s$, then there is $\sigma \in \Gamma$ with $t(\sigma)(\delta_i) = \delta_k$.

Proof

i. Let $\sigma \in \Gamma$. With the same argument as before definition 3.1.41, there is a unique $w_{0,\sigma} \in W_0$ such that $w_{0,\sigma}(\sigma_A(\Delta_0)) = \Delta_0$. So it suffices to show the following *claim*: Using the identification of 3.1.33, this $w_{0,\sigma}$ is the w_{σ} defined before.

As seen earlier, $w_{0,\sigma}$ (being an element of W_0) induces the identity on X/X_0 , which by 3.1.36 implies that $w_{0,\sigma}(\alpha) > 0$ and $\notin R_0$ for $\alpha \in R \setminus R_0$ with $\alpha > 0$. Analogously $\sigma_A(\alpha) > 0$ and $\notin R_0$ for $\alpha \in R \setminus R_0$ with $\alpha > 0$. Thus $w_{0,\sigma} \circ \sigma_A$ stabilises the set of positive elements of R which are not in R_0 . By definition, it stabilises the positive elements of R_0 , so it stabilises the set of all positive elements. Thus, $w_{0,\sigma}(\sigma_A(\Delta))$ is a basis of R consisting of positive elements with respect to the basis Δ and hence must itself be Δ , and our claim follows from the uniqueness property used to define w_{σ} .

- ii. The first claim follows from $w_{\sigma} \in W_0$ and 3.1.34, the second from 3.1.27.
- iii. With part ii this is immediate from 3.1.26 and 3.1.27.
- iv. By part i, $t(\sigma)(\delta_i) = \delta_k$ for some $k \ge s$, so by part ii, $\sigma \delta_i \in R$ is of the form

$$\delta_k + \sum_{j=1}^{s-1} n_j \delta_j$$

with $n_j \in \mathbb{Z}$. Δ is a basis and one coefficient is positive, so all $n_j \geq 0$.

v. For any $\chi, \psi \in V \setminus V_0$, $\rho(\chi) = \rho(\psi)$ implies (and is in fact equivalent to) $\sum_{\sigma \in \Gamma} \sigma_A(\chi) = \sum_{\sigma \in \Gamma} \sigma_A(\psi)$. Thus iv implies

$$\sum_{\sigma \in \Gamma} t(\sigma)(\delta_i) = \sum_{\sigma \in \Gamma} t(\sigma)(\delta_k) + \sum_{j \le s-1} m_j \delta_j$$

for certain $m_j \in \mathbb{Z}$. Since $t(\Gamma)$ stabilises $\{\delta_j : j \ge s\}$ by i, and Δ is a vector space basis of V, one $t(\sigma)(\delta_i)$ has to be δ_k (and all $m_j = 0$).

REMARK 3.1.44. — The twisted Galois action on roots can also be defined (on Δ , and then linearly extended) via the Galois action on parabolic subalgebras, and the correspondence of [**Bo2**, VIII.3.4, Rem.] between elements of Δ and certain parabolic subalgebras. Cf. [**Tit1**, 2.3] and [**BT1**, 6.2].

Corollary 3.1.45

- i. V_0 is generated by $\Delta_0 \cup \{\delta \varepsilon : \delta, \varepsilon \in \Delta \setminus \Delta_0, \rho(\delta) = \rho(\varepsilon)\}.$
- ii. $\overline{\Delta}$ is a basis of \overline{R} .

Proof

40

- i. By (8), V_0 is generated by $\{{}^{\sigma}\chi \chi : \chi \in V, \sigma \in \Gamma\}$ and a fortiori (Δ spans V) by $\{{}^{\sigma}\delta - \delta : \delta \in \Delta, \sigma \in \Gamma\}$. Since ${}^{\sigma}\delta \in R \cap X_0 = R_0 \subset X_a = \mathbb{Z}\Delta_0$ for $\delta \in \Delta_0$, it is spanned by $\Delta_0 \cup \{{}^{\sigma}\delta - \delta : \delta \in \Delta \setminus \Delta_0, \sigma \in \Gamma\}$, so by part iv of the previous proposition, by $\Delta_0 \cup \{t(\sigma)(\delta) - \delta : \delta \in \Delta \setminus \Delta_0, \sigma \in \Gamma\}$ which is contained in the set in the claim.
- ii. It suffices to show that the distinct elements of $\overline{\Delta}$, call them $\lambda_1, \ldots, \lambda_r$, are linearly independent in V/V_0 . For each of them fix one $\delta_{j(i)} \in \Delta \setminus \Delta_0$ with $\rho(\delta_{j(i)}) = \lambda_i$. Then the set $\Delta_0 \cup \{\delta - \delta_{j(i)} : 1 \le i \le r, \delta_{j(i)} \ne \delta \in \Delta \setminus \Delta_0, \rho(\delta) = \rho(\delta_{j(i)})\}$ is linearly independent (because the $\delta \in \Delta$ are), and by part i generates V_0 . Since its cardinality is $|\Delta| - r = \dim_k V - r$, we have $r = \dim_k V/V_0$.

3.1.3.3. Relative Weyl groups revisited. — We can now complete our earlier investigations of the Weyl group. Recall the groups W_{Γ}, W_0 and the map π from 3.1.35; we want to show that π induces an isomorphism $W_{\Gamma}/W_0 \simeq W(\overline{R})$. Along the way, we improve 3.1.15.

Recall from [Bo2, VIII.5.2] that there is a well-defined group homomorphism

$$\varepsilon_{\mathfrak{t}_K} : \operatorname{Aut}_e(\mathfrak{g}_K, \mathfrak{t}_K) \longrightarrow W(R), \quad s \longmapsto [\chi \mapsto \chi \circ s_{|\mathfrak{t}_K}^{-1} \text{ for } \chi \in X]$$

(the contragredient of the restriction to \mathfrak{t}_K) which induces an isomorphism

(9)
$$\operatorname{Aut}_{e}(\mathfrak{g}_{K},\mathfrak{t}_{K})/\ker(\varepsilon_{\mathfrak{t}_{K}})\cong W(R),$$

where $\ker(\varepsilon_{\mathfrak{t}_K}) = \{s \in \operatorname{Aut}_e(\mathfrak{g}_K, \mathfrak{t}_K) : s_{|\mathfrak{t}_K} = \operatorname{id}_{\mathfrak{t}_K}\}.$

Lemma 3.1.46

i. For $s \in \operatorname{Aut}_e(\mathfrak{g}_K, \mathfrak{t}_K)$,

$$\varepsilon_{\mathfrak{t}_{K}}(s) \in W_{\Gamma} \iff s \in \operatorname{Aut}_{e}(\mathfrak{g}_{K},\mathfrak{s}_{K}),$$
$$\varepsilon_{\mathfrak{t}_{K}}(s) \in W_{0} \iff s_{|\mathfrak{s}_{K}} = \operatorname{id}_{\mathfrak{s}_{K}}.$$

ii. $\operatorname{Aut}_e(\mathfrak{g}_K,\mathfrak{s}_K) =$

$$\left(\operatorname{Aut}_{e}(\mathfrak{g}_{K},\mathfrak{t}_{K})\cap\operatorname{Aut}_{e}(\mathfrak{g}_{K},\mathfrak{s}_{K})\right)\cdot\left\{s\in\operatorname{Aut}_{e}(\mathfrak{g}_{K}):s_{|\mathfrak{s}_{K}}=\operatorname{id}_{\mathfrak{s}_{K}}\right\}.^{(7)}$$

iii. There is a canonical isomorphism

$$\operatorname{Aut}_{e}(\mathfrak{g}_{K},\mathfrak{s}_{K})/\{s\in\operatorname{Aut}_{e}(\mathfrak{g}_{K}):s_{|\mathfrak{s}_{K}}=\operatorname{id}_{\mathfrak{s}_{K}}\}\cong W_{\Gamma}/W_{0}.$$

Proof. — Part i follows from the duality of \mathfrak{s}_K and X_0 in the sense of 3.1.27. Part iii is a consequence of (9), the first two parts and one of the isomorphism theorems.

In part ii, the inclusion " \supseteq " is trivial. So let $s \in \operatorname{Aut}_e(\mathfrak{g}_K, \mathfrak{s}_K)$; then $s(\mathfrak{t}_K)$ is another maximal toral subalgebra of \mathfrak{g}_K , and is contained in $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$ (because s stabilises \mathfrak{s}_K and a fortiori its centraliser). So by the conjugacy theorem 3.1.16⁽⁸⁾ applied to \mathfrak{t}_K and $s(\mathfrak{t}_K)$ in $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$, there is $s_1 \in \operatorname{Aut}_e(\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K))$ with $s_1(s(\mathfrak{t}_K)) = \mathfrak{t}_K$, so s_1s in $\operatorname{Aut}_e(\mathfrak{g}_K)$ stabilises both \mathfrak{t}_K and \mathfrak{s}_K . The proof is finished with the following fact: s_1 , like any element of $\operatorname{Aut}_e(\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K))$, can naturally be extended to an element of $\{s \in \operatorname{Aut}_e(\mathfrak{g}_K) : s_{|\mathfrak{s}_K} = \operatorname{id}_{\mathfrak{s}_K}\}.$

To see this, let $n \in \mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$ be $\mathrm{ad}_{\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)}$ -nilpotent; write $n = n_{\mathfrak{z}} + n_{\mathcal{D}}$ for the decomposition $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K) = \mathfrak{Z} \times \mathcal{D}$ into centre and semisimple part. Then $\exp(\mathrm{ad}_{\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)} n) = \exp(\mathrm{ad}_{\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)} n_{\mathcal{D}})$ on $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$, and $n_{\mathcal{D}}$ is nilpotent in \mathcal{D} . By 2.1.8, $\mathrm{ad}_{\mathfrak{g}_K}(n_{\mathcal{D}})$ is nilpotent, and exp of it obviously extends $\exp(\mathrm{ad}_{\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)} n)$ (which fixes \mathfrak{s}_K pointwise).

We can now put things together with 3.1.15. The homomorphism $\varepsilon_{\mathfrak{s}}$ via our identifications extends to

$$\varepsilon_{\mathfrak{s}_K} : \operatorname{Aut}_e(\mathfrak{g}_K, \mathfrak{s}_K) \longrightarrow A(\overline{R}), \quad s \longmapsto [\chi \mapsto \chi \circ s_{|\mathfrak{s}_K}^{-1} \text{ for } \chi \in X/X_0].$$

Its kernel is $\{s \in \operatorname{Aut}_e(\mathfrak{g}_K) : s_{|\mathfrak{s}_K} = \operatorname{id}_{\mathfrak{s}_K}\}$, and we had seen that its image contains $W(\overline{R})$. Going through our identifications shows that π is the composition of $\varepsilon_{\mathfrak{s}_K}$ with the (inverse of the) isomorphism from part iii of the last lemma.

LEMMA 3.1.47. — One has $\operatorname{im}(\varepsilon_{\mathfrak{s}_{K}}) = W(\overline{R})$.

^{7.} This is the analogue of $\mathcal{N}(S) = \mathcal{N}(S,T) \cdot \mathcal{Z}(S)$ in [**BT1**, 5.5]; cf. [**Sat3**, II.2.1.5].

^{8.} Being in the split case, we can also apply [**Bo2**, VIII.3.3] here. $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$ being reductive, not necessarily semisimple, is not a problem, because its centre is contained in \mathfrak{t}_K and fixed by s as well as by every automorphism of $\mathfrak{z}_{\mathfrak{g}_K}(\mathfrak{s}_K)$.

Proof. — Because we know already that the image contains $W(\overline{R})$, and because $A(\overline{R})$ is the semidirect product of $W(\overline{R})$ and $\operatorname{Aut}(\overline{R}, B)$ for an arbitrary basis B of \overline{R} , it suffices to show (for our chosen Γ -basis Δ):

$$\varepsilon_{\mathfrak{s}_K}(s) \in \operatorname{Aut}(\overline{R}, \overline{\Delta}) \Longrightarrow \varepsilon_{\mathfrak{s}_K}(s) = \operatorname{id}.$$

With the identifications of the previous lemma, we can assume w.l.o.g. that $s \in Aut_e(\mathfrak{g}_K, \mathfrak{t}_K)$ and look at $\varepsilon_{\mathfrak{t}_K}(s) \in W_{\Gamma}$. Then we have

$$\overline{\varepsilon_{\mathfrak{t}_K}(s)(\Delta)} = \overline{\Delta}$$

by assumption, and $\varepsilon_{\mathfrak{t}_K}(s)(\Delta_0)$ is another basis of R_0 . So there is a (unique) $w \in W_0$ such that $w\varepsilon_{\mathfrak{t}_K}(s)(\Delta_0) = \Delta_0$, whereas (W_0 acting trivially on X/X_0)

$$\overline{w\varepsilon_{\mathfrak{t}_K}(s)(\Delta)} = \overline{\Delta}.$$

On the other hand, one easily sees that W_{Γ} sends Γ -bases to Γ -bases, so by 3.1.40.i, we have $w_{\varepsilon_{\mathfrak{t}_{K}}}(s)(\Delta) = \Delta$. The action of W(R) on bases is simple, so $\varepsilon_{\mathfrak{t}_{K}}(s) = w^{-1} \in W_{0}$, hence $\varepsilon_{\mathfrak{s}_{K}}(s) = \mathrm{id}$.

PROPOSITION 3.1.48. — One has:

- i. π induces an isomorphism $W_{\Gamma}/W_0 \cong W(\overline{R})$.
- ii. For $w \in W_{\Gamma}$, the following are equivalent:
 - (a) $w \in W_0$.

 $\mathbf{42}$

- (b) $\pi(w) = id.$
- (c) $\pi(w)(\overline{\Delta}) = \overline{\Delta}$.

iii. W_{Γ} operates simply transitively on the set of all Γ -bases.

Proof. — Part i follows from the foregoing discussion, and part ii from i (the nontrivial $(c) \Rightarrow (a)$ also follows directly from 3.1.40). For part iii, we have remarked in the previous proof that W_{Γ} does indeed act on the set of all Γ -bases. The action is simple because so is the action of W(R) on the set of all bases. The transitivity follows with part i and 3.1.40 from the transitivity of the actions of $W(\overline{R})$ and W_0 on the sets of bases of \overline{R} and R_0 , respectively, and the triviality of the W_0 -action on \overline{R} .

REMARK 3.1.49. — For the analogues in the group case, see 6.10 and parts of 5.3 in **[BT1]**, as well as **[Sat2**, §2] and **[Sat3**, II.2.1–2.3]. There, a different route is chosen: Not knowing a priori that \overline{R} is a root system, the group W_{Γ}/W_0 is directly identified with $\mathcal{N}(S)/\mathcal{Z}(S)$ for a maximal split torus S. Then by an intricate investigation into this group's action on (minimal) parabolic subgroups, it is shown that \overline{R} is a root system having W_{Γ}/W_0 as Weyl group.

In a more general context, the question when an analogous subquotient of the Weyl group is again a Weyl group of a root system is analysed in [Sch], see also [Hec]. It turns out that parts i and iii of the proposition are kind of equivalent, and that they impose strong conditions on the Galois action and other symmetry properties of the root system, which we will use later for the classification.

Whereas realising the Weyl group in the form N(T)/Z(T) for a suitable torus T is common in group theory, it seems that its realisation as quotient of automorphism groups of Lie algebras has only been done in [**Bo2**, VIII.5], covering only the split case.

3.1.4. The invariants and their independence from choices. — We can now define our invariants:

▷ The tuple $(X, R, \Delta, \Delta_0, t)$ ⁽⁹⁾ is called the *index* of our semisimple Lie algebra \mathfrak{g} (for now: with respect to all the choices we have made). We will show that, up to a natural notion of equivalence, the index depends only on \mathfrak{g} (in fact, only on its isomorphism class) and not on the choices.

 \triangleright Recall from 3.1.30 the *anisotropic kernel* \mathfrak{g}_a of our semisimple Lie algebra \mathfrak{g} (for now: with respect to the chosen \mathfrak{s}). The following lemma is obvious from its definition:

LEMMA 3.1.50. — Let \mathfrak{g} and \mathfrak{g}' be two semisimple Lie algebras over k with \mathfrak{s} (resp. \mathfrak{s}') chosen as before. Suppose there is an isomorphism $f : \mathfrak{g} \xrightarrow{\sim} \mathfrak{g}'$ with $f(\mathfrak{s}) = \mathfrak{s}'$. Then f induces an isomorphism $f_a : \mathfrak{g}_a \xrightarrow{\sim} \mathfrak{g}'_a$.

With the conjugacy theorem 3.1.16 we conclude:

COROLLARY 3.1.51. — Up to an elementary automorphism in \mathfrak{g} , \mathfrak{g}_a is independent of the choice of \mathfrak{s} . Up to isomorphism, it only depends on the isomorphism class of \mathfrak{g} .

Next, we will see in which way the index depends on g and the choices.

DEFINITION 3.1.52 (Congruence of indices). — Let $(X, R, \Delta, \Delta_0, t), (X', R', \Delta', \Delta'_0, t')$ be two indices. An isomorphism of abelian groups $h: X \xrightarrow{\sim} X'$ such that

- i. $h(R) = R', h(\Delta) = \Delta', h(\Delta_0) = \Delta'_0;$
- ii. $h(\alpha)^{\check{}} = \check{\alpha} \circ h^{-1}$ for all $\alpha \in R$ or equivalently, $(h(\alpha))^{\check{}}(h(\beta)) = \check{\alpha}(\beta)$ for all $\alpha, \beta \in R^{(10)}$;
- iii. $t'(g) = h \circ t(g) \circ h^{-1}$ for every $g \in \mathcal{G}$

is called a *congruence* $h: (X, R, \Delta, \Delta_0, t) \to (X', R', \Delta', \Delta'_0, t')$, and if such h exists, the two indices are called *congruent*. Obviously, congruence defines an equivalence relation on any set of indices.

Choosing, firstly, different splitting fields K and L (as long as both are finite and Galois) in an obvious way does not affect the index. To see this, one immediately reduces to the case $K \subseteq L$, and this is easily checked.

Next, let us vary \mathfrak{s} , \mathfrak{t} and Δ .

^{9.} There is some redundancy here; e.g. X can be recovered from R, different from the case of algebraic groups. We keep it like this since we often need both X and R anyway.

^{10.} In fact, condition ii follows from i (cf. [Win, 3.7.3.1.2/4 and 3.7.3.4.3]), but we wanted to state it explicitly.

PROPOSITION 3.1.53. — Let \mathfrak{g} and \mathfrak{g}' be two semisimple Lie algebras over k with $\mathfrak{s}, \mathfrak{t}, \Delta$ (resp. $\mathfrak{s}', \mathfrak{t}', \Delta'$) chosen as before, such that K is a finite Galois splitting field for both \mathfrak{t} and \mathfrak{t}' . Suppose there is an isomorphism (of K-Lie algebras) $F : \mathfrak{g}_K \xrightarrow{\sim} \mathfrak{g}'_K$ such that $F(\mathfrak{t}_K) = \mathfrak{t}'_K$ and $F(\mathfrak{s}_K) = \mathfrak{s}'_K$. Then in the following cases we have a congruence of the corresponding indices:

- i. $F = \mathrm{id}_K \otimes f$ for an isomorphism $f : \mathfrak{g} \simeq \mathfrak{g}'$.
- ii. $\mathfrak{g} = \mathfrak{g}'$ and $F \in \operatorname{Aut}_0(\mathfrak{g}_K)$.

Proof. — The map $\phi : X' \xrightarrow{\sim} X, \chi \mapsto \chi \circ (F_{|\mathfrak{t}_K})$ is an isomorphism that satisfies $\flat \phi(R') = R$, and

 $\triangleright \phi(X'_0) = X_0 \text{ (by } F(\mathfrak{s}_K) = \mathfrak{s}'_K \text{ and } 3.1.27 \text{)}.$

Call $\psi := \phi^{-1}$ its inverse. A moment's consideration with the defining property of $H_{\psi(\alpha)}$ (in 3.1.9) shows $H_{\psi(\alpha)} = F(H_{\alpha})$, in other words $(\psi(\alpha))^{\check{}} = \check{\alpha} \circ \psi^{-1}$. The second point above implies that $\psi(\Delta)$ is a Γ -basis of R'. So by 3.1.48, there is a unique $w' \in W'_{\Gamma}$ such that $w'(\psi(\Delta)) = \Delta'$, and by 3.1.39 a fortiori $w'(\psi(\Delta_0)) = \Delta'_0$. We claim that $h := w' \circ \psi$ is a congruence (from (X, \ldots) to (X', \ldots)), and with 2.1.11, the one thing left to check is what it does with the Galois actions.

So let $\sigma \in \Gamma$ be given. We have $w_{\sigma} \in W_0$ (and $w'_{\sigma} \in W'_0$) being defined uniquely in W(R) (and W(R')) by $w_{\sigma}(\sigma_{A(R)}(\Delta)) = \Delta$ (and $w'_{\sigma}(\sigma_{A(R')}(\Delta')) = \Delta'$). Setting

$$Th := \sigma_{A(R')} \circ h \circ \sigma_{A(R)}^{-1},$$

that is, $({}^{\sigma}h)(\chi) = {}^{\sigma}(h({}^{\sigma^{-1}}\chi))$, one calculates

$$\left(h \circ w_{\sigma} \circ ({}^{\sigma}h)^{-1}\right) \left(\sigma_{A(R')}(\Delta')\right) = \Delta'$$

so if the element in the left bracket is in W(R'), it must be w'_{σ} . Look at the two cases:

i. Here we have ${}^{\sigma}F := \sigma \circ F \circ \sigma^{-1} = F$, a fortiori ${}^{\sigma}\phi = \phi$, i.e. $\sigma_{A(R)}\psi^{-1} = \psi^{-1}\sigma_{A(R')}$. Hence

$$hw_{\sigma}\sigma_{A(R)}h^{-1}\sigma_{A(R')}^{-1} = w'\psi w_{\sigma}\sigma_{A(R)}\psi^{-1}w'^{-1}\sigma_{A(R')}^{-1}$$
$$= \underbrace{w'}_{\in W(R')}\psi w_{\sigma}\psi^{-1}\underbrace{\sigma_{A(R')}w'^{-1}\sigma_{A(R')}^{-1}}_{\in W(R')}$$

and the middle term is in W(R'), too: Indeed, $\psi(\alpha) = \check{\alpha} \circ \psi^{-1}$ implies $\psi s_{\alpha} \psi^{-1} = s_{\psi(\alpha)}$ and the claim follows.

ii. Let $e := (\mathfrak{g}_K, \mathfrak{t}_K, \Delta, (X_\alpha)_{\alpha \in \Delta})$ be an épinglage (cf. [**Bo2**, VIII.4, Def. 1]). Then $\sigma \in \Gamma$ transforms it into the épinglage $\sigma(e) := (\mathfrak{g}_K, \mathfrak{t}_K, \sigma_{A(R)}(\Delta), (\sigma(X_\alpha))_{\sigma_{A(R)}(\alpha)})$. On the other hand, the given F transforms it into the épinglage

$$e' := \left(\mathfrak{g}_K, \mathfrak{t}'_K, \psi(\Delta), (F(X_\alpha))_{\psi(\alpha)}\right)$$

This, in turn, is transformed by σ into

$$\sigma(e') := (\mathfrak{g}_K, \mathfrak{t}'_K, \sigma_{A(R')}\psi(\Delta), (\sigma(F(X_\alpha)))_{\sigma_{A(R')}\psi(\alpha)}).$$

Now by [**Bo2**, VIII.5.3, Prop. 5], there are unique $f_{\sigma} \in \operatorname{Aut}_0(\mathfrak{g}_K, \mathfrak{t}_K)$ and $f'_{\sigma} \in \operatorname{Aut}_0(\mathfrak{g}_K, \mathfrak{t}_K)$ such that $f_{\sigma}(\sigma(e)) = e$ and $f'_{\sigma}(\sigma(e')) = e'$. By the definition of w_{σ} and w'_{σ} and *ibid.* 2, Prop. 4, we have

$$\epsilon_{\mathfrak{t}_K}(f_\sigma) = w_\sigma \quad \text{and} \quad \epsilon_{\mathfrak{t}'_K}(f'_\sigma) = w'_\sigma.$$

Further note the following: It is easily seen that ${}^{\sigma}F \in \operatorname{Aut}_0(\mathfrak{g}_K)$ (use that $\sigma \circ \exp(\operatorname{ad} n) \circ \sigma^{-1} = \exp(\operatorname{ad}(\sigma(n)))$). Also, $F \circ f_{\sigma}$ and $f'_{\sigma} \circ {}^{\sigma}F$ both transform $\sigma(e)$ to e', and since they are both in $\operatorname{Aut}_0(\mathfrak{g}_K)$, they must be equal by *ibid.* . 3, Prop. 5; or in other words, not just the outer but also the inner diagram in



is commutative. This implies in particular

(10)
$$\psi \circ w_{\sigma} = w'_{\sigma} \circ^{\sigma} \psi$$

hence

$$\psi \circ w_{\sigma} \circ \sigma_{A(R)} = w'_{\sigma} \circ {}^{\sigma} \psi \circ \sigma_{A(R)} = w'_{\sigma} \circ \sigma_{A(R')} \circ \psi.$$

Inserting this into our element shows what we want:

$$hw_{\sigma}\sigma_{A(R)}h^{-1}\sigma_{A(R')}^{-1} = w'(\psi w_{\sigma} \sigma_{A(R)})\psi^{-1}w'^{-1}\sigma_{A(R')}^{-1}$$
$$= w'(w'_{\sigma}\sigma_{A(R')}\psi)\psi^{-1}w'^{-1}\sigma_{A(R')}^{-1}$$
$$= \underbrace{w'w'_{\sigma}}_{\in W(R')}\underbrace{\sigma_{A(R')}w'^{-1}\sigma_{A(R')}^{-1}}_{\in W(R')} \in W(R').$$

In both cases we conclude that for $\chi \in X$,

$$(t'(\sigma) \circ h)(\chi) = w'_{\sigma} (\sigma_A(h(\chi))) = h (w_{\sigma} (\sigma_A(h^{-1} (\sigma_A^{-1} (\sigma_A(h(\chi))))))))$$

= $h (w_{\sigma} (\sigma_A(\chi))) = (h \circ t(\sigma))(\chi),$

so that indeed $t'(\sigma) = h \circ t(\sigma) \circ h^{-1}$.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

Remark 3.1.54

46

- i. Certainly the above proposition holds in more cases, but we were not able to prove it in full generality in which its group analogue is claimed in [Sat2, p. 228] and [Sat3, pp. 85 et *seq.*]. We see a gap in this claim exactly where we had to go into the cases above.
- ii. The independence statement in [**BT1**, 6.2] (whose equation (3) should be read "modulo conjugation") corresponds to our case ii. It is proved via parabolic subgroups and uses in a crucial way that they all are conjugate to "standard" parabolic subgroups; cf. [**Tit1**, 2.3]. One could translate this to our setting, where the equivalence classes of parabolic subalgebras under the action of $\operatorname{Aut}_0(\mathfrak{g}_K)$ have a nice description in [**B02**, VIII.5.3, Rem. 4]. The key fact for both this and our proof (whose heart is equation (10)) is the assertion of [**B02**, VIII.5.3, Prop. 5] that the action of $\operatorname{Aut}_0(\mathfrak{g}_K)$ on the épinglages of \mathfrak{g}_K is simply transitive. We feel unsure about the last lines in Bourbaki's proof, but the assertion can certainly be proven e.g. using VIII.3.3, Corollaire to Prop. 10 instead of VII.3.2, Thm 1, without further scalar extension.

DEFINITION 3.1.55 (Congruence associated with an isomorphism)

In the setting of the above proposition, the congruence $w' \circ \psi$ defined in its proof is called the *congruence associated with* F.

Applying the proposition to F = id already shows that for fixed \mathfrak{s} and \mathfrak{t} , the index of \mathfrak{g} up to congruence does not depend on the choice of Δ , that is, the choice of the Γ -linear order. More precisely it shows that for two such choices, the congruence is given by a specific element of the Weyl group.

PROPOSITION 3.1.56. — Let $\mathfrak{s}, \mathfrak{t}$ respectively $\mathfrak{s}', \mathfrak{t}'$ in \mathfrak{g} be chosen as before, and let K be a splitting field for both \mathfrak{t} and \mathfrak{t}' . Then there is an $F \in \operatorname{Aut}_e(\mathfrak{g}_K)$ such that

$$F(\mathfrak{s}_K) = \mathfrak{s}'_K \quad and \quad F(\mathfrak{t}_K) = \mathfrak{t}'_K$$

Proof. — Theorem 3.1.16 gives us an elementary automorphism $h \in \operatorname{Aut}_e(\mathfrak{g})$ with $h(\mathfrak{s}) = \mathfrak{s}'$. A fortiori, $h(\mathfrak{z}_\mathfrak{g}(\mathfrak{s})) = \mathfrak{z}_\mathfrak{g}(\mathfrak{s}')$, and both $h(\mathfrak{t})$ and \mathfrak{t}' are maximal toral (= Cartan) subalgebras of \mathfrak{g} , contained in $\mathfrak{z}_\mathfrak{g}(\mathfrak{s}')$ and hence Cartan subalgebras of $\mathfrak{z}_\mathfrak{g}(\mathfrak{s}')$. It follows that $h_K(\mathfrak{t}_K)$ and \mathfrak{t}'_K are both split Cartan subalgebras of the reductive Lie algebra $\mathfrak{z}_\mathfrak{g}(\mathfrak{s}')_K$. From [**Bo2**, VIII.3.3] we get $\eta \in \operatorname{Aut}_e((\mathfrak{z}_\mathfrak{g}(\mathfrak{s}'))_K)$ with $\eta(h_K(\mathfrak{t}_K)) = \mathfrak{t}'_K$. Because η is elementary, firstly, it extends to an automorphism $\hat{\eta}$ of \mathfrak{g}_K , and secondly, it fixes \mathfrak{s}'_K . Set $F := \hat{\eta} \circ h_K$.

Combining this with 3.1.53 shows that the index of \mathfrak{g} depends, up to congruence, not on the choices made. Further:

THEOREM 3.1.57. — Let $f : \mathfrak{g} \xrightarrow{\sim} \mathfrak{g}'$ be an isomorphism of Lie algebras. Then for appropriate choices, there is a congruence from the respective index of \mathfrak{g} to that of \mathfrak{g}'

$$h: (X, R, \Delta, \Delta_0, t) \xrightarrow{\sim} (X', R', \Delta', \Delta'_0, t')$$

and an isomorphism $f_a : \mathfrak{g}_a \xrightarrow{\sim} \mathfrak{g}'_a$ of the anisotropic kernels, which are compatible in the sense that the induced map

$$h_{|X_a}: (X_a, R_0, \Delta_0, \Delta_0, t_{|X_a}) \longrightarrow (X'_a, R'_0, \Delta'_0, \Delta'_0, t'_{|X'_a})$$

is the congruence associated with $(f_a)_K := \mathrm{id} \otimes f_a : (\mathfrak{g}_a)_K \xrightarrow{\sim} (\mathfrak{g}'_a)_K$.

Proof. — If we choose $\mathfrak{s}, \mathfrak{t}, \Delta$ and a splitting field K for \mathfrak{t} as before, then $\mathfrak{s}' := f(\mathfrak{s})$ is a maximal split toral subalgebra of $\mathfrak{g}', \mathfrak{t}' := f(\mathfrak{t})$ is a maximal toral one containing \mathfrak{s}' . Call ψ as usual the inverse of the contragredient of $F := \mathrm{id} \otimes f : \mathfrak{g}_K \to \mathfrak{g}'_K$. With Falso ψ commutes with the (untwisted) Galois actions, so that $\Delta' := \psi(\Delta)$ is a Γ -basis (of $R' \subset X'$, defined with respect to \mathfrak{t}'_K). A congruence between the indices defined with respect to these choices is just given by ψ (no twisting with a Weyl group element needed), and taking as f_a of course the restriction to \mathfrak{g}_a of f, and Δ_0 as Γ -basis of R_0 , the compatibility is obvious.

The converse of this is the "isomorphism theorem" from this chapter's title. We will state and prove it in Section 3.3. Before doing this, let us look at some special cases and examples.

3.2. Special cases and examples

In this section, we will consider opposite extreme cases of what our invariants can be. [Spr3, 16.2.2] whose proof, however, needs slight corrections. They will be very useful for the classification. Keep the notation of the previous sections, i.e. choose $\mathfrak{s}, \mathfrak{t}, \Delta$, etc. as before.

3.2.1. The anisotropic case

LEMMA 3.2.1. — The following are equivalent:

- i. $\Delta_0 = \Delta$,
- ii. $R_0 = R$,
- iii. $X_0 = X$,
- iv. $\overline{R} = \emptyset$.
- v. $\mathfrak{s} = \{0\},\$
- vi. $\mathfrak{g}_a = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{g},$
- vii. every element of \mathfrak{g} is semisimple,

viii. g contains no non-zero nilpotent element,

ix. \mathfrak{g} contains no parabolic subalgebra distinct from \mathfrak{g} .

Proof. — Equivalences and implication $i \Leftrightarrow ii \Leftrightarrow iii \Leftrightarrow iv \Leftrightarrow v \Rightarrow vi$ are clear. Algebra \mathfrak{g} being semisimple implies $v \Leftarrow vi$. We have vii \Leftrightarrow viii from the Jordan decomposition, and by 3.1.2, viii \Rightarrow iv. Remark 3.1.29 gives vi \Rightarrow vii (alternatively, $v \Rightarrow$ viii by Jacobson-Morozov).

Finally, the equivalence viii \Leftrightarrow ix is exercise 6.a of [**Bo2**, VIII.10]. Direction " \Leftarrow " follows by contraposition from [**Bo2**, VIII.10, Cor. 2] applied to $\mathbf{n} = k \cdot X$ for a non-zero nilpotent $X \in \mathfrak{g}$, because the radical of \mathfrak{g} and hence the set called " $\mathfrak{n}_{\mathfrak{g}}(\mathfrak{g})$ " in *loc. cit.* (defined in the same paragraph) is zero. For the converse, assume there is a parabolic subalgebra $\mathfrak{p} \subsetneq \mathfrak{g}$. Then by [**Bo2**, VIII.3.4, Prop. 13] the largest nilpotent ideal of $\mathfrak{p}_{\bar{k}}$ is a non-zero sum of root spaces in $\mathfrak{g}_{\bar{k}}$, which (cf. *ibid.*. Prop. 9) consists of nilpotent elements of $\mathfrak{g}_{\bar{k}}$. By [**Bo2**, I.4.5], this ideal is the scalar extension of the largest nilpotent ideal in \mathfrak{p} . In particular, under the usual identifications, its intersection with \mathfrak{p} is nonzero, and consists of elements which are nilpotent in \mathfrak{g} .

DEFINITION 3.2.2. — If these equivalent conditions are satisfied, \mathfrak{g} is called *anisotropic*.

Parts vii to ix (and v, saying that $\{0\}$ is the unique split toral subalgebra in \mathfrak{g}) show directly that this is independent from the choices of $\mathfrak{s}, \mathfrak{t}$ etc. One readily sees e.g. from 3.1.29 that the anisotropic kernel of a semisimple \mathfrak{g} is indeed anisotropic, and of course an anisotropic Lie algebra is its own anisotropic kernel. Also, part viii combined with lemma 2.1.8 gives

COROLLARY 3.2.3. — A semisimple subalgebra of an anisotropic Lie algebra is anisotropic.

REMARK 3.2.4. — The terminology comes from anisotropic quadratic forms; the connection will occur in the next example and become striking in 4.5.14; see also [**Sat3**, I.4.4, Example 1], [**BT1**, 4.24]. [**Sat3**] and [**We1**] use the word *compact*. This is common usage in the theory of real Lie algebras (cf. [**Hel**, Chap. III, X], [**OVi**], [**Oni**], [**Bo2**, IX]) and ultimately stems from the fact that over \mathbb{R} , these Lie algebras correspond to compact Lie groups $Int(\mathfrak{g})$. For the connection between anisotropy and compactness in algebraic groups over local fields cf. [**BT1**, §9] and [**Sat3**, I.4.4.4].

EXAMPLE 3.2.5 (Anisotropic form of A_1). — Let K|k be a quadratic extension, say K = k(y) with $y^2 =: a \in k$. Denote the non-trivial element of $\Gamma = \text{Gal}(K|k)$ by σ :

$$\sigma(r+sy) = r - sy \text{ for } r, s \in k.$$

Assume further that there is an element $b \in k$ which is not a norm of the extension K|k. Consider, inside $R_{K|k} \mathfrak{sl}_2(K)$, the 3-dimensional subalgebra \mathfrak{g} given by all elements

$$\begin{pmatrix} ry & s+ty \\ b(s-ty) & -ry \end{pmatrix}$$

MÉMOIRES DE LA SMF 151

48

with $r, s, t \in k$. One easily sees that \mathfrak{g}_K identifies with $\mathfrak{sl}_2(K)$ (in particular, \mathfrak{g} is simple), a general element of which we denote as

$$x = \begin{pmatrix} x_{11} & x_{12} \\ x_{21} & x_{22} \end{pmatrix}$$

subject to $x_{11} = -x_{22}$. We claim that \mathfrak{g} is anisotropic, for which we verify 3.2.1.viii. Assume that

$$x = \begin{pmatrix} ry & s + ty \\ b(s - ty) & -ry \end{pmatrix}$$

is a nilpotent element of \mathfrak{g} . Then $\mathrm{ad}_{\mathfrak{g}}(x)$ is nilpotent and so is

5

$$\operatorname{ad}_{\mathfrak{g}_K}(1\otimes x) = \operatorname{id}\otimes \operatorname{ad}_{\mathfrak{g}}(x),$$

hence (cf. [**Bo2**, I.6.3, Thm 3]) so is $1 \otimes x = x$ viewed as element of $\mathfrak{sl}_2(K)$. This implies by definition (cf. [**Bo2**, I.6.3, Def. 3]) that the matrix x itself is nilpotent, so its characteristic polynomial must be T^2 , that is, $-r^2y^2 - b(s^2 - t^2y^2) = 0$ which is equivalent to

$$(s = t = 0 = r)$$
 or $\left(b = \frac{-r^2 y^2}{s^2 - t^2 y^2} = N_{K|k} \left(\frac{ry}{s + ty}\right)\right)$

The second case was excluded, so x = 0 and our claim is proved.

NB. — Our choice of a, b is made so that the quadratic form $X_0^2 - aX_1^2 - bX_2^2 + abX_3^2$ (or equivalently, the quadratic form $X_3^2 - bX_1^2 - aX_2^2$) is anisotropic over k. This in turn is equivalent to the quaternion algebra

$$D := \left(\frac{a, b}{k}\right)$$

being a division algebra, cf. [**Pie**, 1.6]. In fact, our \mathfrak{g} is the derived algebra (= semisimple part) of the associative k-algebra D viewed as Lie algebra with commutator bracket. For $K|k = \mathbb{C}|\mathbb{R}$ and a = b = -1, we have the Hamilton quaternions as D, and \mathfrak{g} is the compact real form of $\mathfrak{sl}_2(\mathbb{C})$. For k a \mathfrak{p} -adic field, one can take as K|k the unique unramified extension of degree 2 and as b a uniformiser of k. For more in this direction, see Section 4.4 and compare example (a) in Remark 4.5.20.i.

As \mathfrak{g} is anisotropic, the unique split toral subalgebra is $\mathfrak{s} = \{0\}$. Any onedimensional subspace is a maximal toral subalgebra, but to have K as splitting field, we choose

$$\mathfrak{t} = \left\{ \begin{pmatrix} ry & 0\\ 0 & -ry \end{pmatrix} : r \in k \right\}$$

 \mathfrak{g}_K has the well-known root $\alpha(x) = x_{11} - x_{22}$ (= $2x_{11}$) with respect to the split maximal toral subalgebra

$$\mathfrak{t}_K = \left\{ \begin{pmatrix} x_{11} & 0\\ 0 & x_{22} \end{pmatrix} : x_{11} = -x_{22} \right\}.$$

We have $\rho(\alpha) = \alpha_{|\mathfrak{s}_K} = 0$ and the root spaces

$$(\mathfrak{g}_K)_{\alpha} = \begin{pmatrix} 0 & * \\ 0 & 0 \end{pmatrix}, (\mathfrak{g}_K)_{-\alpha} = \begin{pmatrix} 0 & 0 \\ * & 0 \end{pmatrix}.$$

Let us look at the Galois action on a root space: Writing

$$\begin{pmatrix} 0 & x \\ 0 & 0 \end{pmatrix} = \frac{x}{2} \cdot \underbrace{\begin{pmatrix} 0 & 1 \\ b & 0 \end{pmatrix}}_{\in \mathfrak{g}} + \frac{x}{2y} \cdot \underbrace{\begin{pmatrix} 0 & y \\ -by & 0 \end{pmatrix}}_{\in \mathfrak{g}}$$

we see with $\sigma(y) = -y$ that

$$\sigma\left(\begin{pmatrix}0 & x\\0 & 0\end{pmatrix}\right) = \frac{\sigma(x)}{2} \cdot \begin{pmatrix}0 & 1\\b & 0\end{pmatrix} - \frac{\sigma(x)}{2y} \cdot \begin{pmatrix}0 & y\\-by & 0\end{pmatrix} = \begin{pmatrix}0 & 0\\b\sigma(x) & 0\end{pmatrix}$$

so that $\sigma((\mathfrak{g}_K)_{\alpha}) = (\mathfrak{g}_K)_{-\alpha}$, hence ${}^{\sigma}\alpha = -\alpha$. Further:

i. $R = R_0 = \{\pm \alpha\}$ (type A_1)

ii. $\Delta = \Delta_0$ for both possible bases (automatically Γ -bases) $\{\alpha\}$ and $\{-\alpha\}$

- iii. $X = X_0 = X_a = \mathbb{Z}\alpha$
- iv. $\{0\} = \mathfrak{s} = \mathfrak{Z} \subsetneq \mathfrak{t} \subsetneq \mathfrak{g}_a = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{g}$
- v. $W(R) = W(R)_{\Gamma} = W_0 = \{\pm id\}$
- vi. $w_{\sigma} = -\operatorname{id}$ so that $t(\sigma) = \operatorname{id}$.

3.2.2. The quasi-split case

LEMMA 3.2.6. — The following are equivalent:

- i. $\Delta_0 = \emptyset$.
- ii. $R_0 = \emptyset$.
- iii. $X_a = \{0\}.$
- iv. $\mathfrak{g}_a = \{0\}.$
- v. $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{t}.$
- vi. \mathfrak{g} contains a Borel subalgebra.

Proof. — i \Leftrightarrow ii \Leftrightarrow iii \Leftrightarrow iv \Leftrightarrow v amends 3.1.32. Now if these are satisfied,

$$\mathfrak{g} = \underbrace{\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})}_{\mathfrak{t}} \oplus \bigoplus_{\lambda \in \overline{R}} \mathfrak{g}_{\lambda}$$

and with 3.1.20 one sees that for a system of positive roots \overline{R}^+ in \overline{R} ,

$$\mathfrak{t} \oplus \bigoplus_{\lambda \in \overline{R}^+} \mathfrak{g}_{\lambda}$$

is a Borel subalgebra of \mathfrak{g} . Conversely, assume there is a Borel subalgebra $\mathfrak{b} \subset \mathfrak{g}$. Then its largest nilpotent ideal \mathfrak{N} has \mathfrak{b} as normaliser in \mathfrak{g} ; this follows from stability of the normaliser and the largest nilpotent ideal (cf. [**Bo2**, I.5.5]) under scalar extension.

Hence \mathfrak{N} is the intersection with \mathfrak{g} (or the set of all $\operatorname{Gal}(\bar{k}|k)$ -invariants) of $\mathfrak{N}_{\bar{k}} = \operatorname{sum}$ of all positive root spaces with respect to some toral subalgebra and order of roots. Certainly \mathfrak{N} is a nil subalgebra in \mathfrak{g} , and if it were not maximal nil in \mathfrak{g} , by [**Bo2**, VIII.10, Cor. 2] there would exist a parabolic subalgebra \mathfrak{q} in \mathfrak{g} whose radical strictly contains \mathfrak{N} ; so that the radical of $\mathfrak{q}_{\bar{k}}$ would strictly contain $\mathfrak{N}_{\bar{k}}$, which is absurd by a dimension comparison. Now if $\mathfrak{s}, \mathfrak{t}, K$ are chosen as usual, by step 4 of the proof of 3.1.16 we can assume that $\mathfrak{s} \subseteq \mathfrak{t} \subseteq \mathfrak{zg}(\mathfrak{s}) \subseteq \mathfrak{b}$ and hence \mathfrak{b}_K must be of the form $\mathfrak{t}_K \oplus \bigoplus_{\alpha \in P} (\mathfrak{g}_K)_{\alpha}$ for a system of positive roots $P \subset R$. Since on the other hand, all $(\mathfrak{g}_K)_{\alpha}$ for $\alpha \in R_0$ are in \mathfrak{b}_K , necessarily $R_0 = \emptyset$.

DEFINITION 3.2.7. — If these equivalent conditions are satisfied, \mathfrak{g} is called *quasi-split*.

REMARK 3.2.8. — The property of being *split* is stronger, as the next example shows. Characterisations of split Lie algebras will be given in 3.2.12.

EXAMPLE 3.2.9 (Quasi-split form of $A_2^{(12)}$). — Let K|k be a quadratic extension, say K = k(y) with $y^2 \in k$. Denote the non-trivial element of $\Gamma = \text{Gal}(K|k)$ by σ :

$$\sigma(a+by) = a - by \quad \text{for } a, b \in k$$

Consider, inside $R_{K|k} \mathfrak{sl}_3(K)$, the 8-dimensional subalgebra \mathfrak{g} given by all elements

$$\begin{pmatrix} a+by & c+dy & ey \\ f+gy & -2by & -c+dy \\ hy & -f+gy & -a+by \end{pmatrix}$$

with $a, b, c, d, e, f, g, h \in k$. One easily sees that \mathfrak{g}_K identifies with $\mathfrak{sl}_3(K)$, a general element of which we denote as

$$x = \begin{pmatrix} x_{11} & x_{12} & x_{13} \\ x_{21} & x_{22} & x_{23} \\ x_{31} & x_{32} & x_{33} \end{pmatrix}$$

subject to $x_{11} + x_{22} + x_{33} = 0$. We claim that

$$\mathfrak{s} := \left\{ \begin{pmatrix} a & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -a \end{pmatrix} : a \in k \right\} \quad \text{and} \quad \mathfrak{t} := \left\{ \begin{pmatrix} a + by & 0 & 0 \\ 0 & -2by & 0 \\ 0 & 0 & -a + by \end{pmatrix} : a, b \in k \right\}$$

are maximal split toral and maximal toral subalgebras of \mathfrak{g} , respectively. Indeed,

$$\mathfrak{t}_{K} = \left\{ \begin{pmatrix} x_{11} & 0 & 0 \\ 0 & x_{22} & 0 \\ 0 & 0 & x_{33} \end{pmatrix} : x_{11} + x_{22} + x_{33} = 0 \right\}.$$

^{12.} This is essentially the example from [Sel2, p. 13], generalised from $\mathbb{C}|\mathbb{R}$ and with slightly different normalisations.

is known to be maximal toral in \mathfrak{g}_K , so \mathfrak{t} is maximal toral in \mathfrak{g} by the proof of 3.1.7. On the other hand, one sees that \mathfrak{s} is split toral and that $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{t}$, so any split toral subalgebra containing \mathfrak{s} must be in \mathfrak{t} . But \mathfrak{t} itself is not split, and $\dim_k(\mathfrak{s}) = 1 < 2 = \dim_k(\mathfrak{t})$ proves our claim.

 \mathfrak{g}_K has the well-known roots $\alpha_1(x) = x_{11} - x_{22}$, $\alpha_2(x) = x_{22} - x_{33}$ with respect to \mathfrak{t}_K . We have $\rho(\alpha_1) = \alpha_{1|\mathfrak{s}_K} = \alpha_{2|\mathfrak{s}_K} = \rho(\alpha_2) =: \lambda_K$. Further:

- i. $R = \{\pm \alpha_1, \pm \alpha_2, \pm (\alpha_1 + \alpha_2)\}$ (type A_2),
- ii. $\overline{R} = \{\pm \lambda_K, \pm 2\lambda_K\}$ (type BC_1),
- iii. $R_0 = \Delta_0 = \emptyset$,
- iv. $X = \mathbb{Z}\alpha_1 + \mathbb{Z}\alpha_2$,
- v. $X_0 = \mathbb{Z}(\alpha_1 \alpha_2)$ (so indeed $\rho(\alpha_1) = \rho(\alpha_2)$ under $\rho: X \twoheadrightarrow X/X_0$),
- vi. $X_a = \{0\},\$
- vii. $\{0\} = \mathfrak{g}_a \subsetneq \mathfrak{s} \subsetneq \mathfrak{Z} = \mathfrak{t} = \mathfrak{z}_\mathfrak{g}(\mathfrak{s}),$
- viii. ${}^{\sigma}\alpha_1 = \alpha_2, \; {}^{\sigma}\alpha_2 = \alpha_1,$
- ix. the two $\Gamma\text{-bases}$ are $\Delta=\{\alpha_1,\alpha_2\}$ and $\Delta'=\{-\alpha_1,-\alpha_2\}$,
- x. $W(R)_{\Gamma} = \{ \mathrm{id}, s_{\alpha_1 + \alpha_2} \}, W_0 = \{ \mathrm{id} \},$
- xi. $w_{\sigma} = \text{id so that } t(\sigma) = \sigma_A$ is the transposition of α_1 and α_2 .

Root spaces and coroots "upstairs" are well known:

$$(\mathfrak{g}_{K})_{\alpha_{1}} = \begin{pmatrix} 0 * 0 \\ 0 & 0 \\ 0 & 0 \end{pmatrix}, \quad (\mathfrak{g}_{K})_{\alpha_{2}} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & * \\ 0 & 0 & 0 \end{pmatrix}, \quad (\mathfrak{g}_{K})_{\alpha_{1}+\alpha_{2}} = \begin{pmatrix} 0 & 0 & * \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix};$$
$$H_{\alpha_{1}} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -1 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad H_{\alpha_{2}} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & -1 \end{pmatrix}, \quad H_{\alpha_{1}+\alpha_{2}} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -1 \end{pmatrix},$$

and H_{α} is a K-basis of $[\mathfrak{g}_{\alpha}, \mathfrak{g}_{-\alpha}]$ for each $\alpha \in R$. But "below", for the rational root

$$\lambda: \mathfrak{s} \longrightarrow k, \quad \begin{pmatrix} a & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -a \end{pmatrix} \longmapsto a$$

we have

$$\mathfrak{g}_{\lambda} = \begin{pmatrix} 0 & c + dy & 0 \\ 0 & 0 & -c + dy \\ 0 & 0 & 0 \end{pmatrix}, \quad \mathfrak{g}_{-\lambda} = \begin{pmatrix} 0 & 0 & 0 \\ f + gy & 0 & 0 \\ 0 & -f + gy & 0 \end{pmatrix},$$
$$H_{\lambda} = -H_{-\lambda} = \begin{pmatrix} 2 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -2 \end{pmatrix}, \quad \mathfrak{g}_{2\lambda} = \begin{pmatrix} 0 & 0 & ey \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad H_{2\lambda} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & -1 \end{pmatrix}.$$

In particular $\dim_k(\mathfrak{g}_{\lambda}) = \dim_k(\mathfrak{g}_{-\lambda}) = 2$ as predicted by 3.1.13. In the same vein, we have

$$\left[\underbrace{\begin{pmatrix}0&y&0\\0&0&y\\0&0&0\end{pmatrix}}_{\in\mathfrak{g}_{\lambda}},\underbrace{\begin{pmatrix}0&0&0\\1&0&0\\0&-1&0\end{pmatrix}}_{\in\mathfrak{g}_{-\lambda}}\right] = \begin{pmatrix}y&0&0\\0&-2y&0\\0&0&y\end{pmatrix} \in \mathfrak{t} \setminus \mathfrak{s}$$

so in fact $[\mathfrak{g}_{\lambda}, \mathfrak{g}_{-\lambda}] = \mathfrak{t}$ has dimension 2 and is not generated by H_{λ} . Also note how the root spaces are "skew" with respect to each other: although $(\mathfrak{g}_{\lambda})_{K} = (\mathfrak{g}_{K})_{\alpha_{1}} \oplus (\mathfrak{g}_{K})_{\alpha_{2}}$, we have $\mathfrak{g}_{\lambda} \cap (\mathfrak{g}_{K})_{\alpha_{i}} = \{0\}$ for i = 1, 2.

Finally it might be worthwhile to see the Galois action on a root space explicitly: Writing

$$\begin{pmatrix} 0 & x & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix} = \frac{x}{2} \cdot \underbrace{\begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & -1 \\ 0 & 0 & 0 \end{pmatrix}}_{\in \mathfrak{g}} + \frac{x}{2y} \cdot \underbrace{\begin{pmatrix} 0 & y & 0 \\ 0 & 0 & y \\ 0 & 0 & 0 \end{pmatrix}}_{\in \mathfrak{g}}$$

we see with $\sigma(y) = -y$ that

$$\sigma\left(\!\begin{pmatrix} 0 & x & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}\!\right) = \frac{\sigma(x)}{2} \cdot \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & -1 \\ 0 & 0 & 0 \end{pmatrix} - \frac{\sigma(x)}{2y} \cdot \begin{pmatrix} 0 & y & 0 \\ 0 & 0 & y \\ 0 & 0 & 0 \end{pmatrix} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -\sigma(x) \\ 0 & 0 & 0 \end{pmatrix}$$

so that indeed $\sigma((\mathfrak{g}_K)_{\alpha_1}) = (\mathfrak{g}_K)_{\alpha_2}$.

REMARK 3.2.10. — This example can be generalised to construct non-split quasisplit Lie algebras of type A_n for arbitrary $n \ge 2$. (For n = 1, the construction degenerates to the split $\mathfrak{sl}_2(k)$.) Namely, let K = k(y) be as above, d = n + 1 and consider inside $R_{K|k}\mathfrak{sl}_d(K)$ those matrices $(x_{i,j})$ that satisfy $x_{i,j} = -\sigma(x_{d+1-j,d+1-i})$; that is, those traceless $d \times d$ matrices over K such that each entry is the negative conjugate of the one mirrored at the secondary diagonal (in particular, the entries on the secondary diagonal are k-multiples of y); or in yet other words, the traceless $d \times d$ matrices $(x_{i,j})$ over K satisfying $(x_{i,j}) \cdot H + H \cdot t(\sigma(x_{i,j})) = 0$ where H is the $d \times d$ matrix with entries 1 on the secondary diagonal and 0 else. A maximal split toral subalgebra is

$$:= \{ \operatorname{diag}(x_{1,1}, \dots, x_{n/2, n/2}, 0, -x_{n/2, n/2}, \dots, -x_{1,1}) : x_{i,i} \in k \}$$

 $\mathfrak{s} := \left\{ \operatorname{diag}(x_{1,1}, \dots, x_{d/2, d/2}, -x_{d/2, d/2}, \dots, -x_{1, 1}) : x_{i, i} \in k \right\}$

according to whether n is odd or even. One calculates that for odd n, the rational root system \overline{R} is of type $C_{d/2}$, whereas for even n, it is of type $BC_{n/2}$ (we will show this with an easy method in 3.4.1). We have $w_{\sigma} = \text{id}$ so that the twisted and untwisted Galois action coincide, and σ transposes α_i and α_{n-i+1} , where α_i are the conventionally named roots of $\mathfrak{sl}_d(K)$.⁽¹³⁾

^{13.} Cf. [Bo2, VIII.13, exercise 16.a] and 4.5.22 for much more generality.

REMARK 3.2.11. — For different choices of the quadratic extension K|k, the above gives non-isomorphic Lie algebras. In the index, the difference is in the kernel of the twisted Galois action $t : \text{Gal}(\bar{k}|k) \to \text{Aut}(R, \Delta)$, whereas the image of t is always the same. We call the fixed field of the kernel of t the "fixed field of t" for short.

3.2.3. The split case

LEMMA 3.2.12. — The following are equivalent:

- i. g is quasi-split, and the twisted Galois action is trivial (i.e. $t \equiv id$).
- ii. The non-twisted Galois action on X is trivial.
- iii. $X_0 = \{0\}.$
- iv. $\mathfrak{s} = \mathfrak{t}$.
- v. $R = \overline{R}$.
- vi. t is split toral.
- vii. $(\mathfrak{g}, \mathfrak{t})$ is a split semisimple Lie algebra in the sense of [**Bo2**, VIII.2].

Proof. — i \Rightarrow ii by 3.1.43.ii and 3.2.6.iii. The equivalence iii \Leftrightarrow iv follows from 3.1.27, and ii \Leftrightarrow iii as well as iv \Leftrightarrow v \Leftrightarrow vi \Leftrightarrow vii \Leftrightarrow i are clear.

REMARK 3.2.13 (Splitting fields). — As a matter of convention, we call a *splitting* field for our semisimple \mathfrak{g} a field K such that there is a maximal split toral \mathfrak{s} in \mathfrak{g} , and a maximal toral \mathfrak{t} containing \mathfrak{s} , such that K is a splitting field for \mathfrak{t} . Caveat: this does not mean that any toral subalgebra, nor even every other \mathfrak{t}' of this kind, is split by K.

Any splitting field for \mathfrak{g} must contain the fixed field of t. So if this fixed field is a splitting field itself – e.g. for quasi-split \mathfrak{g} – it is the unique *minimal* splitting field, meaning that every splitting field for any choice $\mathfrak{t} \supseteq \mathfrak{s}$ has to contain it. Not all Lie algebras have such a minimal splitting field. In fact different non-split toral subalgebras \mathfrak{t} , even if chosen with respect to the same \mathfrak{s} , might even have linearly disjoint splitting extensions. For example in 5.2.2 we will see Lie algebras for which every extension of a given degree d is a splitting field (for some \mathfrak{t} chosen with respect to $\mathfrak{s} = 0$ there).

3.3. The isomorphism theorem

In this section we prove the converse of 3.1.57. In this sense, the description of the isomorphism class of a semisimple Lie algebra is reduced to the description of its index and anisotropic kernel. For *split* semisimple Lie algebras, the statement reduces to a variant of the classical uniqueness theorem as formulated by Chevalley and others, see e.g. [**Bo2**, VIII.4.4]. In fact we will use this result for a splitting extension, and then derive the general version by Galois descent, including the notorious Hilbert 90. Our proof follows [**Sat2**] and [**Sat3**, II.2.4], i.e. translates it to Lie algebras.

MÉMOIRES DE LA SMF 151

Let $\mathfrak{g}, \mathfrak{g}'$ be two semisimple Lie algebras over k. Fix choices $\mathfrak{s}, \mathfrak{s}', \ldots$ as before, including a common finite Galois splitting field K for both \mathfrak{t} and \mathfrak{t}' . With respect to these choices, we have the indices $(X, R, \Delta, \Delta_0, t)$ and $(X', R', \Delta', \Delta'_0, t')$, and the anisotropic kernels \mathfrak{g}_a and \mathfrak{g}'_a .

Assume that there is a congruence

$$h: (X, R, \Delta, \Delta_0, t) \longrightarrow (X', R', \Delta', \Delta'_0, t')$$

and an isomorphism

$$f_a:\mathfrak{g}_a\xrightarrow{\sim}\mathfrak{g}'_a$$

which are compatible in the sense that the induced map

$$h_{|X_a}: (X_a, R_0, \Delta_0, \Delta_0, t_{|X_a}) \longrightarrow (X_a', R_0', \Delta_0', \Delta_0', t_{|X_a'}')$$

is the congruence associated with $(f_a)_K := \mathrm{id} \otimes f : (\mathfrak{g}_a)_K \xrightarrow{\sim} (\mathfrak{g}'_a)_K$.

THEOREM 3.3.1 (Satake, Tits, Weisfeiler). — Under the above assumptions, there is an isomorphism $f : \mathfrak{g} \to \mathfrak{g}'$ inducing f_a and such that h is associated with

$$f_K := \mathrm{id} \otimes f : \mathfrak{g}_K \longrightarrow \mathfrak{g}'_K.$$

A special case is:

COROLLARY 3.3.2. — If \mathfrak{g} is quasi-split, its isomorphism class is uniquely determined by its index.

The proof of the theorem will take up the rest of this section.

3.3.1. Requisites. — For each $\alpha \in R$, choose a basis of $(\mathfrak{g}_K)_{\alpha}$, i.e. a non-zero element $e_{\alpha} \in (\mathfrak{g}_K)_{\alpha}$, in such a way that $[e_{\alpha}, e_{-\alpha}] = H_{\alpha}$ (this is possible by 3.1.9). Make an analogous choice of $e'_{\alpha'}$ for $\alpha' \in R'$. These choices remain fixed throughout this section. By 3.1.22, for each $(\sigma, \alpha) \in \Gamma \times R$ there is a unique $\xi_{\sigma,\alpha} \in K^*$ defined by the relation:

(11)
$$\sigma(e_{\alpha}) = \xi_{\sigma,\alpha} \cdot e_{\sigma_A(\alpha)}$$

Now for $\sigma, \tau \in \Gamma$ we have $\sigma(\tau(e_{\alpha})) = \sigma(\xi_{\tau,\alpha}) \cdot \sigma(e_{\tau_A(\alpha)}) = \sigma(\xi_{\tau,\alpha}) \cdot \xi_{\sigma,\tau_A(\alpha)} \cdot e_{\sigma_A\tau_A(\alpha)}$, so that:

(12)
$$\xi_{\sigma\tau,\alpha} = \sigma(\xi_{\tau,\alpha}) \cdot \xi_{\sigma,\tau_A(\alpha)}$$

(So for each $\alpha \in R$, the map $\sigma \mapsto \xi_{\sigma,\alpha}$ is a cocycle from the subgroup $\Gamma_{\alpha} := \{\sigma \in \Gamma : \sigma(\alpha) = \alpha\}$ to K^* , but we will not need this.) Define $\xi'_{\sigma,\alpha'} \in K^*$ analogously.

Next, suppose we are given an isomorphism $F : \mathfrak{g}_K \xrightarrow{\sim} \mathfrak{g}'_K$ such that $F(\mathfrak{t}_K) = \mathfrak{t}'_K$ (write $F : (\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ for short). We get the isomorphism

$$\psi: X \xrightarrow{\sim} X', \quad \chi \longmapsto \chi \circ \left((F_{|\mathfrak{t}_K})^{-1} \right)$$

with $\psi(R) = R'$. This in turn gives, for $\alpha \in R$, unique $\eta_{\alpha} \in K^*$ defined by

(13)
$$F(e_{\alpha}) = \eta_{\alpha} \cdot e'_{\psi(\alpha)}.$$

From the root space decomposition it is clear that if two isomorphisms F and G give rise to the same tuple $(\psi, (\eta_{\alpha})_{\alpha \in R})$, we have F = G, so F is uniquely determined by this tuple. Write

$$F \longleftrightarrow (\psi, (\eta_{\alpha})_{\alpha \in R}).$$

We will later see some sufficient conditions for a tuple to occur, i.e. to arise from an isomorphism. The proof of the following remark contains a necessary one:

REMARK 3.3.3. — For any basis B of the root system R, an isomorphism F as above is already determined by ψ and $(\eta_{\alpha})_{\alpha \in B}$.

Proof. — The chosen e_{α} give rise to structure constants $N_{\alpha,\beta} \in K^*$ defined by

(14)
$$[e_{\alpha}, e_{\beta}] = N_{\alpha,\beta} \cdot e_{\alpha+\beta} \text{ for } \alpha, \beta \in R \text{ with } \alpha + \beta \in R$$

also remember $[e_{\alpha}, e_{-\alpha}] = H_{\alpha}$. Analogously, define $N'_{\alpha',\beta'}$. It is then not hard to show that F commuting with the Lie brackets and $F(H_{\alpha}) = H_{\psi(\alpha)}$ implies

$$\eta_{\alpha+\beta} = \eta_{\alpha} \cdot \eta_{\beta} \cdot \frac{N'_{\psi(\alpha),\psi(\beta)}}{N_{\alpha,\beta}}, \quad \eta_{-\alpha} = \eta_{\alpha}^{-1}$$

for $\alpha, \beta \in R$ with $\alpha + \beta \in R$, from which the claim follows.

For $\sigma \in \Gamma$ and an isomorphism F as above, set ${}^{\sigma}F := \sigma \circ F \circ \sigma^{-1}$ – the σ 's meaning the k-automorphisms of \mathfrak{g}'_K and \mathfrak{g}_K – which is again an isomorphism $(\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$. The next lemma shows how: i) this Galois action and ii) composition translate to the tuple. For $\phi \in \operatorname{Hom}_{\mathbb{Z}}(X, X')$, we define ${}^{\sigma}\phi := \sigma_{A'} \phi \sigma_A^{-1}$ with the $\sigma_A \in A(R), \sigma_{A'} \in A(R')$ again meaning the elements defined by the Galois actions on X and X'.

Lemma 3.3.4

i. For σ and F as above, we have

$${}^{\sigma}F\longleftrightarrow \left({}^{\sigma}\psi, \left(\sigma(\eta_{\sigma_{A}^{-1}(\alpha)})\cdot \frac{\xi_{\sigma,\psi(\sigma_{A}^{-1}(\alpha))}}{\xi_{\sigma,\sigma_{A}^{-1}(\alpha)}}\right)_{\alpha\in R}\right).$$

ii. For isomorphisms

$$(\mathfrak{g}_K,\mathfrak{t}_K) \xrightarrow{F} (\mathfrak{g}'_K,\mathfrak{t}'_K) \xrightarrow{F'} (\mathfrak{g}''_K,\mathfrak{t}''_K)$$

(with $X'', R'', (e''_{\alpha})_{\alpha \in R''}$, etc.) we have

$$F' \circ F \longleftrightarrow (\psi' \circ \psi, (\eta'_{\psi(\alpha)} \cdot \eta_{\alpha})_{\alpha \in R})$$

Proof. — Let $\alpha \in R$. For $\chi \in X$ we have $({}^{\sigma}\psi)(\chi) = {}^{\sigma}(\psi({}^{\sigma^{-1}}\chi)) = {}^{\sigma}(\psi(\sigma^{-1}\circ\chi\circ\sigma))$ $= {}^{\sigma}(\sigma^{-1}\circ\chi\circ\sigma\circ(F_{|\mathfrak{t}_{K}})^{-1}) = \chi\circ\sigma\circ(F_{|\mathfrak{t}_{K}})^{-1}\circ\sigma^{-1}$ $= \chi\circ({}^{\sigma}F_{|\mathfrak{t}_{K}})^{-1}.$

where the internal σ 's with which the maps are composed are the k-automorphisms of $\mathfrak{t}_K, \mathfrak{t}'_K$ and K. Further,

and $\sigma(\xi_{\sigma^{-1},\alpha})$ is the inverse of $\xi_{\sigma,\sigma^{-1}(\alpha)}$ by (12). (Note $e'_{\sigma_{A'}\psi\sigma_A^{-1}(\alpha)} = e'_{(\sigma\psi)(\alpha)}$.) This shows i. For ii., just remark

$$F'(F(e_{\alpha})) = F'(\eta_{\alpha} \ e'_{\psi(\alpha)}) = \eta_{\alpha} \ F'(e'_{\psi(\alpha)}) = \eta_{\alpha} \ \eta'_{\psi(\alpha)} \ e''_{\psi'(\psi(\alpha))}.$$

Set $T_X := \operatorname{Hom}_{\mathbb{Z}}(X, K^*)$, see [**Bo2**, VIII.5..2] – our X is the root lattice called Q(R) there. There is an embedding $i : T_X \hookrightarrow \operatorname{Aut}(\mathfrak{g}_K, \mathfrak{t}_K)$ called "f" in *loc. cit*.: for $\Phi \in T_X$ and any weight $\alpha \in R \cup \{0\}$, the restriction of $i(\Phi)$ to the weight space $(\mathfrak{g}_K)_{\alpha}$ is given by scalar multiplication with $\Phi(\alpha)$. In fact our tuples, in the case of $F \in \operatorname{Aut}(\mathfrak{g}_K, \mathfrak{t}_K)$, are explicit versions of the exact sequence

$$1 \to T_X \longrightarrow \operatorname{Aut}(\mathfrak{g}_K, \mathfrak{t}_K) \longrightarrow A(R) \to 1,$$

with respect to the same chosen basis $(e_{\alpha})_{\alpha \in R}$ for domain and target. (Vaguely said, T_X is a multiplicative ersatz torus, and in the group setting, $i(\Phi)$ would be the inner automorphism given by conjugation with Φ .)

LEMMA 3.3.5. — Let $F \in \operatorname{Aut}(\mathfrak{g}_K, \mathfrak{t}_K)$. Then for $\Phi \in T_X$, we have $F = i(\Phi)$ if and only if $F \leftrightarrow (\operatorname{id}, (\Phi(\alpha))_{\alpha \in R})$. In particular, given any basis B of R and arbitrary $\eta_{\alpha} \in K^*$ for $\alpha \in B$, there are unique η_{α} for $\alpha \in R \setminus B$ and a unique $F \in \operatorname{Aut}(\mathfrak{g}_K, \mathfrak{t}_K)$ such that $F \leftrightarrow (\operatorname{id}, (\eta_{\alpha})_{\alpha \in R})$. This F is in $i(T_X)$.

Proof. — That $i(\Phi) \leftrightarrow (\mathrm{id}, (\Phi(\alpha))_{\alpha \in R})$ for $\Phi \in T_X$ is clear; and we had already remarked that F is uniquely determined by its tuple. For the second assertion, B is a \mathbb{Z} -basis of X, so there is a unique $\Phi \in T_X$ with $\Phi(\alpha) = \eta_\alpha$ for $\alpha \in B$. Then by the first part, $F := i(\Phi)$ has a tuple of the prescribed form, with $\Phi(\alpha) = \eta_\alpha$ for all $\alpha \in R$. The uniqueness then follows from 3.3.3.

To end this subsection, we explicitly remark the following:

PROPOSITION 3.3.6. — $F : (\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ is of the form $\mathrm{id} \otimes f$ with $f : (\mathfrak{g}, \mathfrak{t}) \xrightarrow{\sim} (\mathfrak{g}', \mathfrak{t}')$ if and only if $\sigma F = F$ for all $\sigma \in \Gamma$.

Proof. — ${}^{\sigma}F = F$ for all $\sigma \in \Gamma$ means that F is Γ -equivariant, which implies that by restriction it induces an isomorphism (of Lie algebras over k)

$$f: \mathfrak{g} = (\mathfrak{g}_K)^{\Gamma} \simeq (\mathfrak{g}'_K)^{\Gamma} = \mathfrak{g}'.$$

with $f(\mathfrak{t}) = f((\mathfrak{t}_K)^{\Gamma}) = f(\mathfrak{t}_K)^{\Gamma} = (\mathfrak{t}'_K)^{\Gamma} = \mathfrak{t}'$. We naturally identify F with $\mathrm{id} \otimes f$. The other direction is trivial.

3.3.2. Construction of the isomorphism. — Keeping assumptions and notations, we now construct an isomorphism from \mathfrak{g}_K to \mathfrak{g}'_K and then improve it until we can let it descend to an isomorphism f as in the theorem. We start with a refined version of the isomorphism theorem for split semisimple Lie algebras.

PROPOSITION 3.3.7. — Let $\psi : X \xrightarrow{\sim} X'$ be an isomorphism with $\psi(R) = R'$. Let B be any basis of R, and for any $\alpha \in B$, let $\eta_{\alpha} \in K^*$. Then there are $\eta_{\alpha} \in K^*$ for $\alpha \in R \setminus B$ and an isomorphism $F : (\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ such that $F \leftrightarrow (\psi, (\eta_{\alpha})_{\alpha \in R})$.

Proof. — By the mentioned theorem – see e.g. $[\mathbf{Bo2}, \text{VIII.4.4}, \text{ hm 2.i}]$ – there is an isomorphism $F_1 : (\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ such that $\psi(\chi) = \chi \circ F_{1|\mathfrak{t}_K}^{-1}$ for all $\chi \in X$. Consequently, we have $F_1 \leftrightarrow (\psi, (\theta_\alpha)_{\alpha \in R})$ for certain $\theta_\alpha \in K^*$. There is a unique $\Phi \in T_X$ with $\Phi(\alpha) = \theta_\alpha^{-1} \eta_\alpha$ for all $\alpha \in B$. Setting $\eta_\alpha := \Phi(\alpha) \theta_\alpha$ for all $\alpha \in R$ and using 3.3.5 and 3.3.4.ii, we see that $F := F_1 \circ i(\Phi)$ is an isomorphism $(\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ satisfying $F \leftrightarrow (\psi, (\eta_\alpha)_{\alpha \in R})$.

So in order to get a good F, we need ψ and η_{α} with good properties. For this, we will now make use of the extra data we have got.

Let the isomorphism $(f_a)_K : (\mathfrak{g}_a)_K \xrightarrow{\sim} (\mathfrak{g}'_a)_K$ (sending $(\mathfrak{t}_a)_K$ to $(\mathfrak{t}'_a)_K$) correspond to the tuple $(\psi_0, (\eta^0_\alpha)_{\alpha \in R_0})$.

PROPOSITION 3.3.8. — There is an isomorphism $F : (\mathfrak{g}_K, \mathfrak{t}_K) \xrightarrow{\sim} (\mathfrak{g}'_K, \mathfrak{t}'_K)$ with $F \leftrightarrow (\psi, (\eta_\alpha)_{\alpha \in \mathbb{R}})$, satisfying the following properties:

i. (a) $\psi_{|X_a} = \psi_0;$

58

(b) $\eta_{\alpha} = \eta_{\alpha}^{0}$ for $\alpha \in R_{0}$;

ii. there is $w' \in W'_{\Gamma}$ such that $w'\psi = h$;

- iii. (a) ${}^{\sigma}\psi = \psi$ for all $\sigma \in \Gamma$;
 - (b) For all $(\sigma, \alpha) \in \Gamma \times R$,

$$\xi_{\sigma,\psi(\alpha)}' = \xi_{\sigma,\alpha} \cdot \frac{\eta_{\sigma_A(\alpha)}}{\sigma(\eta_\alpha)} \cdot$$

Proof

First step. — By definition (3.1.55), the congruence associated with $(f_a)_K$ is given by the map $w' \psi_0$ for a unique $w' \in (W'_0)_{\Gamma} = W'_0$; by assumption, it is equal to $h_{|X_a}$. So define

$$\psi := w'^{-1}h : X \xrightarrow{\sim} X'$$

which certainly satisfies $\psi(R) = R'$, i.(a) and ii. We show that we also have iii.(a).

Let $\sigma \in \Gamma$. Remember that ${}^{\sigma}\psi$ was defined as $\sigma_{A'}\psi\sigma_A^{-1}$, where $\sigma_{A'}$ means the element in A(R') and σ_A the one in A(R) through which σ operates on X' and X. We naturally extend everything to the vector spaces V and V'. On these, we have A(R)- (respectively, A(R')-)invariant scalar products, hence a decomposition $V = V_a \oplus V_a^{\perp}$ such that V_a^{\perp} can A(R)-equivariantly be identified with V/V_a (and analogous for V'). Further, property ii in the definition 3.1.52 of a congruence shows that we have $h(V_a^{\perp}) = (V'_a)^{\perp}$.

Now it suffices to check σ -invariance for the restrictions $\psi_{|V_a|}$ and $\psi_{|V_a^{\perp}|}$. The first one is extended from ψ_0 , which is σ -invariant by 3.3.4.i because $(f_a)_K$ is.

From our congruence we know $t'(\sigma) = h \circ t(\sigma) \circ h^{-1}$; on the other hand, we had (see the proof of 3.1.43.i)

$$t(\sigma) = w_{\sigma}\sigma_A$$
 (resp. $t'(\sigma) = w'_{\sigma}\sigma_{A'}$)

with $w_{\sigma} \in W_0$, $w'_{\sigma} \in W'_0$. Since (by 3.1.34) W_0 acts trivially on V_a^{\perp} , and W'_0 trivially on $(V'_a)^{\perp}$, we get

$$\sigma_{A'} \circ h_{|V^{\perp}} = (h \circ \sigma_A)_{|V^{\perp}}$$

and, because w' was in W'_0 as well,

$$\sigma_{A'} \circ \psi_{|V_a^{\perp}} = (\psi \circ \sigma_A)_{|V_a^{\perp}}$$

which proves our claim.

Second step. — We define η_{α} for $\alpha \in \Delta$ so that i.(b) and iii.(b) are satisfied. Of course $\eta_{\alpha} := \eta_{\alpha}^{0}$ for $\alpha \in \Delta_{0}$. For a preliminary consideration, choose arbitrary $\tilde{\eta}_{\alpha} \in K^{*}$ for $\alpha \in \Delta \setminus \Delta_{0}$. The previous proposition shows the existence of an isomorphism $\tilde{F}: (\mathfrak{g}_{K}, \mathfrak{t}_{K}) \xrightarrow{\sim} (\mathfrak{g}'_{K}, \mathfrak{t}'_{K})$ such that $\tilde{F} \leftrightarrow (\psi, (\tilde{\eta}_{\alpha})_{\alpha \in R})$ with certain $\tilde{\eta}_{\alpha}$ for $\alpha \in R \setminus \Delta$. Using i.(a) and 3.3.3 we see that the restriction of such \tilde{F} to $(\mathfrak{g}_{a})_{K}$ coincides with $(f_{a})_{K}$, so that necessarily $\tilde{\eta}_{\alpha} = \eta_{\alpha}^{0}$ for $\alpha \in R_{0}$. Further, using 3.3.4 and ${}^{\sigma}\psi = \psi$, we get

$${}^{\sigma}\widetilde{F}\circ\widetilde{F}^{-1}\longleftrightarrow (\operatorname{id}, (\widetilde{\zeta}_{\sigma,\sigma_{A}^{-1}(\alpha)})_{\alpha\in R})$$

with

(15)
$$\widetilde{\zeta}_{\sigma,\alpha} := \frac{\xi'_{\sigma,\psi(\alpha)}}{\xi_{\sigma,\alpha}} \cdot \frac{\sigma(\widetilde{\eta}_{\alpha})}{\widetilde{\eta}_{\sigma_A(\alpha)}}$$

so that iii.(b) is equivalent to $\widetilde{\zeta}_{\sigma,\alpha} = 1$ for all $(\sigma,\alpha) \in \Gamma \times R$. (NB: This is already satisfied for $(\sigma,\alpha) \in \Gamma \times R_0$ because the η^0_{α} came from $(f_a)_K \leftrightarrow (\psi_0, (\eta^0_{\alpha})_{\alpha \in R_0})$, and $\mathrm{id}_{(\mathfrak{g}_a)_K} = {}^{\sigma}(f_a)_K \circ (f_a)_K^{-1} \leftrightarrow (\mathrm{id}_{X_a}, (\widetilde{\zeta}_{\sigma,\sigma_A^{-1}(\alpha)})_{\alpha \in R_0}).)$

Now 3.3.5 shows that for $\sigma \in \Gamma$ there is a $\Phi_{\sigma} \in T_X$ with ${}^{\sigma}\widetilde{F} \circ \widetilde{F}^{-1} = i(\Phi_{\sigma})$, hence

$$\widetilde{\zeta}_{\sigma,\sigma_A^{-1}(\alpha)} = \Phi_{\sigma}(\alpha)$$

for all $\alpha \in R$. Because Φ_{σ} (a fortiori $\Phi_{\sigma} \circ \sigma_A$) is a group homomorphism, it follows that

 $\triangleright \ \widetilde{\zeta}_{\sigma,\alpha} = \widetilde{\zeta}_{\sigma,\beta} \text{ if } \alpha - \beta \in X_a \text{ (use } \sigma_A^{-1}(R_0) = R_0 \text{ and the NB above)};$

▷ it suffices to choose $(\tilde{\eta}_{\alpha})_{\alpha \in \Delta \setminus \Delta_0}$ such that $\tilde{\zeta}_{\sigma,\alpha} = 1$ for $\alpha \in \Delta \setminus \Delta_0$ and $\sigma \in \Gamma$. From (12) and (15) one also calculates (for $\sigma, \tau \in \Gamma, \alpha \in R$):

(16)
$$\widetilde{\zeta}_{\sigma\tau,\alpha} = \widetilde{\zeta}_{\sigma,\tau_A(\alpha)} \cdot \sigma(\widetilde{\zeta}_{\tau,\alpha})$$

60

Now fix a first choice of $\tilde{\eta}_{\alpha} \in K^*$ for $\alpha \in \Delta \setminus \Delta_0$. Fix any such α . Set

$$\overline{\Gamma}_{\alpha} := \big\{ \sigma \in \Gamma : t(\sigma)(\alpha) = \alpha \big\},$$

a normal subgroup of Γ . Then 3.1.43.ii (or iv), the first bullet above and (16) show that $\sigma \mapsto \widetilde{\zeta}_{\sigma,\alpha}$ is a cocycle $\overline{\Gamma}_{\alpha} \to K^*$. Thus by Hilbert 90, there is $\mu_{\alpha} \in K^*$ such that $\widetilde{\zeta}_{\sigma,\alpha} = \sigma(\mu_{\alpha})\mu_{\alpha}^{-1}$ for $\sigma \in \overline{\Gamma}_{\alpha}$.

Set $\eta_{\alpha} := \mu_{\alpha}^{-1} \widetilde{\eta}_{\alpha}$, keeping the other $\widetilde{\eta}_{\bullet}$ on $\Delta \setminus \{\alpha\}$. In this second choice, we have newly determined η_{β} for all roots β of the form $\sum_{\delta \in \Delta} n_{\delta} \delta$ with $n_{\alpha} \neq 0$; in particular $\eta_{\sigma_A(\alpha)} = \mu_{\alpha}^{-1} \widetilde{\eta}_{\sigma_A(\alpha)}$ for all $\sigma \in \overline{\Gamma}_{\alpha}$ by 3.1.43.iv and the relations in 3.3.3. So for this improved choice we have

$$\zeta_{\sigma,\alpha} = \frac{\xi'_{\sigma,\psi(\alpha)}}{\xi_{\sigma,\alpha}} \cdot \frac{\sigma(\mu_{\alpha}^{-1}\widetilde{\eta}_{\alpha})}{\mu_{\alpha}^{-1}\widetilde{\eta}_{\sigma_A(\alpha)}} = \widetilde{\zeta}_{\sigma,\alpha} \cdot \frac{\mu_{\alpha}}{\sigma(\mu_{\alpha})} = 1$$

for all $\sigma \in \overline{\Gamma}_{\alpha}$. The next step will be to extend this to all $\sigma \in \Gamma$.

Let $\sigma_1, \ldots, \sigma_m$ be a complete set of representatives of $\Gamma/\overline{\Gamma}_{\alpha}$ with $\sigma_1 = \text{id. Consider}$ the orbits, with respect to the *t*-action, of Γ in $\Delta \setminus \Delta_0$: the one which contains α is $\{\alpha, t(\sigma_2)\alpha, \ldots, t(\sigma_m)\alpha\}$. Pick $j \in \{2, \ldots, m\}$ and define

$$\beta := (\sigma_j^{-1} w_{\sigma_j})(\alpha) = (\sigma_{jA}^{-1} \circ w_{\sigma_j} \circ \sigma_{jA})(\alpha).$$

Since $w_{\sigma_j} \in W_0$, also $\sigma_j^{-1} w_{\sigma_j} \in W_0$ (3.1.33), and thus by 3.1.34,

$$\beta = \alpha + \sum_{\delta \in \Delta_0} n_{\delta} \delta$$
 with certain $n_{\delta} \in \mathbb{Z}_{\geq 0}$.

On the other hand, $t(\sigma_j)\alpha$ is an element of $\Delta \setminus (\Delta_0 \cup \{\alpha\})$. So when we now make a third choice, which differs from the second one in defining:

$$\eta_{t(\sigma_j)\alpha} := \sigma_j(\eta_\beta) \cdot \frac{\xi'_{\sigma_j,\psi(\beta)}}{\xi_{\sigma_j,\beta}}$$

where the things on the right hand side are from the second choice, then η_{β} remains the same in the third as in the second choice. This is made so that in this third choice, by (15) and using $\sigma_{jA}(\beta) = t(\sigma_j)\alpha$, we get

$$\zeta_{\sigma_j,\beta} = \frac{\sigma_j(\eta_\beta)}{\eta_{\sigma(\beta)}} \cdot \frac{\xi'_{\sigma_j,\psi(\beta)}}{\xi_{\sigma_j,\beta}} = 1$$

and hence also $\zeta_{\sigma_j,\alpha} = 1$ because $\beta - \alpha \in X_a$. Writing a general σ as product of some σ_j with something in $\overline{\Gamma}_{\alpha}$ and using the cocycle condition (16), we now have achieved $\zeta_{\sigma,\alpha} = 1$ for all $\sigma \in \Gamma$.

But plugging this result into (16) once more, we see that we even have $\zeta_{\sigma,\tau(\alpha)} = 1$ for every $\sigma, \tau \in \Gamma$, and so by $\tau(\alpha) - t(\tau)(\alpha) \in X_a$ and the first bullet, we have reached

$$\zeta_{\sigma,\alpha'} = 1$$

where α' is any root in the *t*-orbit of α , and σ is arbitrary.

Now if the orbit is the whole set $\Delta \setminus \Delta_0$, we are finished. If not, pick a new α from another orbit and repeat the procedure – there is no interference with the earlier adjustments. After finitely many steps, the proof is complete, and the F given by the ψ and η_{α} is constructed and satisfies ${}^{\sigma}F = F$ for any $\sigma \in \Gamma$.

The proof of the theorem is finished as follows. The above F by 3.3.6 descends to an isomorphism $f : (\mathfrak{g}, \mathfrak{t}) \xrightarrow{\sim} (\mathfrak{g}', \mathfrak{t}')$ which is of the form described in the theorem: by part i of the proposition, it induces f_a , and by part ii, $\mathrm{id} \otimes f \cong F$ induces the congruence h.

3.4. Visualisation: Satake-Tits diagrams

A convenient way to visualise an index $(X, R, \Delta, \Delta_0, t)$ was introduced by Satake in [Sat1]. Closely resembling is the visualisation in [Tit1], so the name *Satake-Tits* diagram seems appropriate. What one does is to take the Dynkin diagram of the root system R (in V, with respect to the basis Δ) and to decorate it by

 \triangleright colouring black those vertices whose corresponding roots are in Δ_0 (Tits instead encircles those in $\Delta \setminus \Delta_0$; confusingly, in the table of [**Spr3**, p. 320], the vertices of $\Delta \setminus \Delta_0$ are blackened);

 \triangleright drawing arrows which show how the *t*-action permutes the vertices (Tits instead encircles the ones in a common *t*-orbit). Indeed, by 3.1.42 and [**Bo2**, VI.4.2, Cor.], each $t(\sigma)$ "is" a diagram automorphism.

So the Satake-Tits diagrams of our examples 3.2.5 and 3.2.9 are:



Of course, equivalent indices give the same diagram. A Lie algebra is anisotropic (resp. quasi-split, resp. split) if and only if all vertices in its diagram are black (resp.

white, resp. white and there are no arrows). Since in all cases except for the type D_4 , the automorphism group of a connected Dynkin diagram has order ≤ 2 , the arrows do not get very complicated in many cases. Indeed if [K : k] = 2 for K the fixed field of t, we can and will do without the labelling of the arrows (they would all be the non-trivial automorphism of K|k), and instead add the field K to the description. This information describes the index up to equivalence.

REMARK 3.4.1. — We call a Satake-Tits diagram *closed* if every pair of vertices in it is linked by a combination of edges and/or arrows. We will see in Section 4.1 that a Lie algebra with closed Satake-Tits diagram arises by scalar restriction from an "absolutely simple one" whose underlying Dynkin diagram is connected; and in 4.3.5 that a partition of a Satake-Tits diagram in its closed components corresponds to a decomposition of the Lie algebra in its simple factors. So a Lie algebra is simple (resp. absolutely simple) if its Satake-Tits diagram is closed (resp. has connected underlying Dynkin diagram).

We end this section and chapter with a useful application of Satake-Tits diagrams.

3.4.1. Application: how to compute the rational root system. — We describe a sometimes convenient way to read off the k-rational roots \overline{R} from the Satake-Tits diagram of a Lie algebra \mathfrak{g} . The method is given by Tits in [**TiW**, 42.3.5], cf. [**Tit1**, 2.5]. Assume that the k-rank of \mathfrak{g} is ≥ 1 (in other words, \overline{R} is not empty), and that the underlying Dynkin diagram is connected, so that R is irreducible.

LEMMA 3.4.2. — The root system \overline{R} is irreducible. Fixing a Γ -basis Δ , ρ projects the highest root to the highest root of \overline{R} with respect to $\overline{\Delta}$.

Proof (see [**Spr3**, 15.5.6]; for the highest root $\alpha \in R$, see [**Bo2**, VI.1.8]). — One sees that for any positive root $\beta \in R$, there is a finite sequence $\beta = \beta_0, \beta_1, \ldots, \beta_n = \alpha$ such that $\beta_i - \beta_{i-1} = n_i \delta_i$ for certain $n_i \ge 0$ and $\delta_i \in \Delta$. Reduction of this sequence gives an analogous one in \overline{R} which shows that in fact $\rho(\alpha)$ is the unique highest root in \overline{R} , so that \overline{R} is irreducible.

Using that \overline{R} is an irreducible root system, it can now be determined by the surjectivity of $\rho : R \rightarrow \overline{R}$, 3.1.43, 3.1.45, and the plates at the end of [**Bo2**, VI]. Let us show the method with two typical examples: We will later encounter the diagrams



of respective types E_6 and D_4 . By 3.1.43.v and 3.1.45, the k-rank of \mathfrak{g} (= rank of \overline{R}) is the number of t-orbits containing white vertices (Tits calls these the "distinguished orbits"). So in both examples, the rational root system has rank 2. For the E_6 type, $\rho(\alpha_2)$ and $\rho(\alpha_4)$ (α_2 is the "lower vertex" in Bourbaki's odd-looking labelling, α_4 is the one above it) are a basis of \overline{R} ; Bourbaki's plate shows that the root system R of type E_6 has highest root

$$\alpha := \alpha_1 + 2\alpha_2 + 2\alpha_3 + 3\alpha_4 + 2\alpha_5 + \alpha_6$$

so \overline{R} has highest root $\rho(\alpha) = 2\rho(\alpha_2) + 3\rho(\alpha_4)$. The only root system of rank 2 which contains such a combination of basis roots is G_2 , so this must be it, with $\rho(\alpha_4)$ being the shorter root.

For the D_4 type, analogously, one basis root of \overline{R} is $\rho(\alpha_2)$ (α_2 is the "middle" vertex), and the other is $\rho(\alpha_1) = \rho(\alpha_3) = \rho(\alpha_4)$: being in one *t*-orbit, they all have the same image. Now R has highest root

$$\alpha_1 + 2\alpha_2 + \alpha_3 + \alpha_4$$

so \overline{R} contains $3\rho(\alpha_1)+2\rho(\alpha_2)$, and again \overline{R} must be of type G_2 , now with the common projection of the "outer" roots as the short one.

One more example: We had seen the quasi-split forms of A_n . Say $n \ge 3$ is odd, d := n + 1. The Satake-Tits diagram is



 \overline{R} has basis

$$\delta_1 := \rho(\alpha_1) = \rho(\alpha_n), \\ \delta_2 := \rho(\alpha_2) = \rho(\alpha_{n-1}), \dots, \\ \delta_{d/2-1} := \rho(\alpha_{d/2-1}) = \rho(\alpha_{d/2+1}), \\ \delta_{d/2} := \rho(\alpha_{d/2})$$

and we see that it has highest root

$$\sum_{i=1}^{d/2-1} 2\delta_i + \delta_{d/2}$$

and also contains

$$(0 \cdot \delta_1 +) \sum_{i=2}^{d/2-1} 2\delta_i + \delta_{d/2}.$$

The first information still gives us the possibilities $B_{d/2}$ and $C_{d/2}$, with $\delta_{d/2}$ being the leftmost or rightmost vertex in the usual Dynkin diagram labelling. But then the second information rules out $B_{d/2}$ (except for d = 4, where $B_2 = C_2$), so \overline{R} is of type $C_{d/2}$ as claimed in 3.2.10.

This method can be used to *exclude* many diagrams, because they would give rise to impossible rational root systems. We do not need it in this generality, but will exploit it in the case of k-rank 1, see 4.3.3.
CHAPTER 4

CLASSIFICATION OVER GENERAL FIELDS

Interpreting the isomorphism theorem as a *uniqueness* statement, we can no longer avoid the question of *existence*. The question is: Which diagrams and which anisotropic kernels can occur over a given field k? For example, if k is algebraically closed, one knows that there cannot be any arrows or black vertices (i.e. no anisotropic kernels). In Sections 4.1–4.4 we develop methods to exclude or construct various diagrams, by a combination of *algebraic* and *combinatorial* reasoning. Section 4.5 gives a general theory for the classical types A-D, and Section 4.6 deals with the quasi-split forms. Both suggest, as earlier examples did, that for a complete classification one has to take account of the *arithmetic* of the given field, which will be done in the next chapter.

In this chapter, char(k) = 0 if not noted otherwise. All Lie algebras are understood to be of finite dimension. We say that an index or a Satake-Tits diagram is *k*-admissible if there is a semisimple Lie algebra over k which has this index or diagram.

4.1. Reduction to absolutely simple Lie algebras

It is obvious that to classify semisimple Lie algebras, we only have to classify simple ones. But we can reduce further.

In this section, our ground field k is only assumed to be perfect, with algebraic closure \bar{k} .⁽¹⁾

DEFINITION 4.1.1. — A Lie algebra \mathfrak{g} over k is called *absolutely simple* if $\mathfrak{g}_{\bar{k}}$ is simple over \bar{k} , or equivalently, \mathfrak{g}_{K} is simple over K for every extension K|k.

It turns out that every simple Lie algebra is the scalar restriction of an absolutely simple one. This was announced by Albert and proven – for a larger class of non-associative algebras than Lie algebras – with different methods by Landherr

^{1.} The perfectness assumption is used only in the proposition.

(cf. [Lan1]) and Jacobson (cf. [Jac2]) in the 1930s. We present here a nice little theory around this fact, generally following Jacobson, and along the way solving [Bo2, VIII.4, Exercise 4].

For a Lie algebra \mathfrak{g} over k, let $A(\mathfrak{g})$ be the (associative, unital) k-subalgebra of $\operatorname{End}_k(\mathfrak{g})$ generated by all $\operatorname{ad}_{\mathfrak{g}}(x), x \in \mathfrak{g}$. Remark straightaway that for any field extension $L|k, a \otimes \operatorname{ad}_{\mathfrak{g}}(x) \mapsto \operatorname{ad}_{\mathfrak{g}_L}(a \otimes x)$ defines a natural isomorphism of associative L-algebras:

(17)
$$L \otimes_k A(\mathfrak{g}) \cong A(\mathfrak{g}_L)$$

Also remark that \mathfrak{g} is a (left) $A(\mathfrak{g})$ -module, and that an ideal of \mathfrak{g} is the same as an $A(\mathfrak{g})$ -submodule. Further define:

$$K:=K(\mathfrak{g}):=\left\{s\in \mathrm{End}_k(\mathfrak{g}):s\circ \mathrm{ad}_\mathfrak{g}(x)=\mathrm{ad}_\mathfrak{g}(x)\circ s \text{ for all } x\in \mathfrak{g}\right\}$$

which as finite dimensional k-vector space is the same as $\mathfrak{z}_{\mathfrak{gl}}(\mathfrak{g})(\mathrm{ad}_{\mathfrak{g}}(\mathfrak{g}))$; but we view it as associative k-algebra and remark that as such it identifies with $\mathrm{End}_{A(\mathfrak{g})}(\mathfrak{g})$.

Now let \mathfrak{g} be simple. Then K is a skew field by Schur's lemma. In fact, it is a field; namely, since $\mathfrak{g} = [\mathfrak{g}, \mathfrak{g}]$ it suffices to see that two elements $s, t \in K$ commute on a commutator [x, y] for $x, y \in \mathfrak{g}$. But

$$s(t([x,y])) = s([x,ty]) = [sx,ty] = t([sx,y]) = t(s([x,y]))$$

where we have used, from left to right, that t commutes with $\operatorname{ad}_{\mathfrak{g}}(x)$, s with $-\operatorname{ad}_{\mathfrak{g}}(ty)$, t with $\operatorname{ad}_{\mathfrak{g}}(sx)$ and s with $-\operatorname{ad}_{\mathfrak{g}}(y)$. We call K the *centroid* of \mathfrak{g} and remark that \mathfrak{g} has a natural structure as Lie algebra over K. When viewing it as such, we denote it by $K\mathfrak{g}$.

We claim that the centroid of ${}^{K}\mathfrak{g}$ is K. Indeed, it is

$$\left\{ s \in \operatorname{End}_{K}(^{K}\mathfrak{g}) : s \circ \operatorname{ad}_{\kappa}\mathfrak{g}(x) = \operatorname{ad}_{\kappa}\mathfrak{g}(x) \circ s \text{ for all } x \in ^{K}\mathfrak{g} \right\}$$
$$\subseteq \left\{ s \in \operatorname{End}_{k}(\mathfrak{g}) : s \circ \operatorname{ad}_{\mathfrak{g}}(x) = \operatorname{ad}_{\mathfrak{g}}(x) \circ s \text{ for all } x \in \mathfrak{g} \right\} = K$$

and the other inclusion is clear. Further, the inclusion $A(\mathfrak{g}) \subseteq \operatorname{End}_k(\mathfrak{g})$ factors through natural maps $A(\mathfrak{g}) \hookrightarrow \operatorname{End}_K({}^K\mathfrak{g}) \hookrightarrow \operatorname{End}_k(\mathfrak{g})$, and the first arrow is bijective by Jacobson's density theorem [**Bo1**, VIII.5.5].⁽²⁾ Consequently, the following are equivalent:

i.
$$\mathfrak{g}$$
 is simple and $K = k$

ii. $A(\mathfrak{g}) = \operatorname{End}_k(\mathfrak{g}).$

In this case we call \mathfrak{g} central simple. So e.g. ${}^{K}\mathfrak{g}$ is central simple if \mathfrak{g} is simple. It follows from (17) that every scalar extension of a central simple Lie algebra is again central simple, a fortiori absolutely simple. But we have much more:

PROPOSITION 4.1.2. — Let \mathfrak{g} be a simple Lie algebra and L|k a Galois extension containing the centroid K. Then $\mathfrak{g}_L \simeq \mathfrak{g}_1 \times \cdots \times \mathfrak{g}_r$ where r = [K:k] and the \mathfrak{g}_i are

66

^{2.} Or $\S4.2$ in earlier editions. The theorem is absent from [**Jac2**], as he only proved it eight years later!

absolutely simple Lie algebras over L. In particular, \mathfrak{g} is central simple if and only if it is absolutely simple.

Proof. — Writing K = k[X]/(f) where f is a minimal polynomial of a primitive element of K|k, we have $L \otimes_k K \cong \prod_{i=1}^r L_i$ (as *L*-algebras) where the L_i are all L but with an *L*-action twisted via certain elements $\sigma_i : L \simeq L_i$ of the Galois group $\operatorname{Gal}(L|k)$, permuting the zeros of $f \in L[X]$. In particular, r = [K:k]. Then with (17),

$$A(\mathfrak{g}_L) \cong L \otimes_k \operatorname{End}_K({}^K \mathfrak{g}) \cong \operatorname{End}_{L \otimes_k K} \left((L \otimes_k K) \otimes_K ({}^K \mathfrak{g}) \right)$$
$$\cong \operatorname{End}_{\prod_{i=1}^r L_i} \left(\bigoplus_{i=1}^r ({}^K \mathfrak{g})_{L_i} \right) \cong \prod_{i=1}^r \operatorname{End}_{L_i} \left(({}^K \mathfrak{g})_{L_i} \right).$$

Calling e_i the *i*-th idempotent in the last product, the $A(\mathfrak{g}_L)$ -module $e_i \cdot \mathfrak{g}_L$ is a simple ideal \mathfrak{g}_i in \mathfrak{g}_L , which is in fact the simple *L*-Lie algebra deduced from $\binom{K}{\mathfrak{g}}_L$ by scalar extension (i.e. twisting the *L*-action) with σ_i . (It can be shown that as soon as *L* is a splitting field for \mathfrak{g} , the \mathfrak{g}_i are all isomorphic as Lie algebras over *L*.)

As promised, we can write our simple Lie algebra as scalar restriction of an absolutely simple one: $\mathfrak{g} = R_{K|k}({}^{K}\mathfrak{g})$. Now let $\operatorname{char}(k) = 0$. The proposition and its proof show how the Satake-Tits diagram of \mathfrak{g} looks like: It consists of [K:k] copies of the Satake-Tits diagram of ${}^{K}\mathfrak{g}$ (whose underlying Dynkin diagram is irreducible), and these copies are connected by arrows. To be more precise, for a finite Galois splitting field L, one should index the copies by L_i corresponding to $L \otimes_k K \cong \prod_{i=1}^r L_i$; the arrows between the copies correspond to the way $\operatorname{Gal}(L|k)$ interchanges the L_i and hence the \mathfrak{g}_i in the proposition. In particular, a simple Lie algebra has closed Satake-Tits diagram in the sense of 3.4.1.

As an example, start with a split simple (automatically absolutely simple) Lie algebra \mathfrak{G} over a finite extension K|k and consider $\mathfrak{g} := R_{K|k}\mathfrak{G}$. The Satake-Tits diagram will consist of [K:k] copies of the Dynkin diagram of \mathfrak{G} with arrows between them; so if [K:k] > 1, \mathfrak{g} is non-split quasi-split (but also: not absolutely simple). More generally, every finite scalar restriction of a quasi-split Lie algebra is quasisplit. So the Satake-Tits diagram of $R_{K|k} \mathfrak{sl}_2(K)$ for a Galois resp. non-Galois cubic extension K|k are:



Here, σ is a generator of $\operatorname{Gal}(K|k) \simeq \mathbb{Z}/3$, resp. σ_i are the three transpositions in $S_3 \simeq \operatorname{Gal}(L|k)$, L being the normal closure of K. We will use this in Section 4.6.

4.2. Galois cohomology and forms of certain types

General references for this section are [Ser2, VII, Appendix and X], [Ser3, III.1] and [Wei]. The variants in [Jac6, X] and [Sel1, IV..1–2] have their origin in Jacobson's pre-cohomological version (cf. [Jac3, 6]). Compare also [Sat3, I.3].

Let K|k be a field extension and \mathfrak{G} a Lie algebra over K. A K|k-form of \mathfrak{G} is a pair (\mathfrak{g}, F) consisting of a Lie algebra \mathfrak{g} over k and an isomorphism $F : \mathfrak{G} \xrightarrow{\sim} \mathfrak{g}_K$ of Lie algebras over K. Sometimes we drop the specific F and just say that \mathfrak{g} is a K|k-form of \mathfrak{G} .

Now let \mathfrak{g}_1 and \mathfrak{g} be Lie algebras over k such that (\mathfrak{g}_1, F) is a K|k-form of \mathfrak{g}_K , and suppose K|k is Galois. Every $\sigma \in \operatorname{Gal}(K|k)$ acts on both $(\mathfrak{g}_1)_K$ and \mathfrak{g}_K , and we set ${}^{\sigma}F := \sigma \circ F \circ \sigma^{-1}$ which is again a K-isomorphism from \mathfrak{g}_K to $(\mathfrak{g}_1)_K$. This is compatible with the notation before and in 3.3.4, and in the special case $\mathfrak{g} = \mathfrak{g}_1$ gives a left $\operatorname{Gal}(K|k)$ -action on $\operatorname{Aut}(\mathfrak{g}_K)$. In the general case, set

$$a_{\sigma} := F^{-1} \circ {}^{\sigma}F.$$

Then it is easily checked that $a_{\sigma} \in \operatorname{Aut}(\mathfrak{g}_K)$ and that $a_{\sigma\tau} = a_{\sigma}({}^{\sigma}a_{\tau})$ for all σ, τ in $\operatorname{Gal}(K|k)$, so that $\sigma \mapsto a_{\sigma}$ is a cocycle. Next, if (\mathfrak{g}_1, F) and (\mathfrak{g}_2, F') are two K|kforms of \mathfrak{g}_K as above, and if there is a k-isomorphism of Lie algebras $f : \mathfrak{g}_2 \xrightarrow{\sim} \mathfrak{g}_1$, for the cocycles a_{σ} and b_{σ} corresponding to F and F' as above, we have

$$b_{\sigma} = \phi^{-1} \circ a_{\sigma} \circ ({}^{\sigma}\phi)$$

for all $\sigma \in \text{Gal}(K|k)$, where $\phi := F^{-1} \circ f_K \circ F' \in \text{Aut}(\mathfrak{g}_K)$. In other words, the corresponding cocycles are cohomologous.

THEOREM 4.2.1. — The above assignment induces a bijection from the set of isomorphism classes of K|k-forms of \mathfrak{g}_K to $H^1(\operatorname{Gal}(K|k), \operatorname{Aut}(\mathfrak{g}_K))$.

Although this is proven in much greater generality e.g. in [Ser2, X.2], let us stress the following down-to-earth approach to the inverse of this map. Given a cocycle

$$a: \operatorname{Gal}(K|k) \longrightarrow \operatorname{Aut}(\mathfrak{g}_K), \quad \sigma \longmapsto a_{\sigma},$$

we get for each $\sigma \in \operatorname{Gal}(K|k)$ a map $u_a(\sigma) := a_\sigma \circ \sigma$ (where the second σ denotes the canonical action on \mathfrak{g}_K). Then $u_a(\sigma)$ is a σ -semilinear, bijective map from \mathfrak{g}_K to itself, compatible with the Lie bracket, we have $u_a(\operatorname{id}) = \operatorname{id}$ and $u_a(\sigma\tau) = u_a(\sigma) \circ u_a(\tau)$. One calls the u_a 's *Galois semi-automorphisms* and says they define the *Galois action* twisted with the cocycle a. Define $\mathfrak{g}(a)$ to be the set of elements of \mathfrak{g}_K which are fixed by all $u_a(\sigma), \sigma \in \operatorname{Gal}(K|k)$. Then $\mathfrak{g}(a)$ is a Lie algebra over k, and one shows that the K-span of $\mathfrak{g}(a)$, which identifies with $\mathfrak{g}(a)_K$, is all of \mathfrak{g}_K . Hence $\mathfrak{g}(a)$ is a K|kform of \mathfrak{g}_K . If b is a cocycle cohomologous to a, say $b_{\sigma} = \phi^{-1} \circ a_{\sigma} \circ (\sigma \phi)$ for some $\phi \in \operatorname{Aut}(\mathfrak{g}_K)$, then ϕ restricts to an isomorphism $\mathfrak{g}(b) \xrightarrow{\sim} \mathfrak{g}(a)$ of Lie algebras over k.

Now let char(k) = 0 and let all extensions be inside a fixed algebraic closure $\bar{k}|k$. As noted in Chapter 3, for every semisimple \mathfrak{g} over k there is a finite Galois extension K|k

MÉMOIRES DE LA SMF 151

68

such that \mathfrak{g}_K is split over K. The split semisimple Lie algebras, in turn, are classified and can be defined over any prime field (for us, \mathbb{Q}). We can express this as follows: For each (isomorphism class of) irreducible root system $R \in \{A_n, B_n, \ldots, G_2\}$, fix a split Lie algebra $\mathfrak{S}(R)$ of corresponding type over \mathbb{Q} . (This corresponds to fixing "base points" in cohomology sets.) Then for every *absolutely simple* Lie algebra \mathfrak{g} over kthere is a finite Galois extension K|k such that \mathfrak{g} is a K|k-form of one $\mathfrak{S}(R)_K$.⁽³⁾ In this case, we say that \mathfrak{g} is a *form of type* R, and it is easily seen that this type is indeed well-defined and depends only on (much less than) the k-isomorphism class of \mathfrak{g} ; in fact, it "is" the \overline{k} -isomorphism class of $\mathfrak{g}_{\overline{k}}$, as \mathfrak{g} is a $\overline{k}|k$ -form of $\mathfrak{S}(R)_{\overline{k}}$.

Putting this information together and using the limit procedures for infinite Galois cohomology, we get:

THEOREM 4.2.2. — The isomorphism classes of forms of type R over k are in bijection with the pointed set

$$H^1(\operatorname{Gal}(\bar{k}|k),\operatorname{Aut}(\mathfrak{S}(R)_{\bar{k}})),$$

the class of the split form being the distinguished point. Those forms which have the Galois splitting extension K|k correspond to the ones in

$$H^1(\operatorname{Gal}(K|k),\operatorname{Aut}(\mathfrak{S}(R)_K))$$

which by inflation identifies with a subset of the above.

This translates our problem to a computation of Galois cohomology sets. Of course this is an equally difficult problem, but the cohomology machinery gives new methods to attack it. The classification of the *split* Lie algebras, which provides the underlying Dynkin diagrams to the Satake-Tits diagrams in our general approach, here pops up via their respective automorphism groups. The other ingredient are Galois groups, in which the field arithmetic is encoded.

Remark 4.2.3

- i. For $k = \mathbb{R}$, the Galois group $\mathbb{Z}/2$ looks innocent, and still one gets a rich theory. But the messy-looking classification using certain "involutions" in real or complex Lie groups, going back to E. Cartan and to be found e.g. in **[OVi**, Chap. 5.1] or **[Oni**, §3], finds a nice interpretation in a cohomological theorem by Borel and Serre, **[Ser3**, III.4.5].
- ii. A posteriori we will see that splitting fields for absolutely simple Lie algebras often have small degrees; in fact, over a p-adic field k, all forms of type different from A_n are split by extensions of degree ≤ 6 , and the splitting field can often be chosen to be unramified. But to arrive at this conclusion, we first have to exploit the general theory.

^{3.} In fact it follows from the previous section that a simple \mathfrak{g} is a K|k-form of some finite product of copies of $\mathfrak{S}(R)_K$ for a unique irreducible R.

DEFINITION 4.2.4. — A K|k-form of the split Lie algebra $\mathfrak{S}(R)_K$ is called an *inner* form if a corresponding cocycle has image in $\operatorname{Aut}_0(\mathfrak{S}(R)_K)$, i.e. if its corresponding element is in the image of the map

$$H^1(\operatorname{Gal}(K|k), \operatorname{Aut}_0(\mathfrak{S}(R)_K)) \longrightarrow H^1(\operatorname{Gal}(K|k), \operatorname{Aut}(\mathfrak{S}(R)_K))$$

induced by inclusion. Otherwise, it is called an outer form.

With the methods of the next subsection and [**Bo2**, VIII.5] one can show e.g. that an absolutely simple Lie algebra is an inner form of a split Lie algebra if and only if there are arrows in its Satake-Tits diagram. Note that $\operatorname{Aut}_0(\mathfrak{S})$ is of index ≤ 2 in $\operatorname{Aut}(\mathfrak{S})$ for split simple \mathfrak{S} except for those of type D_4 , where the quotient is the symmetric group S_3 . In this case, a form whose corresponding cocycle has images of order 3 or 6 modulo $\operatorname{Aut}_0(\mathfrak{S}(D_4))$ is called *of trialitarian type*.

4.2.1. Application: An existence statement. — Connecting the cohomological viewpoint with parts of our preparation for the isomorphism theorem in Chapter 3, we will later be able to prove *existence* of certain forms. The idea of the following is in **[Sat3**, II.3.1]; although it is necessarily a little technical, we hope that our presentation can add to clarification.

In our preparation for the isomorphism theorem we had expressed certain isomorphisms of Lie algebra as tuples. We will now do this for the automorphisms which are in the image of a cocycle. Let \mathfrak{g}' be a Lie algebra over k and fix $\mathfrak{s}', \mathfrak{t}', K|k$ as usual, giving its index (X', R', \ldots) . Then \mathfrak{g}' is a K|k-form of $\mathfrak{S}(R)_K$. Because $\mathfrak{S}(R)$ is actually defined over \mathbb{Q} , we can choose a maximal split toral \mathfrak{t} (giving rise to a root system R and lattice X) and bases for the corresponding root spaces $(e_\alpha)_{\alpha \in R}$ defined over \mathbb{Q} . Because the split maximal toral (= split Cartan) subalgebras in $\mathfrak{S}(R)_K$ are conjugate, we can assume the isomorphism $F : \mathfrak{S}(R)_K \to \mathfrak{g}'_K$ satisfies $F(\mathfrak{t}_K) = \mathfrak{t}'_K$. Let $\sigma \mapsto a_\sigma$ be the corresponding cocycle. We recall the notations of Section 3.3.1 and have a very special case: Because \mathfrak{t} and the e_α are defined over \mathbb{Q} , the Galois action on X is trivial, i.e. all $\sigma_A = \mathrm{id}$, and all $\xi_{\sigma,\alpha} = 1$ in (11). Further, calling $\psi : X \to X'$ the contragredient of F, we also just define $e'_{\psi(\alpha)} := F(e_\alpha)$ for all $\alpha \in R$, so that by definition (see (13)) we have

$$F \longleftrightarrow \left(\psi, (1)_{\alpha \in R}\right)$$

Then it follows immediately from 3.3.4 that

$$F^{-1} \longleftrightarrow (\psi^{-1}, (1)_{\alpha \in R'}),$$

 ${}^{\sigma}F \longleftrightarrow ({}^{\sigma}\psi, (\xi'_{\sigma,\psi(\alpha)})_{\alpha \in R}),$
 $a_{\sigma} = F^{-1} \circ {}^{\sigma}F \longleftrightarrow (a^*_{\sigma}, (\xi'_{\sigma,\psi(\alpha)})_{\alpha \in R}),$

where $a^*_{\sigma} = \psi^{-1} \circ {}^{\sigma}\psi$ is the contragredient of a_{σ} and the ξ'_{\bullet} are defined by $\sigma(e'_{\alpha}) = \xi'_{\sigma,\alpha} \cdot e'_{\sigma_{A'}(\alpha)}$. Indeed ${}^{\sigma}\psi = \sigma_{A'} \circ \psi$ and $a^*_{\sigma} = \psi^{-1} \circ \sigma_{A'} \circ \psi$ by triviality of the σ_A . From now on we stay in X', i.e. identify X and X' via ψ , so that $a^*_{\sigma} = \sigma_{A'}$. (One could say

that a_{σ} is the linear map whose "matrix" in the good basis e_{α} is the "matrix" of the semilinear map σ on \mathfrak{g}'_{K} with respect to the image, via F, of this basis.)

Fact 1. — The ξ'_{\bullet} satisfy

(18)
$$\xi'_{\sigma\tau,\alpha} = \sigma(\xi'_{\tau,\alpha}) \cdot \xi'_{\sigma,\tau_{A'}(\alpha)}$$

(for all $\alpha \in R, \sigma, \tau \in \Gamma$) by (12).

Fact 2. $-a_{\sigma}^*$ can be recovered from the index and $a_{\sigma|X'_a}^*$. Namely, for $\chi' \in X'$ and $\sigma \in \Gamma$ we have $t'(\sigma)(\chi') = w'_{\sigma}(\sigma_{A'}(\chi'))$ so that $a_{\sigma}^* = \sigma_{A'} = (w'_{\sigma})^{-1}t'(\sigma)$. We know $t'(\sigma)$ from the index and w'_{σ} from the restriction to X'_a .

Fact 3. — Knowing a_{σ}^* , the tuple $(\xi'_{\sigma,\alpha})_{\alpha \in R}$ is determined by $(\xi'_{\sigma,\alpha})_{\alpha \in R_0}$. Namely, this follows from the isomorphism theorem.

So with the information of its index $(X', R', \Delta', \Delta'_0, t')$, and assuming its anisotropic kernel is given as a form of $\mathfrak{S}(R'_0)_K$ (where K|k is finite Galois and contains the fixed field of t), one can actually construct the cocycle a and hence the Lie algebra \mathfrak{g}' as form of the split Lie algebra $\mathfrak{S}(R')_K$. This suggests to try the process with something that "could be an index":

DEFINITION 4.2.5. — A possible index is a tuple $(X, R, \Delta, \Delta_0, t)$ where R is a root system spanning the free \mathbb{Z} -module X, Δ is a basis of R, Δ_0 a subset of Δ and t a homomorphism from $\operatorname{Gal}(\bar{k}|k)$ to A(R), such that Δ and Δ_0 are stable under each $t(\sigma)$. Call R_0 the subsystem generated by Δ_0 , X_a the submodule spanned by R_0 .

Let a possible index $(X, R, \Delta, \Delta_0, t)$ be given and assume the following:

i. We have an anisotropic k-Lie algebra \mathfrak{g}_a given as a form of the split Lie algebra $\mathfrak{S}(R_0)_K$, with index $(X_a, R_0, \Delta_0, \Delta_0, t_{|X_a})$, where K|k is finite Galois and contains the fixed field of t.

As above this gives a tuple $(a^*_{\sigma | X_a}, (\xi_{\sigma,\alpha})_{\alpha \in R_0})$, and elements $w_{\sigma} \in W(R_0) \subseteq W(R)$ defined by $t(\sigma)(\chi) = w_{\sigma}(\sigma_A(\chi))$ for $\chi \in X_a$. Define, on all of X

$$a_{\sigma}^* := (w_{\sigma})^{-1} \circ t(\sigma)$$

ii. The $(\xi_{\sigma,\alpha})_{\alpha\in R_0}$ can be extended to $(\xi_{\sigma,\alpha})_{\alpha\in R}$ such that:

a) each $(a^*_{\sigma}, (\xi_{\sigma,\alpha})_{\alpha \in R})$ comes from a (necessarily unique) automorphism of $\mathfrak{S}(R)_K$, call it a_{σ} ;

b) (18) is satisfied.

PROPOSITION 4.2.6. — Under these conditions, there is a K|k-form \mathfrak{g} of the split Lie algebra $\mathfrak{S}(R)_K$ which has index $(X, R, \Delta, \Delta_0, t)$ and anisotropic kernel \mathfrak{g}_a . In particular, the index is k-admissible.

Proof. — With 3.3.4 one calculates that condition (ii) defines a cocycle $\sigma \mapsto a_{\sigma}$ from Γ with values in Aut($\mathfrak{S}(R)_K$). With the construction after theorem 4.2.1, this cocycle defines a form (\mathfrak{g}, F) of type R, so its index is (X, R, ?, ?, ?). The Galois action on \mathfrak{g} is

described by a_{σ} , in particular $\sigma_A = a_{\sigma}^*$. Define X_0 with respect to this Γ -action; from anisotropy of \mathfrak{g}_a and the Γ -action restricting to the original one on $X_a = \mathbb{Z} \cdot \Delta_0$, it follows that $\Delta_0 \subseteq (\Delta \cap X_0)$. Now we claim that for all $\sigma \in \Gamma$, $a_{\sigma}^*(\delta) = w_{\sigma}^{-1} \circ t(\sigma)(\delta)$ is positive (with respect to Δ) for $\delta \in \Delta \setminus \Delta_0$. Namely, $\delta' := t(\sigma)(\delta) \in \Delta \setminus \Delta_0$ by assumption, and $w_{\sigma}^{-1}(\delta') = \delta' + \sum_{\delta_i \in \Delta_0} n_i \delta_i$ for certain $n_i \in \mathbb{Z}_{\geq 0}$ because w_{σ} sends roots to roots and induces the identity on X/X_a . The claim is proven. It implies that indeed $\Delta_0 = \Delta \cap X_0$, since for $\delta \notin \Delta_0$ it implies that $\sum_{\sigma \in \Gamma} \sigma_A(\delta)$ is $\neq 0$, so that $\delta \notin X_0$. It also implies that Δ is a Γ -basis. It is then immediate that the twisted action with respect to this coincides with the given t.

4.3. Necessary and sufficient conditions on the index

In this section we exploit some facts which, although rather simple for the most part, will reduce the number of admissible indices by large. Unless noted otherwise, \mathfrak{g} is a semisimple k-Lie algebra with index $(X, R, \Delta, \Delta_0, t)$ with respect to $\mathfrak{s}, \mathfrak{t}, K$ as always, $\operatorname{char}(k) = 0$ and all extensions of k are in a fixed algebraic closure \bar{k} with $\mathcal{G} = \operatorname{Gal}(\bar{k}|k)$.

4.3.1. The opposition involution. — For any root system R with basis D, the opposition involution of D is the element $I \in A(R)$ defined as follows: there is a unique $w_I \in W(R)$ such that $w_I(D) = -D$. Set $I(\alpha) := -w_I(\alpha)$ for $\alpha \in R$. This defines an automorphism of the Dynkin diagram of (R, D) which stabilises each irreducible component. If R is irreducible, it is the identity, except in the cases A_n $(n \ge 2)$, E_6 and D_n (n odd (!)) where it is the unique non-trivial involution of the Dynkin diagram.

LEMMA 4.3.1. — Let I be the opposition involution of Δ . Then I stabilises Δ_0 and commutes with all $t(\sigma)$, $\sigma \in \mathcal{G}$.

Proof. — We show that $w_I \in W_{\Gamma}$. There is $w_1 \in W_{\Gamma}$ such that $\pi(w_1) \in W(\overline{R})$ maps $\overline{\Delta}$ to $-\overline{\Delta}$. Then $w_1(\Delta \setminus \Delta_0) = -(\Delta \setminus \Delta_0)$. Since $w_1(\Delta_0)$ is another basis of R_0 , there further is $w_2 \in W_0$ with $w_2(w_1(\Delta_0)) = -\Delta_0$. Then $w_2w_1(\Delta) = -\Delta$ and hence $w_I = w_2w_1 \in W_{\Gamma}$. This implies that I stabilises X_0 and thus Δ_0 . Now let $\sigma \in \mathcal{G}$. Since $t(\sigma) \in A(R)$, we have $(t(\sigma))^{-1} w_I t(\sigma) \in W(R)$, but since $t(\sigma)$ stabilises Δ and is linear, this element also maps Δ to $-\Delta$ and hence is w_I .

So roughly said, the black (and white) vertices in a Satake-Tits diagram of the mentioned types have to be symmetric. E.g. the following is impossible:



4.3.2. Admissible subindices. — In this subsection, stability under $t(\mathcal{G})$ and containment of (parts of the) anisotropic kernel in subdiagrams of a Satake-Tits diagram will allow us to descend corresponding subalgebras from \mathfrak{g}_K to \mathfrak{g} , which gives us admissibility criteria.

First, let Δ' be a subset of Δ which contains Δ_0 and is stable under every $t(\sigma)$. Set $X' := X \cap \mathbb{Q} \cdot \Delta'$, $R' := R \cap X'$, and denote by t' the induced twisted Galois action on X'. R' is the subsystem of R generated by (i.e. consisting of roots which are non-negative or non-positive combinations of elements of) Δ' , and we have $X' = \mathbb{Z}R'$, cf. [**Bo2**, VI.1.7].

Set $\overline{R}' := \rho(R') \setminus \{0\}$ which is the same as the subsystem of \overline{R} generated by $\overline{\Delta}' = \rho(\Delta') \setminus \{0\}$. The conditions on Δ' ensure by 3.1.43.v that $R \cap \rho^{-1}(\overline{R}' \cup \{0\}) = R'$. Further, set

$$\mathfrak{s}_0:=\bigcap_{\lambda\in\overline{R}'}\ker(\lambda)=\bigcap_{\lambda\in\overline{\Delta'}}\ker(\lambda)\subseteq\mathfrak{s}.$$

By 3.1.21, $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}_0)$ is reductive, so $\mathfrak{g}' := \mathcal{D}(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}_0))$ is semisimple. We have

$$\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}_0) = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \oplus \bigoplus_{\lambda \in \overline{R}'} \mathfrak{g}_{\lambda} \quad \text{and} \quad \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}_0)_K = \mathfrak{z}_{\mathfrak{g}_K}((\mathfrak{s}_0)_K) = \mathfrak{t}_K \oplus \bigoplus_{\alpha \in R'} (\mathfrak{g}_K)_{\alpha}.$$

One also sees $\mathfrak{s} \cap \mathfrak{z}(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}_0)) = \mathfrak{s}_0$ and thus

$$\mathfrak{s} \cong \mathfrak{s}_0 \times (\mathfrak{s} \cap \mathfrak{g}') \subseteq \mathfrak{z}(\mathfrak{z}_\mathfrak{g}(\mathfrak{s}_0)) \times \mathfrak{g}' \cong \mathfrak{z}_\mathfrak{g}(\mathfrak{s}_0)$$

LEMMA 4.3.2 (Erasing lemma). — The Lie algebra \mathfrak{g}' is semisimple with maximal split toral subalgebra $\mathfrak{s}' := \mathfrak{s} \cap \mathfrak{g}' = \sum_{\lambda \in \overline{R}'} kH_{\lambda}$ and maximal toral subalgebra $\mathfrak{t}' := \mathfrak{t} \cap \mathfrak{g}'$. With respect to these, its index is $(X', R', \Delta', \Delta_0, t')$.

Proof. — (In fact, \mathfrak{g}' is the standard Levi subalgebra of the parabolic subalgebra corresponding to Δ' via [**Bo2**, VIII.3.4, Rem.]. 3.1.2.1 was the case $\Delta' = \Delta_0$.) We have $\mathfrak{s} \cap \mathfrak{g}' = \sum_{\lambda \in \overline{R}'} kH_{\lambda}$ by definition of the H_{λ} (3.1.9) and 3.1.11.iv. One also sees that \mathfrak{g}' contains the anisotropic kernel \mathfrak{g}_a as subalgebra,

$$\mathfrak{g}' \supseteq \mathfrak{g}_a \oplus \sum_{\lambda \in \overline{R}'} kH_\lambda \oplus \bigoplus_{\lambda \in \overline{R}'} \mathfrak{g}_\lambda \quad \text{and} \quad \mathfrak{g}'_K = \sum_{\alpha \in R'} KH_\alpha \oplus \bigoplus_{\alpha \in R'} (\mathfrak{g}_K)_\alpha$$

as vector spaces. It follows that $\mathfrak{t}'_K = \sum_{\alpha \in R'} KH_\alpha$ is maximal toral in \mathfrak{g}'_K , hence \mathfrak{t}' is maximal toral in \mathfrak{g}' and a fortiori, \mathfrak{s}' is maximal split toral in \mathfrak{g}' . The rest now follows easily, using $X_0 \subseteq X'$.

This puts severe restrictions on indices. In a k-admissible Satake-Tits diagram it means the following: Erase the $t(\mathcal{G})$ -orbit of any set of white vertices (i.e. the vertices and all edges touching them); then what is left must again be a k-admissible diagram.

There is a kind of a converse of 4.3.2 which allows us to build k-admissible diagrams from ones with smaller k-rank:

PROPOSITION 4.3.3 (Patching proposition). — Let $(X, R, \Delta, \Delta_0, t)$ be a possible index (4.2.5). Assume there are subsets Δ', Δ'' of Δ both of which are invariant under $t(\mathcal{G})$ and such that

$$\Delta = \Delta' \cup \Delta'', \quad \Delta' \cap \Delta'' = \Delta_0.$$

Like in the beginning of this subsection, we can form the "possible subindices" $(X', R', \Delta', \Delta_0, t')$ and $(X'', R'', \Delta'', \Delta_0, t'')$. Let \mathfrak{g}_a be an anisotropic Lie algebra over k, given as K|k-form of $\mathfrak{S}(R_0)_K$, with index $(X_a, R_0, \Delta_0, \Delta_0, t|_{X_a}) =$ $(X'_a, R'_0, \Delta_0, \Delta_0, t_{|X'_a}) = (X''_a, R''_0, \Delta_0, \Delta_0, t_{|X''_a})$, and K|k finite Galois and containing the fixed field of t. Then if both possible subindices are k-admissible with Lie algebras $\mathfrak{g}',\mathfrak{g}''$ split over K and having \mathfrak{g}_a as anisotropic kernels, then $(X, R, \Delta, \Delta_0, t)$ is k-admissible with Lie algebra \mathfrak{g} having anisotropic kernel \mathfrak{g}_a .

Proof. — We want to apply 4.2.6. Like in Section 4.2.1, for all $\sigma \in \Gamma := \text{Gal}(K|k)$, the Lie algebras $\mathfrak{g}_a, \mathfrak{g}'$ and \mathfrak{g}'' define $a^*_{\sigma|X'}$ and $a^*_{\sigma|X''}$ which both extend $a^*_{\sigma|X_a}$ and by $\Delta' \cap \Delta'' \subseteq \Delta_0$ patch well to a common a_{σ}^* on X. Further they define tuples of scalars $(\xi^0_{\sigma,\alpha})_{\alpha\in R_0}, \ (\xi'_{\sigma,\alpha})_{\alpha\in R'}$ and $(\xi''_{\sigma,\alpha})_{\alpha\in R''}$ which coincide for common α (necessarily in R_0) and each satisfy (18) for their respective α 's. Now for $\alpha \in \Delta$ we set $\xi_{\sigma,\alpha} := \xi'_{\sigma,\alpha}$ or $\xi_{\sigma,\alpha}''$ according to whether $\alpha \in \Delta'$ or Δ'' . By 3.3.7 for $\psi = a_{\sigma}^* : X \to X$ these $(\xi_{\sigma,\alpha})_{\alpha\in\Delta}$ can be extended to $(\xi_{\sigma,\alpha})_{\alpha\in R}$ such that there exists an $a_{\sigma}\in \operatorname{Aut}(\mathfrak{S}(R)_K)$ with $a_{\sigma} \leftrightarrow (a_{\sigma}^*, (\xi_{\sigma,\alpha})_{\alpha \in R})$. (Note that certainly $\xi_{\sigma,\alpha}^0 = \xi_{\sigma,\alpha}$ for $\alpha \in R_0$ because of the assumption on the anisotropic kernel, but we do not claim these new ξ 's to coincide with any of the above ξ' or ξ'' for $\alpha \in R \setminus \Delta$.) To apply 4.2.6, it remains to check that $\sigma \mapsto a_{\sigma}$ is a cocycle or equivalently, these $\xi_{\sigma,\alpha}$ satisfy (18) for all $\alpha \in R$. They certainly do for $\alpha \in \Delta$ and we will see that this implies the stronger claim. Namely, with 3.3.4 one computes that for $\sigma, \tau \in \Gamma$:

$$a_{\sigma\tau}^{-1}a_{\sigma}({}^{\sigma}a_{\tau}) \leftrightarrow (\mathrm{id}, (\xi_{\sigma\tau,\alpha}^{-1} \cdot \xi_{\sigma,\tau_A(\alpha)} \cdot \sigma(\xi_{\tau,\alpha}))_{\alpha \in R})$$

so that by 3.3.5, we have $\xi_{\sigma\tau,\alpha}^{-1} \cdot \xi_{\sigma,\tau_A(\alpha)} \cdot \sigma(\xi_{\tau,\alpha}) = \Phi(\alpha)$ for some $\Phi \in \operatorname{Hom}_{\mathbb{Z}}(X, K^*)$. But as said, $\Phi_{|\Delta} \equiv 1$ and thus $\Phi \equiv 1$ and everything is proven. \square

REMARK 4.3.4. — It is possible to generalise this to finitely many sub-bases Δ' , and also to relax a little the condition that they all contain Δ_0 : It is only necessary that in the Satake-Tits diagram, the subdiagrams contain with each black vertex all black vertices linked to it by a (possibly double or triple) edge, cf. [Sat3, II.3.1].

REMARK 4.3.5. — Let $\mathfrak{g} = \prod_{i=1}^{r} \mathfrak{g}_i$ be a semisimple Lie algebra decomposed into its simple factors. With Section 4.1 one sees that this corresponds to a decomposition of the Satake-Tits diagram into closed components (cf. 3.4.1). Conversely, assume we have a partition of the Satake-Tits diagram of a semisimple \mathfrak{g} into closed subdiagrams S_i . Then a combination of the erasing lemma, the patching proposition (in the remarked generalised version) and the isomorphism theorem shows that \mathfrak{g} is the

 $\mathbf{74}$

direct product of subalgebras \mathfrak{g}_{S_i} with indicated diagrams, which are then necessarily simple.

4.3.3. The case of k-rank 1. — Let \mathfrak{g} be an absolutely simple Lie algebra over k whose k-rank is 1. This means that in its Satake-Tits diagram, there is either exactly one white vertex, or exactly two with an arrow between them.

(A priori there could also be the case of a diagram of type D_4 where the outer vertices are white and in one $t(\mathcal{G})$ -orbit, while the inner vertex α_2 is black:



where the action is via $\mathbb{Z}/3\mathbb{Z}$ and S_3 , respectively. But this is impossible since R contains its highest root $\alpha := \alpha_1 + 2\alpha_2 + \alpha_3 + \alpha_4$ and thus the root system \overline{R} would contain $\rho(\alpha_1)$ and $\rho(\alpha) = 3 \cdot \rho(\alpha_1)$.)

Let α (resp. α, β) be the root(s) in $\Delta \setminus \Delta_0$. In this situation, we have:

LEMMA 4.3.6. — If $\sum_{\delta \in \Delta} n_{\delta} \delta$ is any positive root in R, then $n_{\alpha} \in \{1, 2\}$ (resp. $n_{\alpha} = n_{\beta} = 1$).

Proof. — \overline{R} is a root system of rank 1 with basis $\overline{\Delta} = \{\rho(\alpha)\}$ (resp. = $\{\rho(\alpha) = \rho(\beta)\}$). It is either A_1 (which enforces the case $n_{\alpha} = 1$) or BC_1 , which implies the statement.

This little fact rules out many indices in the exceptional types. Going through the plates at the end of [**Bo2**, VI], we see which single white vertex (or orbit) is possible:

i. For inner forms of type E_6 , invoking also 4.3.1, the only possibility is $\alpha = \alpha_2$:



ii. For outer forms of type E_6 , the possibilities are $\alpha = \alpha_2$ and $\{\alpha, \beta\} = \{\alpha_1, \alpha_6\}$:



iii. For E_7 , the possibilities for a single white vertex are $\alpha_1, \alpha_2, \alpha_6$ and α_7 :



iv. For E_8 , the possibilities for a single white vertex are α_1 and α_8 :



●──●

v. For F_4 , the possibilities for a single white vertex are α_1 and α_4 :

vi. For G_2 , the only possibility for a single white vertex is α_2 :

4.3.4. An A_n application

PROPOSITION 4.3.7 (see Tits [**Tit1**]). — An inner form of type A_n with k-rank r can exist only if d := (n+1)/(r+1) is an integer. Its Satake-Tits diagram has to be



Proof. — Induction on r which is the number of white vertices. For r = 0 the assertion is trivial, and for r = 1, 4.3.1 implies that n has to be odd and the white vertex must be the central one. Now assume $n_0 \ge r_0 \ge 2$ and the assertion shown for $r < r_0$ and arbitrary n. Given a k-admissible diagram of inner type A_{n_0} with exactly r_0 white vertices, say the first white vertex from the left is the d-th vertex of all. Erase it. By the erasing lemma 4.3.2, what remains on the right of this has to be a k-admissible diagram, of inner type A_{n_0-d} and with k-rank $r_0 - 1$. By the induction hypothesis, $d' := (n_0 - d + 1)/r$ is an integer, the abridged diagram to the right looks as above with d' for d, and in particular, the second white vertex from the left in the original diagram is the (d + d')-th vertex of all. Erase this one, and again by 4.3.2, what is on the left of it,



is a k-admissible diagram. So by the case of k-rank 1 we must have d = d'. Hence $d(r_0 + 1) = n_0 - d + 1 + d = n_0 + 1$, i.e. $d = (n_0 + 1)/(r_0 + 1)$ and the whole diagram is of the asserted form.

We will describe these forms explicitly in the next section and 4.5.21.

4.4. Skew fields and anisotropic types

In this section, we show a method to produce absolutely simple anisotropic Lie algebras which generalises example 3.2.5. It will turn out later that over a p-adic field k, these are the only absolutely simple anisotropic forms.

For this section and the next, recall the concept of *central simple algebra* (CSA for short) over a field k, and how it leads to the *Brauer group* Br(k) (cf. [Ser2, X..4–7], [Pie, 12.5], [Rei, Chap. 7]). For a CSA A over k, we have its invariants

 \triangleright degree $\text{Deg}(A) = \sqrt{\dim_k A}$ and

 \triangleright index $\operatorname{Ind}(A) = \operatorname{Deg}(D)$ where D is a division algebra equivalent to A in $\operatorname{Br}(k)$. Further there is the

 \triangleright exponent (or period) Exp(A) which is the order of the class of A in Br(k). It is known (cf. [**Pie**, 14.4, Prop. b.ii]) that Exp(A) and Ind(A) have the same prime divisors.

We also make use of the *reduced characteristic polynomial*, and the *reduced norm* which is denoted as Nrd, cf. [**Pie**, Chap. 16], [**Rei**, Section 9].

When viewing an associative k-algebra as a Lie algebra over k with respect to the commutator bracket [a, b] := ab - ba, we denote it by the corresponding Fraktur letter: So the associative algebra A when viewed as Lie algebra is called \mathfrak{A} , etc.

Let k be any field and $X \in M_n(k)$. On the n^2 -dimensional k-vector space $M_n(k)$, consider the operator $\operatorname{ad}(X) \in \operatorname{End}_k(M_n(k))$ given by

$$\operatorname{ad}(X)(Y) := XY - YX.$$

The following lemma says that the eigenvalues of ad(X) are the pairwise differences of the eigenvalues of X.

LEMMA 4.4.1. — Assume that over an algebraic closure \bar{k} , the characteristic polynomial of X decomposes as $\chi_X(T) = \prod_{i=1}^n (T - \lambda_i)$. Then the characteristic polynomial of $\operatorname{ad}(X)$ over \bar{k} decomposes as

$$\chi_{\mathrm{ad}(X)}(T) = \prod_{1 \le i,j \le n} \left(T - (\lambda_i - \lambda_j) \right).$$

Proof. — For $X = \text{diag}(\lambda_1, \ldots, \lambda_n)$ this is easy, the elementary matrix E_{ij} being an eigenvector for $\lambda_i - \lambda_j$, and the general version can be derived by scalar extension to \bar{k} and Jordan forms. Compare e.g. [**Bo2**, VII.2.2, Example 3] for a conceptual or [**Lan1**, p. 47] for a computational proof.

Now let D be a k-central division algebra with n := Deg(D). Its (Lie-algebraic) centre $\mathfrak{z} = \mathfrak{z}(\mathfrak{D})$ is its (associative-algebraic) centre k. Assume that k is perfect.

PROPOSITION 4.4.2. — Every nilpotent element of \mathfrak{D} is contained in \mathfrak{z} .

Proof. — It is known that there exists a finite field extension K|k in \bar{k} such that we have a K-algebra isomorphism

$$K \otimes_k D \simeq M_n(K)$$

with regard to which we identify $D = 1 \otimes D$ with a sub-k-algebra of the right hand side. Let now $a \in D$ such that $a \in \mathfrak{D}$ is nilpotent; this means that $\operatorname{ad}(a) \in \operatorname{End}_K(M_n(K))$ is nilpotent when restricted to D. But D generates $M_n(K)$ as K-vector space, so $\operatorname{ad}(a)$ is nilpotent, whence its characteristic polynomial is:

$$\chi_{\mathrm{ad}(a)}(T) = T^{(n^2)}$$

The lemma implies that the characteristic polynomial of $a \in M_n(K)$ is

$$\chi_{M_n(K),a}(T) = (T-\lambda)^n$$

with some $\lambda \in \bar{k}$. Note that by definition, $\chi_a := \chi_{M_n(K),a}$ is the reduced characteristic polynomial of $a \in D$, which depends neither on the choice of the splitting field K nor on the splitting isomorphism $K \otimes_k D \simeq M_n(K)$, and whose coefficients are already in k: cf. [**Pie**, Prop. 16.1], [**Rei**, Thm 9.3]. On the other hand, the minimum polynomial $\min_{k,a}(T)$ divides $\chi_a(T)$ in k[T] (cf. [**Pie**, Lemma 16.3b and Exercise 16.3.3], [**Rei**, Exercise 9.1]). Thus in $\bar{k}[T]$ we have $\min_{k,a}(T) = (T - \lambda)^m$ for some $1 \le m \le n$. But since D is a skew field and k is perfect, a is contained in a separable field extension k(a)|k inside D. By Galois theory necessarily m = 1 and $a = \lambda \in k = \mathfrak{z}$. (The argument works more generally if char(k) does not divide n.)

COROLLARY 4.4.3. — Let char(k) = 0. $\mathfrak{D}/\mathfrak{z} \simeq \mathcal{D}\mathfrak{D}$ is either zero (namely, if and only if D = k, i.e. n = 1) or an anisotropic form of type A_{n-1} .

Proof. — Assume $k \subsetneq D$. Then with the splitting field K from above, we get $\mathfrak{D}_K \simeq \mathfrak{gl}_n(K)$, $\mathfrak{z}_K = \mathfrak{z}(\mathfrak{D}_K)$ and thus $(\mathfrak{D}/\mathfrak{z})_K \simeq \mathfrak{D}_K/\mathfrak{z}_K \simeq \mathfrak{sl}_n(K)$. If $x \in \mathfrak{D}/\mathfrak{z}$ is nilpotent, then so is $\mathrm{ad}_{\mathfrak{D}}(\widehat{x})$ for any representative \widehat{x} of x in \mathfrak{D} , so $\widehat{x} \in \mathfrak{z}$ by the proposition, hence x = 0. So $\mathfrak{D}/\mathfrak{z}$ is anisotropic.

Another corollary is [**Bo2**, VIII.10, Exercise 7b] and will be taken up in 5.1.4:

COROLLARY 4.4.4. — If over a field k with char(k) = 0 all semisimple Lie algebras are quasi-split, then every finite-dimensional division algebra D over k is a field; or in other words, Br(K) = 0 for every finite extension K|k.

Proof. — If D is not a field, let $K \subsetneq D$ be its centre. Apply the previous corollary for D over K to get the simple $R_{K|k}(\mathfrak{D}/\mathfrak{z})$ which contains no nilpotent element and thus is anisotropic.

4.5. Involutorial algebras and the classical types

For the "classical types" A-D there is a theory which relates forms of Lie algebras (and algebraic groups) to objects of associative algebra which in turn can be classified. This unified approach was laid out in [Wei], concentrating on the group case. The corresponding theory for Lie algebras can be found in our main source [Sel1, IV.3] but goes back to Landherr and Jacobson; for a version with little Galois cohomology, cf. [Jac6, X] and exercises 16 and 17 to [Bo2, VIII.13]. A modern resource for the vast theories surrounding this is [Inv]. From a Galois cohomological viewpoint, the upshot is that the automorphism groups of certain objects are naturally equivalent, which induces the correspondences.

In this section, k is generally assumed to be of characteristic $\neq 2$, although this is mostly for simplicity. When we go into details with Lie algebras, we restrict to $\operatorname{char}(k) = 0$.

DEFINITION 4.5.1 (Involutorial algebras, first and second kind)

- i. An involutorial k-algebra is a pair (A, ι) consisting of a finite dimensional kalgebra A and a k-linear map $\iota : A \to A$ satisfying $\iota^2 = \mathrm{id}_A$ and $\iota(xy) = \iota(y)\iota(x)$ for all $x, y \in A$. In this context, ι is called an involution (or "involutorial anti-automorphism"). For any field extension K|k, we have the scalar extension $(K \otimes_k A, \mathrm{id} \otimes \iota)$ which we call (A_K, ι_K) and which is an involutorial K-algebra. An isomorphism of involutorial k-algebras $f : (A, \iota) \to (B, j)$ is of course an isomorphism of k-algebras $f : A \to B$ with $j \circ f = f \circ \iota$.
- ii. An involutorial k-algebra (A, ι) is called *simple* if A contains no proper ι -stable ideal. It is called *absolutely simple* if (A_K, ι_K) is simple for all extensions K|k.
- iii. An absolutely simple involutorial k-algebra (A, ι) is called
 - \triangleright of the first kind if A is a central simple algebra over k. Otherwise, it is called
 - \triangleright of the second kind. Then either
 - $\triangleright \land A$ contains a proper ideal B, in which case it is easily seen that $A = B \oplus \iota(B)$ and B is a central simple k-algebra: such an (A, ι) is called of type AI. Or
 - $\triangleright \land A$ is simple, in which case one sees that its centre C is a quadratic field extension of k such that $\iota_{|C}$ is the non-trivial automorphism of C|k. Such an (A, ι) is called of type AII.

Note that (A_K, ι_K) is of the same kind as (A, ι) , and also of type AI if (A, ι) is. On the other hand, (A_C, ι_C) is of type AI if (A, ι) is of type AII with centre C. Also note that for the second kind, our terminology coincides with the one of [**Inv**, 2.B], but deviates slightly from the general terminology of [**Inv**], where what we call an involutorial k-algebra of type AII would be a central simple algebra over C with involution of the second kind (the ground field is different).

THEOREM 4.5.2 (Wedderburn, Albert). — Let (A, ι) be an involutorial k-algebra of the first kind (resp. of type AII). Then there is a unique $n \in \mathbb{N}_{\geq 1}$ and a division algebra D with centre k (resp. C), unique up to isomorphism, such that $A \simeq \operatorname{End}_D(D^n)$, and on D there is an involution I such that $I_{|k} = \operatorname{id}_k$ (resp. $I_{|C} = \iota_{|C}$); in particular, (D, I) is an involutorial k-algebra of the first kind (resp. of type AII).

Proof. — The Wedderburn part is found in any book about algebras. For the involution, see [**Alb**, Chap. II, Thm 5]. Remark that it is *not necessarily* so that I is the restriction of ι to D (or even that ι restricts to D) under a specific isomorphism $A \simeq \operatorname{End}_D(D^n)$, but there is one for which it is. Compare [**Inv**, Thm 3.1] for a more general variant.

This is complemented by the following crucial theorem. Recall the notions from Section 2.3.

THEOREM 4.5.3. — Let (D, σ) be an involutorial k-algebra such that D is a skew field. If it is of the first kind, there is a one-to-one correspondence

$$\begin{array}{c} \text{non-degenerate } \sigma \text{-hermitian and} \\ \sigma \text{-skew-hermitian forms on } D^m \\ (up \ to \ a \ factor \ in \ k^*) \end{array} \right\} \longleftrightarrow \left\{ \begin{array}{c} \text{involutions of the first kind} \\ \text{on the } k\text{-}CSA \ \text{End}_D(D^m) \end{array} \right.$$

If it is of the second kind (necessarily of type AII), there is a one-to-one correspondence

$$\left.\begin{array}{c} \text{non-degenerate} \\ \sigma\text{-hermitian forms on } D^m \\ (up \ to \ a \ factor \ in \ k^*) \end{array}\right\} \longleftrightarrow \left\{\begin{array}{c} \text{involutions } \iota \ of \ the \ second \ kind \\ on \ the \ C\text{-}CSA \ \operatorname{End}_D(D^m) \\ satisfying \ \iota|_C = \sigma|_C \end{array}\right.$$

Both correspondences are given by mapping a form h to $\iota : A \mapsto A^*$, where A^* is the (left) adjoint to A with respect to h.

Proof. — This is [**Inv**, Thm. 4.2] applied to our situation $(E = D, M = D^n, \theta = \sigma)$. (The left/right conventions in [**Inv**] are different from ours/Bourbaki's, but there is no essential difference.)

4.5.1. The first kind: Types *B*, *C* and *D*. — Let (A, ι) be of the first kind, so that *A* is a CSA over *k*, and let n = Deg(A).

REMARK 4.5.4. — Exp(A) is 1 or 2. Consequently, Ind(A) is a power of 2; if n is odd, it is 1 which means that A is split.

Proof. — The involution gives an isomorphism of k-algebras $A \simeq A^{\text{op}}$, and as is well-known, the class of A^{op} in Br(k) is the inverse of the class of A.

Set

$$Sym(A, \iota) := \{a \in A : \iota(a) = a\},$$

Skew $(A, \iota) := \{a \in A : \iota(a) = -a\}$

The following are straightforward:

LEMMA 4.5.5 (Basic properties)

- i. $A = \text{Sym}(A, \iota) \oplus \text{Skew}(A, \iota)$ as k-vector spaces.
- ii. Skew(A, ι) is stable under the commutator bracket, thus a Lie subalgebra of A (remember, this is the notation for A viewed as Lie algebra); when viewed as such, call it G(A, ι).
- iii. For any field extension K|k, we have a natural identification $\mathfrak{S}(A_K, \iota_K) = (\mathfrak{S}(A, \iota))_K$.

Now let us first look at the split case $A = M_n(k)$. By 4.5.3 (with $(D, \sigma) = (k, id)$), ι corresponds to a nondegenerate bilinear form b on k^n which is well-defined up to multiplication with a scalar in k^* and either symmetric or skew-symmetric. In the first case, call ι of orthogonal type, in the second case, of symplectic type.

In the non-split case, take the same definition after extension with a splitting field, by which we mean an extension K|k such that $A_K \simeq M_n(K)$. The type does not depend on the choice of such a splitting field and isomorphism. Indeed we have (cf. [Inv, Prop. 2.6]):

LEMMA 4.5.6. — ι is of orthogonal type if and only if

$$\dim_k(\operatorname{Skew}(A,\iota)) = \frac{n(n-1)}{2}.$$

 ι is of symplectic type if and only if

$$\dim_k(\operatorname{Skew}(A,\iota)) = \frac{n(n+1)}{2}.$$

Moreover, if ι is of symplectic type, n is even.

(The last remark just comes from the fact that on an odd-dimensional vector space, all skew-symmetric forms are degenerate.)

From now on, let char(k) = 0.

▷ For odd $n (\geq 3)$, A is split by 4.5.4 and ι is of orthogonal type. For an appropriate extension K|k (in fact, it can be chosen as a composite of quadratic extensions), there is an isomorphism $A_K \simeq M_n(K)$ such that the symmetric bilinear form b_K corresponding to ι_K is given by $b_K(x, y) = {}^t x S y$ where

$$S := \begin{pmatrix} 0 & I_{\nu-1} \\ I_{\nu-1} & 0 \\ & & 1 \end{pmatrix}$$

with $n = 2\nu + 1$. The involution ι_K is then given by $A \mapsto S^{-1}({}^tA)S$. Thus $\mathfrak{S}(A_K, \iota_K) = (\mathfrak{S}(A, \iota))_K$ is, by the classical description, ⁽⁴⁾ a split Lie algebra of type B_{ν} .

▷ For even n and ι of symplectic type, as soon as A_K is split, by [Bo1, IX.5.1] there is an isomorphism $A_K \simeq M_n(K)$ such that the symplectic form b_K is given by $b_K(x,y) = {}^t x S y$ where

$$S := \begin{pmatrix} 0 & I_{\nu} \\ -I_{\nu} & 0 \end{pmatrix}$$

with $n = 2\nu$. Analogous to above, $(\mathfrak{S}(A, \iota))_K$ is a split Lie algebra over K of type C_{ν} .

 \triangleright For even $n (\geq 4)$ and ι of orthogonal type, after extension with a splitting field of A, and then (analogously to type B) a series of quadratic extensions, for b_K we now have a matrix

$$S := \begin{pmatrix} 0 & I_{\nu} \\ I_{\nu} & 0 \end{pmatrix}$$

and $(\mathfrak{S}(A,\iota))_K$ is split of type D_{ν} for $n = 2\nu$.

We thus have:

PROPOSITION 4.5.7. — $\mathfrak{S}(A, \iota)$ is an absolutely simple Lie algebra of respective type B, C or D in the cases considered above.

The great thing is that this construction gives all k-forms of absolutely simple Lie algebras of the respective types (except for trialitarian types D_4). The crucial facts we need for this, first noted by Jacobson, are the next two lemmata.

LEMMA 4.5.8. — Skew (A, ι) generates A as associative k-algebra (i.e. the smallest sub-k-algebra of A containing it is the whole algebra A.)

Proof. — See [Inv, Lemma 2.26 (1)]. One reduces by scalar extension to standard split cases, which are done by direct matrix calculations. \Box

LEMMA 4.5.9. — Let $\mathfrak{S}(A, \iota)_K$ be split as above. If it is not of type D_4 , then every $\mathfrak{f} \in \operatorname{Aut}((\mathfrak{S}(A, \iota)_K))$ is induced by a unique automorphism f of (A_K, ι_K) , which in turn is of the form $X \mapsto F^{-1}XF$ for $F \in \operatorname{GL}_n(K)$ with $\iota_K(F)F = S^{-1}({}^tF)SF \in K^* \cdot I_n$.⁽⁵⁾ For type D_4 , the same holds for \mathfrak{f} which are of order ≤ 2 modulo $\operatorname{Aut}_0((\mathfrak{S}(A, \iota)_K))$.

Proof. — See e.g [Sel1, Lemma IV.3.1] except for the D_4 case which is easily amended. One uses the explicit description of automorphisms of the split Lie algebras (cf. [Bo2, VIII.13]) to see that they are of the form $X \mapsto F^{-1}XF$ for F as described; with the previous lemma, their extension to automorphisms of (A, ι) is unique. (One

^{4.} See e.g. [Jac6, IV.6]. There are different but equivalent normalisations in [Bo2, VIII.13]. Without coordinates, the split forms of types B and D are those where b corresponds to a quadratic form of maximal Witt index $\nu = \lfloor \frac{1}{2}n \rfloor$.

^{5.} F is a (matrix of) similate with respect to (S or) b_K , i.e. $b_K(Fx, Fy) \in K^* \cdot b_K(x, y)$.

can even descend to an analogous result for non-split Lie algebras, cf. [Jac3, Thm 3] and [Inv, Prop. 2.25].) \Box

This shows by the general Galois cohomology machinery that the forms of $\mathfrak{S}(A, \iota)$ and the forms of (A, ι) are in correspondence:

THEOREM 4.5.10 (Jacobson, Weil). — Let \mathfrak{g} be an absolutely simple Lie algebra over k of type B_{ν} , C_{ν} or D_{ν} (in the case D_{ν} , $\nu \geq 2$ and non-trialitarian). Then there is an involutorial k-algebra of the first kind (A, ι) as described in the respective cases above such that $\mathfrak{g} \simeq \mathfrak{S}(A, \iota)$.

Proof. — Let K|k be a finite Galois splitting field for \mathfrak{g} . By the discussion after 4.2.1 we have a cocycle $\sigma \mapsto a_{\sigma}$ from $\operatorname{Gal}(K|k)$ into $\operatorname{Aut}(\mathfrak{S}(M_n(K), \iota_K))$, with ι_K given via a matrix S as in the descriptions of the split form of the respective type; and \mathfrak{g} is isomorphic to the fixed elements of $\mathfrak{S}(M_n(K), \iota_K)$ under all $a_{\sigma} \circ \sigma$. Now by the lemma, we view the cocycle as one from $\operatorname{Gal}(K|k)$ to $\operatorname{Aut}(M_n(K), \iota_K)$, and define Ato be the fixed set in $M_n(K)$ of all $a_{\sigma} \circ \sigma$, $\sigma \in \operatorname{Gal}(K|k)$. One checks immediately that $(A, \iota_{K|A})$ is an involutorial k-algebra of the respective type, and with 4.5.8 one sees that $\mathfrak{g} \simeq \mathfrak{S}(A, \iota_{K|A})$.

REMARK 4.5.11. — We do not address the question how *unique* such an (A, ι) is – which is dealt with extensively in Jacobson's work – because here we are only interested in existence, which provides us with k-admissible diagrams. In most cases (see the later remark 4.5.20 for typical exceptions), the (A, ι) to a form \mathfrak{g} is unique up to isomorphism; the condition "up to isomorphism" translates to equivalence modulo scaling ("cogredience" in Jacobson's terminology) of certain (skew-)hermitian forms. But for a \mathfrak{p} -adic field k, admissibility conditions and the isomorphism theorem will imply that we almost only need a more careful look at the "uniqueness" of the analogous construction for type A, taken up later.

Now we connect the theorem with the classification of Chapter 3: what is the index of a Lie algebra $\mathfrak{S}(A, \iota)$?

4.5.1.1. The index of $\mathfrak{S}(A, \iota)$. — Let $d := \operatorname{Ind}(A)$ and m = n/d. We have an isomorphism $A \simeq \operatorname{End}_D(D^m)$, where D is a k-division algebra Brauer equivalent to A. By 4.5.2, there is an involution of the first kind on D. In the split case D = k it is the identity, and we had already seen that the involutions of orthogonal/symplectic type on A are induced by symmetric bilinear/symplectic forms on k^m . If, on the other hand, $d \geq 2$, we assume that the involution on D is of symplectic type ⁽⁶⁾ and call it γ . Then we have the following refined version of 4.5.3, also part of [Inv, Thm 4.2.1]:

^{6.} This is always possible by composing with an inner automorphism (cf. [Inv, Prop. 2.7.3]) and will be the canonical choice in later applications, where D is a quaternion algebra.

THEOREM 4.5.12. — There are one-to-one correspondences

$$\begin{array}{c} non-degenerate\\ \gamma-skew-hermitian forms on D^m\\ (up \ to \ a \ factor \ in \ k^*) \end{array} \right\} \longleftrightarrow \begin{cases} involutions \ of \ the \ first \ kind\\ of \ orthogonal \ type\\ on \ the \ k-CSA \ \operatorname{End}_D(D^m) \end{cases} \\ \\ \begin{array}{c} non-degenerate\\ \gamma-hermitian \ forms \ on \ D^m\\ (up \ to \ a \ factor \ in \ k^*) \end{array} \right\} \longleftrightarrow \begin{cases} involutions \ of \ the \ first \ kind\\ of \ symplectic \ type\\ on \ the \ k-CSA \ \operatorname{End}_D(D^m) \end{cases}$$

Now let h(.,.) be a symmetric bilinear form on k^m or a γ -(skew-)hermitian form on D^m . Write V for k^m or D^m in these cases, and let ι be the left adjoint with respect to h on A, that is, $h(v, aw) = h(\iota(a)v, w)$ for all $a \in A, v, w \in V$.

LEMMA 4.5.13. — If h(.,.) is anisotropic and $\mathfrak{g} := \mathfrak{S}(A,\iota)$ is semisimple (which by our earlier results is the case for $n \geq 3$), then \mathfrak{g} is anisotropic.⁽⁷⁾

Proof. — V naturally is a \mathfrak{g} -module as \mathfrak{g} is a subalgebra of \mathfrak{A} . Assume that there is a nilpotent element $x \neq 0$ in \mathfrak{g} . By Jacobson-Morozov, x is part of an \mathfrak{sl}_2 -triple (x, H, y) in \mathfrak{g} , so that V also becomes a finite dimensional $\mathfrak{sl}_2(k)$ -module, and decomposes into eigenspaces V_λ with $\lambda \in \mathbb{Z}$ for the operation of H. Since $0 \neq H \in A$, we can take λ different from 0. Then for all $v \in V_\lambda \neq \{0\}$ we have

$$-\lambda h(v,v) = h(-\lambda v,v) = h(-Hv,v) = h(\iota(H)v,v) = h(v,Hv) = h(v,\lambda v) = \lambda h(v,v)$$

which implies h(v, v) = 0, a contradiction to h being anisotropic.

Now let us first investigate the case d = 1 or equivalently, the CSA A is split over k. h(.,.) is a symmetric bilinear form which we now call b(.,.). \mathfrak{g} is of type $B_{\lfloor \frac{1}{2}n \rfloor}$ for odd n and of type $D_{n/2}$ for even n. By Witt decomposition, we can set $V = k^n$ together with a matrix

$$S := \begin{pmatrix} 0 & I_r & 0 \\ I_r & 0 & 0 \\ 0 & 0 & S_a \end{pmatrix}$$

such that r is the Witt index of $b(v, w) = ({}^tv)Sw$, and $S_a \in M_{n-2r,n-2r}(k)$ defines the anisotropic form $b_a(v_a, w_a) = ({}^tv_a)S_aw_a$ on k^{n-2r} . The corresponding involution (the left adjoint with respect to b) on $A = M_n(k)$ is given by

$$\iota(X) = {}^{t}S^{-1} \cdot {}^{t}X \cdot {}^{t}S.$$

A computation with the relation $\iota(X) \stackrel{!}{=} -X$ shows that our \mathfrak{g} consists of those matrices of the form

^{7.} The converse is also true, as will become clear in the following.

$$\left(\begin{array}{ccc}
A & B & G \\
C & -(^{t}A) & H \\
E & F & J
\end{array}\right)$$

where $A, B, C \in M_r(k), E, F \in M_{n-2r \times r}(k), G, H \in M_{r \times n-2r}(k)$, with the following relations: ${}^{t}B = -B, {}^{t}C = -C$ (i.e. B and C are skew-symmetric), ${}^{t}F \cdot {}^{t}S_a = -G$, ${}^{t}E \cdot {}^{t}S_a = -H$, and ${}^{t}S_a^{-1} \cdot {}^{t}J \cdot {}^{t}S_a = -J$. If we assume, as we always can, S_a to be a diagonal matrix diag (b_1, \ldots, b_{n-2r}) , the last relation means that the (i, j)-entry of Jis its (j, i)-entry times $-b_i^{-1}b_j$; in particular, the diagonal of J is zero.

We visibly have the r-dimensional k-split toral subalgebra

$$\mathfrak{s} := \left\{ \operatorname{diag}(a_1, \dots, a_r, -a_1, \dots, -a_r, 0, \dots, 0) : a_i \in k \right\}$$

and will see that it is maximal split toral. Indeed, one checks that its centraliser – which contains any larger toral subalgebra – is

$$\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \left\{ \begin{pmatrix} a_1 & & & & \\ & \ddots & & 0 & \\ & a_r & & & \\ & & -a_1 & & \\ & & & \ddots & \\ & 0 & & & -a_r & \\ & & & & & J \end{pmatrix} : J \text{ as above, } a_i \in k \right\}.$$

So the matrices J in the lower right corner give the Lie algebra $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})/\mathfrak{s}$. For $n-2r \geq 3$, we had seen in the lemma that this is an anisotropic semisimple Lie algebra, hence \mathfrak{s} is maximal split toral in \mathfrak{g} and $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{s} \oplus \mathfrak{g}_a$ for the anisotropic kernel \mathfrak{g}_a of \mathfrak{g} (with respect to \mathfrak{s}), which consists of the matrices with entries 0 outside of the block J. Setting $\mathfrak{t} := \mathfrak{s} \oplus \mathfrak{t}_a$ for \mathfrak{t}_a maximal toral in \mathfrak{g}_a , and fixing the usual root basis in a split extension, one concludes that in the Satake-Tits diagram, the vertices $\alpha_1, \ldots, \alpha_r$ are white, the remaining ones are black.

In the cases n - 2r = 0 and n - 2r = 1 (where J = 0), we get the split forms of type D_r and B_r , respectively.

In the remaining case n = 2r + 2, $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})/\mathfrak{s}$ is one-dimensional, in particular abelian, and we must have a non-split quasi-split form of type $D_{n/2}$ with maximal toral subalgebra $\mathfrak{t} = \mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$. The anisotropic part of the space, with $S_a = \operatorname{diag}(b_1, b_2)$, corresponds to the form $b_1 X^2 + b_2 Y^2$, with discriminant $d = -b_1 b_2 \cdot k^{*2} \notin k^{*2}$. (Note that on an even-dimensional vector space, the discriminant is invariant under scaling.) Visibly $K := k(\sqrt{-b_1 b_2})$ is the minimal splitting field, and the non-trivial element of $\operatorname{Gal}(K|k)$ under the twisted action t must correspond to the non-trivial automorphism of the Dynkin diagram that flips the two outer vertices α_r and α_{r+1} .

Before taking up the case d > 1, with these results we can already settle

4.5.1.2. Type B_{ν} . — Because by 4.5.4 the CSA A is automatically split. We thus see from the above discussion that the corresponding Lie algebra of type B_{ν} has the following Satake-Tits diagram with total number of vertices $\nu = \frac{n-1}{2}$:



REMARK 4.5.14. — This is the most direct connection to the theory and nomenclature of quadratic forms, and might explain why Tits calls his version of the isomorphism theorem a "Witt-type theorem": In this example, all the information of the (Satake-Tits) index is in the (Witt) index r; and the isomorphism class of the anisotropic kernel (of the Lie algebra) is determined by the isomorphism class of the anisotropic kernel (of the quadratic form). Witt's decomposition theorem appears here as suggestive special case of the isomorphism theorem.

4.5.1.3. Type D_{ν} . — Here, the CSA A might be split or not. If it is, the previous discussion applies. For $r = \nu = \frac{1}{2}n^2$, we have "the" split Lie algebra. If $r = \nu - 1$, we get the quasi-split forms



with minimal splitting field $k(\sqrt{d})$ where d is a representative in k of the discriminant of the bilinear form that defines (A, ι) ; each quadratic extension can occur. As soon as $r \leq \nu - 2$, the Satake-Tits diagram is either



and this depends on the discriminant (which, recall, on an even-dimensional space is invariant under scaling):

LEMMA 4.5.15. — Let $\nu \geq 2, n = 2\nu$ and (A, ι) be given by a symmetric bilinear form b on k^n with Witt index r as above. The Lie algebra $\mathfrak{g} = \mathfrak{S}(A, \iota)$ is an outer form of type D_{ν} (i.e. there is an arrow in the Satake-Tits diagram) if and only if the discriminant d(b) is not in k^{*2} . In this case, the fixed field of the twisted Galois action t is $k(\sqrt{d})$ where d is a representative of d(b) in k^* .

MÉMOIRES DE LA SMF 151

or

Proof. — By induction on $\mu := \nu - r$ which is half the dimension of the anisotropic kernel of b (and of \mathfrak{g}_a for $\mu \geq 2$). We know the assertion for $\mu = 0$ (split case) and $\mu = 1$ (non-split quasi-split case) where d(b) was necessarily in k^{*2} resp. not in k^{*2} . Now let $\mu \geq 2$. Resuming the notations of the general discussion, we can assume that the $2\mu \times 2\mu$ -matrix S_a is diagonal, say $S_a = \operatorname{diag}(b_1, b_2, \ldots, b_{2\mu})$. By definition, d(b) is the residue of $(-1)^{\mu} \cdot b_1 \cdots b_{2\mu} \mod k^{*2}$.

Remark that for any pair of indices $1 \le i \ne j \le 2\mu$, the scalar $-b_i b_j$ or equivalently $-b_i/b_j$ cannot be a square in k^* , since otherwise the form defined by S_a would have the isotropic vector ${}^t(0,\ldots,0,1,0,\ldots,0,x,0,\ldots,0)$ with 1 the *i*-th and $x := \sqrt{-b_i/b_j}$ the *j*-th entry.

Set $K(i, j) := k(\sqrt{-b_i b_j})$. Then at least the corresponding two-dimensional space in $V_{K(i,j)}$ is hyperbolic with respect to $b_{K(i,j)}$, and the discriminant is

$$d(b_{K(i,j)}) = (-1)^{\mu-1} \cdot \prod_{\ell \neq i,j} b_{\ell} \equiv d(b) \mod K(i,j)^{*2}$$

Now there are two cases:

- ▷ There is a pair (i, j) such that d(b_{K(i,j)}) ∉ K(i, j)^{*2}. Then certainly d(b) ∉ k^{*2} on the one hand, and on the other hand by the induction hypothesis, 𝔅(A, ι)_{K(i,j)} is an outer form, a fortiori 𝔅(A, ι) is an outer form and all assertions "descend".
 ▷ d(b_{K(i,j)}) ∈ K(i, j)^{*2} for all pairs (i, j). We make a further distinction:
 - ▷▷ $d(b) \notin k^{*2}$. But above we saw $d(b) \in K(i,j)^{*2}$ for all pairs, and K(i,j)|kis quadratic, so $k(\sqrt{-b_i b_j}) = K(i,j) = k(\sqrt{d})$ for $d = (-1)^{\mu} \prod b_{\ell}$. This means that all $-b_i b_j$ are congruent (to d and thus to each other) mod k^{*2} . Then $1 \equiv (-b_i b_j)(-b_j b_{\ell}) \equiv b_i b_{\ell} \mod k^{*2}$, hence all b_i are in the same square class, hence by scaling and equivalence we can assume that all $b_i = 1$. With $d(b) \notin k^{*2}$ this is impossible unless μ is odd and $-1 \notin k^{*2}$, a typical "real" case which can be checked by hand (the non-trivial automorphism of $k(\sqrt{-1})|k$ induces the opposition involution on the diagram).
 - ▷▷ $d(b) \in k^{*2}$. We have to show that our Lie algebra is an inner form. By the induction hypothesis, we know that all $\mathfrak{S}(A, \iota)_{K(i,j)}$ are inner forms. So if we had a non-trivial *t*-action, its fixed field would have to be a proper extension of k contained in all K(i, j), so they would all be equal. Like before we conclude that without loss of generality, all $b_i = 1$, and the *t*-fixed field and all K(i, j) must be $k(\sqrt{-1})$, in particular $-1 \notin k^{*2}$ which now gives a contradiction for odd μ . For even μ , one again checks that the non-trivial automorphism induces the opposition involution on the diagram which for even μ , however, is the identity, contradicting our assumption.

To complete the discussion of type D_{ν} , we now sketch the case d > 1 or equivalently, A is non-split. Then $A \simeq \operatorname{End}_D(D^m)$ and we see that by the first case of 4.5.12, our

Lie algebras come from (and, cf. 4.5.11, are classified "up to scaling in k" by) γ -skewhermitian forms on D^m , where we assume γ to be of symplectic type. Let such a form have index r, then our Lie algebra is given by those matrices $X \in A = M_m(D)$ such that $\iota(X) = {}^t H^{-1} \cdot {}^t(\gamma(X)) \cdot {}^t H = -X$ where H is a γ -skew-hermitian matrix. The coordinates can be chosen such that

$$H := \begin{pmatrix} 0 & -I_r & 0 \\ I_r & 0 & 0 \\ 0 & 0 & H_a \end{pmatrix}.$$

where H_a defines an anisotropic skew-hermitian form on D^{m-2r} . Then \mathfrak{g} consists of the matrices

$$\begin{pmatrix} A & B & G \\ C & -^t(\gamma(A)) & K \\ E & F & J \end{pmatrix}$$

where $A, B, C \in M_r(D)$, $E, F \in M_{m-2r \times r}(D)$, $G, K \in M_{r \times m-2r}(D)$ are subject to the following relations: ${}^{t}B = \gamma(B)$, ${}^{t}C = \gamma(C)$ (i.e. B and C are γ -hermitian), ${}^{t}\gamma(F) \cdot {}^{t}H_a = G$, ${}^{t}\gamma(E) \cdot {}^{t}H_a = -K$, and ${}^{t}H_a^{-1} \cdot {}^{t}\gamma(J) \cdot {}^{t}H_a = -J$. If we assume, as we always can, H_a to be a diagonal matrix diag (b_1, \ldots, b_{n-2r}) – where necessarily the $b_i \in \text{Skew}(D, \gamma)$ –, the last relation means that the (i, j)-entry of J is the γ -conjugate of its (j, i)-entry times $-b_i^{-1}b_j$; in particular, the diagonal of J consists of elements of Skew (D, γ) .

A split toral subalgebra of k-dimension r is given by

$$\mathfrak{s} := \{ \operatorname{diag}(a_1, \dots, a_r, -a_1, \dots, -a_r, 0, \dots, 0) : a_i \in k \}$$

and its centraliser $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ consists of the matrices



with J as above and $c_i \in D$. Let \mathfrak{u} be those matrices as above with all $c_i \in D \setminus k$, so that $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) = \mathfrak{s} \oplus \mathfrak{u}$. By 4.5.13, as soon as 2r < m (for d > 2) or 2r < m - 1 (for d = 2), the matrices with zeros outside of J describe an anisotropic subalgebra \mathfrak{a} , but the anisotropic kernel is bigger: it is all of \mathfrak{u} , as one can show with the investigation of skew fields in Section 4.4 (compare the later discussion of type AI), and it decomposes as $(\mathcal{D}\mathfrak{D})^r \oplus \mathfrak{a}$. This \mathfrak{a} on its own is a form of type D_{μ} with $\mu = \frac{1}{2}(n - 2rd) = \nu - rd$. One infers that the general Satake-Tits diagram is



and again the occurrence of an arrow depends on whether the discriminant of the γ -skew-hermitian form, which in this case can be well-defined as

 $(-1)^m \cdot \operatorname{Nrd}(\det(h(v_i, v_i))) \mod k^{*2}$

for a *D*-basis v_1, \ldots, v_m of *V* (cf. [**Tsu**, 1.3]), is in k^{*2} or not. If not, there is an arrow and its fixed field is the corresponding quadratic extension.

In the special case m = 2r, equivalently $\nu = rd$, the anisotropic space V_a vanishes and this leads to a diagram with white vertices $\alpha_d, \alpha_{2d}, \ldots, \alpha_{rd}$, and all others black. For example for d = 2 we get:



Note the consistency with 4.3.1: Since d is a power of 2 by 4.5.4, the rank $\nu = rd$ is *even*, so the opposition involution does *not* flip the two rightmost vertices $\alpha_{\nu-1}$ (black) and α_{ν} (white).

The last exceptional case is d = 2 and $\nu = m = 2r + 1$ (hence $r \ge 1$, since a type D_1 does not exist). Here D = Q is a quaternion division algebra and γ is the *standard involution* from Section 2.2. Indeed, by [**Inv**, Prop. 2.21], this is the unique symplectic involution on Q, whereas the orthogonal ones are those of the form $\iota_u(x) = u \cdot \gamma(x) \cdot u^{-1}$ where u is an invertible element of Skew (Q, γ) which is uniquely determined by ι_u up to a factor in k^* . Now the space V_a is of Q-dimension 1, the anisotropic form h_a on it is given by $h_a(v, w) = vH_a\gamma(w)$ with a pure quaternion $H_a = q_a \neq 0$, unique up to scaling with k^* , and the condition $q_a^{-1} \cdot \gamma(J) \cdot q_a = -J$ is equivalent to $J \in k \cdot q_a$. One calculates that the Satake-Tits diagram is



and the fixed field of the arrow is the well-defined quadratic extension $k(\sqrt{a})$ where $a = -\operatorname{Nrd}(q_a)$, a representative of the discriminant.

4.5.1.4. Type C_{ν} . — As noted in the split cases after lemma 4.5.6, if the CSA A is split, then $\mathfrak{S}(A, \iota)$ is the split form of type C_{ν} with $n = 2\nu$.

Otherwise, if $d \ge 2$, for type C_{ν} we are interested in the second case of 4.5.12 and see that our Lie algebras now come from γ -hermitian forms. The discussion is very similar to the one above, and we just give the result: If such a form has index r, the Satake-Tits diagram of the corresponding Lie algebra is



for $\nu > rd$ respectively $\nu = rd$. Again d is a power of 2 by 4.5.4, and of course d divides $n = 2\nu$.

As a final remark, the case d > 1 is strongly restricted for fields of interest:

REMARK 4.5.16. — If in Br(k) there is only one element of order 2, then by 4.5.2 and 4.5.4, in the above cases necessarily $D \simeq D^{\text{op}}$ is "the" division algebra representing this element. Specifically, if k is \mathbb{R} or a p-adic field, D is the quaternion division algebra over k and d = 2 (if not 1) in types C_{ν} and D_{ν} above.

4.5.2. The second kind: Type A. — Let (A, ι) be of the second kind. If it is of type AI with $A \simeq B \oplus \iota(B)$, we set n := Deg(B); if it is of type AII with centre C set n := Deg(A) with A viewed as CSA over C. So in both cases we have $\dim_k A = 2n^2$, and n is "stable under scalar extension". Setting again

$$Skew(A,\iota) := \{a \in A : \iota(a) = -a\}$$

we see that this is stable under the commutator bracket and thus a Lie algebra $\mathfrak{S}(A, \iota)$ over k; also, $\mathfrak{S}(A_K, \iota_K) \cong (\mathfrak{S}(A, \iota))_K$. Here we have $\dim_k \mathfrak{S}(A, \iota) = n^2$ (for both types). Skew (A, ι) generates A as k-algebra as soon as n > 1 for type AI and n > 2for type AII, cf. [Jac6, X.4, Lemma 4].

Let us first look at an (A, ι) of type AI, i.e. $A \simeq B \oplus \iota(B)$ for a central simple k-algebra B. Then $\iota(B) \simeq B^{\text{op}}$ and we can identify (A, ι) with $(B \times B^{\text{op}}, (a, b^{\text{op}}) \mapsto (b, a^{\text{op}}))$ (cf. [Inv, 2.14]); or specifically if $B \simeq M_n(k)$, we can also arrange $\iota(X, Y) = ({}^tY, {}^tX)$. It is obvious that in this case, the skew elements identify with the "antidiagonal" $(b, -b^{\text{op}})$ (resp. $(X, -{}^tX)$) and thus with a copy of B; hence the corresponding Lie algebra is $\mathfrak{S}(A, \iota) \simeq \mathfrak{B}$. If $n \ge 2$ and $B \simeq M_r(D)$ with D a central k-division algebra of degree d (so n = rd), we thus have

$$\mathfrak{S}(A,\iota) \simeq \mathfrak{gl}_r(D) \text{ (resp. } \mathfrak{D} \text{ for } r = 1),$$
$$\mathfrak{g} := \mathcal{D}\mathfrak{S}(A,\iota) \simeq \mathcal{D}\mathfrak{gl}_r(D) \text{ (resp. } \mathcal{D}\mathfrak{D} \text{ for } r = 1)$$

and if K|k splits D (i.e. $D_K \simeq M_d(K)$) we have

$$\mathfrak{g}_K \simeq \mathfrak{sl}_n(K)$$

so that \mathfrak{g} is a form of type A_{n-1} .

Now since for (A, ι) of type AII with centre C, (A_C, ι_C) is of type AI and thus can be further split as above, we see that $\mathcal{D}\mathfrak{S}(A, \iota)$ is an absolutely simple Lie algebra of type A_{n-1} . Again we have a converse [Sel1, Lemma and Thm 4.3.1]:

THEOREM 4.5.17 (Landherr, Jacobson, Weil). — Let $n \ge 2$.

i. Let $(A_K, \iota_K) \simeq (M_n(K) \times M_n(K), (x, y) \mapsto ({}^ty, {}^tx))$ so that $\mathcal{D}\mathfrak{S}(A, \iota)_K \simeq \mathfrak{sl}_n(K)$. Every $\mathfrak{f} \in \operatorname{Aut}((\mathfrak{S}(A, \iota)_K))$ is induced by a unique automorphism f of (A_K, ι_K) , which in turn is of the form

$$(X,Y)\longmapsto (F^{-1}XF,{}^t(F^{-1t}YF)) \quad or \quad (X,Y)\longmapsto (F^{-1}YF,{}^t(F^{-1t}XF))$$

for an $F \in \operatorname{GL}_n(K)$.

 ii. Like in 4.5.10 it follows: let g be an absolutely simple Lie algebra of type A_{n-1}. Then there is an involutorial k-algebra of the second kind (A, ι) with dim_k A = 2n² such that g ≃ D𝔅(A, ι).

Remark that $x \mapsto -tx$, the non-trivial element of $\operatorname{Aut}(\mathfrak{sl}_n(K)) / \operatorname{Aut}_0(\mathfrak{sl}_n(K))$ for $n \geq 3$, corresponds to $(X, Y) \mapsto (Y, X)$ under i.

COROLLARY 4.5.18. — If $\mathfrak{g} \simeq \mathcal{D}\mathfrak{S}(A,\iota)$ as in part ii of the theorem, \mathfrak{g} is an inner form (of the split $\mathcal{D}\mathfrak{S}(A_K,\iota_K)$ as in part i) if and only if (A,ι) is of type AI.

Proof. — If \mathfrak{g} is inner, then by what we just remarked the automorphisms in the image of a corresponding cocycle are all of the first form described in the theorem: They stabilise both factors. Then so do their corresponding Galois semi-automorphisms. Hence their fixed point set A visibly is a product of two non-trivial factors. Conversely, assume $A = A_1 \times A_2$ with non-trivial factors and identify it with a subset of a split A_K as in the theorem. A_{1K} and A_{2K} must correspond to the two factors $M_n(K)$; but if in the image of a cocycle corresponding to A there were an automorphism that flips the factors, so would the corresponding Galois semi-automorphism; and for every element $(a_1, a_2) \in A_1 \times A_2$, being fixed by it, a_1 would be determined by a_2 which is absurd.

The distinction between AI and AII goes back to Landherr, who originally (cf. [Lan1, p. 50]) distinguished the corresponding Lie algebras by characteristic polynomials of their matrix representations. — As promised in 4.5.11 we are now also interested in how unique an (A, ι) for \mathfrak{g} as in the theorem is:

PROPOSITION 4.5.19. — Let (A_1, ι_1) and (A_2, ι_2) be absolutely simple involutorial k-algebras of the second kind, of k-dimensions $2n_1^2$ resp. $2n_2^2$, $n_i \ge 2$, such that there is an isomorphism of k-Lie algebras $\mathfrak{f} : \mathfrak{S}(A_1, \iota_1) \simeq \mathfrak{S}(A_2, \iota_2)$. Then \mathfrak{f} is induced by

an isomorphism $f: (A_1, \iota_1) \simeq (A_2, \iota_2)$, and f is unique except for the case that the algebras are of type AII and $n_i = 2$. In particular, if the (A_i, ι_i) are of type AI and given as $A_i \simeq B_i \times B_i^{\text{op}}$, then either $B_1 \simeq B_2$ or $B_1 \simeq B_2^{\text{op}}$.

Proof. — See [Jac6, X.4, Thms 10 and 11] or even [Lan1, Satz 3] (for type AI), cf. also [Inv, Prop. 2.25]. The strategy – which works for both types like we (or rather Weil) defined them, rendering Jacobson's distinction unnecessary – is to embed everything in a common split (A_K , ι_K) resp. the Lie algebra of its skew elements. Then, balancing cleverly the distinct Galois actions, one reduces to the case of automorphisms of this, which we had settled in the theorem. Now one descends back to the given forms. This shows existence of the isomorphism, and uniqueness follows again because the associative algebras in question are generated by their skew elements. \Box

Remark 4.5.20

i. In low degrees, there exist exceptional isomorphisms among the first kind $("B_2 = C_2")$ which we have circumvented by not investigating uniqueness there at all, see 4.5.11. There also exist exceptional isomorphisms intertwining the two kinds $("A_1 = B_1 = C_1, D_2 = A_1 \times A_1, A_3 = D_3")$ which we have circumvented by restricting the proposition above to "inside" the second kind. Since in the end we will classify not by involutorial algebras, but by indices and anisotropic kernels, our results not only suffice, but even will imply e.g. the following identities of anisotropic forms over a \mathfrak{p} -adic field k, where Q is its quaternion division algebra with standard involution γ :

(a) (" $A_1 = C_1 = B_1$ ") the derived Lie algebra of Q over k is isomorphic to $\mathfrak{S}(Q, \gamma)$ (this is immediate!), and to $\mathfrak{S}(M_3(k), \iota)$ where ι is the adjoint with respect to any anisotropic quadratic form on k^3 (specifically, the reduced norm on the pure quaternions; every other is cogredient to this); ⁽⁸⁾;

(b) (" $A_3 = D_3$ ") the derived Lie algebra of a 16-dimensional k-central division algebra (there are two anti-isomorphic ones, which by the proposition give the same Lie algebra) is isomorphic to $\mathfrak{S}(M_3(Q), \iota)$ where ι comes from any anisotropic γ -skew-hermitian form on Q^3 (in fact there is only one). For starters one sees that both Lie algebras have dimension 15.

ii. Another instance where the classification with involutorial algebras falls short is related to triality (cf. [**Jac6**, X, Exercise 3]): Over $k = \mathbb{R}$, the Lie algebras $\mathfrak{S}(M_4(\mathbb{H}), \iota)$ (with ι coming from any skew-hermitian form on \mathbb{H}^4) and $\mathfrak{S}(M_8(\mathbb{R}), \iota)$ (with ι coming from a quadratic form on \mathbb{R}^8 of Witt index 2) are isomorphic, although the involutorial algebras are not. This will also appear over **p**-adic fields k, see 5.4.7. One look at the Satake-Tits diagrams gives an idea:

^{8.} This is true verbatim for $k = \mathbb{R}, Q = \mathbb{H}$; and is nothing else than example 3.2.5.



Turning back to type A, we can now classify both types and describe their diagrams.

4.5.2.1. The inner forms/Type AI

PROPOSITION 4.5.21 (cf. 4.3.7). — If \mathfrak{g} is an inner form of type A_n over k, then there is a k-central division algebra D, $\operatorname{Deg}(D) = d$, and an $r \in \mathbb{Z}_{\geq 0}$ such that d(r+1) = n+1 and \mathfrak{g} is isomorphic to the derived algebra of $M_{r+1}(D)$ viewed as Lie algebra over k. The isomorphism class of \mathfrak{g} determines D up to isomorphism or anti-isomorphism. Conversely, such derived algebra of $M_{r+1}(D)$ is an inner form of type A_n . Its rational root system is of type A_r , and its Satake-Tits diagram is:



In particular, every anisotropic inner form of type A_n is isomorphic to the derived Lie algebra of a k-central division algebra D of degree n + 1; two such division algebras D_1, D_2 have isomorphic derived Lie algebras if and only if $D_1 \simeq D_2$ or $D_1 \simeq D_2^{\text{op}}$.

Proof. — All assertions except for the one about the diagram and the rational roots are combinations of 4.5.17.ii and 4.5.19. To compute the diagram, note the following: Inside the derived algebra of $M_{r+1}(D)$, the diagonal matrices with entries in k are a split toral subalgebra \mathfrak{s} of dimension r. On the other hand, the diagonal matrices with entries in in $D \setminus k^*$ form an anisotropic subalgebra isomorphic to a product of r copies of \mathcal{DD} . Such a \mathcal{DD} , in turn, by 4.4.3 and 4.5.18 has Satake-Tits diagram

$$d-1$$

Finally one sees that $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ consists exactly of all diagonal matrices in \mathfrak{g} , hence is the k-direct sum of \mathfrak{s} and the named anisotropic subalgebra, from which it follows that \mathfrak{s} is indeed maximal split toral, the k-rank of our algebra is r, the named anisotropic subalgebra is its anisotropic kernel (with respect to \mathfrak{s}), and the Satake-Tits diagram is as described. (For this, one could also invoke 4.3.7 as soon as one knows that the k-rank is r.) The rational roots are computed as in 3.4.1.

4.5.2.2. The outer forms/Type AII. — Here again we content ourselves with a description of the admissible diagrams, since for \mathfrak{p} -adic k, we will get a uniqueness/isomorphism statement for free.

Let (A, ι) be of type AII, specifically, $A \simeq M_{m+1}(D)$ with D a central division algebra over C of degree d, C|k a quadratic extension, $m+1 = \frac{n+1}{d}$, $n \ge 2$, and there

is an involution σ of the second kind on D, so that $\sigma_{|C}$ is the non-trivial element of Gal(C|k). By the second part of 4.5.3, the Lie algebras $\mathfrak{S}(A,\iota)$ are classified (not necessarily one-to-one) by the σ -hermitian forms on $D^{n+1/d}$.

PROPOSITION 4.5.22. — Let such a form have Witt index r. Then $\mathcal{D}\mathfrak{S}(A,\iota)$ has Satake-Tits diagram



with C being the fixed field of the twisted Galois action.

Proof. — (We restrict to the case d = 1 i.e. D = C. This is obviously the only case if $k = \mathbb{R}$, but even so for k a \mathfrak{p} -adic field, as will turn out in 5.4.3. Cf. [**TiW**, 42.3.4] for a geometric interpretation of the group analogue. See also [**Bo2**, VIII.13, Exercise 16.b].) Fix an element y with C = k(y), $y^2 \in k$. In generalisation of 3.2.10 and similar to the types B - D, our Lie algebra $\mathfrak{g} = \mathcal{D}\mathfrak{S}(A, \iota)$ is given by those traceless (!) matrices $X \in A = M_{n+1}(C)$ such that $\iota(X) = {}^tH^{-1} \cdot {}^t(\sigma(X)) \cdot {}^tH = -X$ where H is a σ -hermitian matrix. The coordinates can be chosen such that

$$H := \begin{pmatrix} 0 & I_r & 0 \\ I_r & 0 & 0 \\ 0 & 0 & H_a \end{pmatrix}.$$

where H_a defines an anisotropic form on C^{n-2r} .

NB. In 3.2.10, we had maximal possible Witt index $r = \lfloor \frac{n+2}{2} \rfloor$ and chose the (equivalent but) different coordinatisation

$$H = \begin{pmatrix} 0 & 1 \\ & 1 \\ & \ddots & \\ 1 & & 0 \end{pmatrix}.$$

Now \mathfrak{g} (with our new general coordinates) consists of the matrices

$$\begin{pmatrix} A & B & G \\ D & -^t(\sigma(A)) & K \\ E & F & J \end{pmatrix}$$

with trace 0 and where $A, B, D \in M_r(C), E, F \in M_{n-2r \times r}(C), G, H \in M_{r \times n-2r}(C)$ are subject to the following relations: ${}^tB = -\sigma(B), {}^tD = -\sigma(D)$ (i.e. B and D are σ -skew-hermitian), ${}^t\sigma(F) \cdot {}^tH_a = -G, {}^t\sigma(E) \cdot {}^tH_a = -K$, and ${}^tH_a^{-1} \cdot {}^t\sigma(J) \cdot {}^tH_a = -J$. If we assume, as we always can, H_a to be the diagonal matrix diag (b_1, \ldots, b_{n-2r}) –

MÉMOIRES DE LA SMF 151

94

where necessarily the $b_i \in k$ –, the last relation means that the (i, j)-entry of J is the σ -conjugate of its (j, i)-entry times $-b_i^{-1}b_j$; in particular, the diagonal of J consists of k-multiples of y. Again one checks that a split toral subalgebra of k-dimension r is given by

$$\mathfrak{s} := \{ \operatorname{diag}(a_1, \dots, a_r, -a_1, \dots, -a_r, 0, \dots, 0) : a_i \in k \}$$

and its centraliser $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ consists of the traceless matrices



with $c_i \in C$ and J as above. No element of $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}) \setminus \mathfrak{s}$ is ad-diagonalisable over k, hence \mathfrak{s} is maximal split toral. One further checks that a maximal toral \mathfrak{t} containing \mathfrak{s} is given by those elements in $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ where J is diagonal (note that indeed $\dim_k(\mathfrak{t}) = n$). With an argument similar to 4.5.13, as soon as $2r \leq n-1$, the traceless matrices J describe an anisotropic subalgebra which identifies with $\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s})$ modulo its centre. This is the anisotropic kernel \mathfrak{g}_a , ⁽⁹⁾ and the C-rank of $(\mathfrak{g}_a)_C$ is n-2r.

To "see" the swapping of the roots, one can introduce the toral subalgebra

$$\mathfrak{u} := \big\{ \operatorname{diag}(b_1 y, \ldots, b_r y, b_1 y, \ldots, b_r y, 0, \ldots, 0) : b_i \in k, \sum_{i=1}^r b_i = 0 \big\}.$$

Then $\mathfrak{s} \oplus \mathfrak{u} \oplus \mathfrak{t}_a = \mathfrak{t}$ as long as $2r \leq n$. Call α_i the linear form sending a general element of \mathfrak{s} as above to $a_i - a_{i+1}$, and β_i the one sending a general element of \mathfrak{u} as above to $b_i - b_{i+1}$. One can check that $(\alpha_i \pm y\beta_i)_{1 \leq i \leq r}$ are part of a root basis of $R(\mathfrak{g}_C, \mathfrak{t}_C)$, and σ switches the sign: they are indeed the white vertices in the diagram. If 2r = n + 1, so the form is quasi-split, one has to set $\beta_r = 0$, mirroring the trace 0 condition and producing the lone middle vertex α_r .

4.6. Quasi-split forms

Absolutely simple, non-split quasi-split Lie algebras can a priori exist in the following types:

^{9.} After minor sign changes and the trace 0 condition, this is the decisive difference to the discussion for type D_{ν} . It comes from the commutativity of C, opposed to the divison algebra D there having k as centre.



 D_4 (trialitarian types)

As noted in Section 3.4, the isomorphism class of a semisimple Lie algebra over k having such a diagram is uniquely determined by it together with the fixed field of t, which is a quadratic (resp. cubic or degree 6 for the trialitarian types) splitting field for the Lie algebra, in fact its unique minimal splitting field (inside a fixed \bar{k}).

Those of type A_n were described in 3.2.10 and, with other coordinates, in the previous section. We saw that the isomorphism classes of absolutely simple, non-split, quasi-split k-Lie algebras of type A_n , $n \ge 2$, correspond bijectively to the quadratic extensions K|k (inside our fixed \bar{k}).

Those of non-trialitarian type D_n have been dealt with in the general discussion in Section 4.5.1. Their isomorphism classes are in correspondence with anisotropic quadratic forms in two variables up to scaling and equivalence, which (by $b_1X_1^2 + b_2X_2^2 \rightarrow -b_1b_2$) is again the set k^*/k^{*2} or that of quadratic extensions of k. The rational root system of such a form is easily computed to be of type B_{n-1} .

A quasi-split form of type E_6 for a given quadratic extension K|k can be patched from the corresponding quasi-split form of type A_5 and a copy of $\mathfrak{sl}_2(k)$ with the patching Proposition 4.3.3:



We see that again these forms are in correspondence with quadratic extensions of k, because those of type A_5 are. The rational root system of such a form is of type F_4 as the method of Section 3.4.1 shows.

For the trialitarian types, it should be clear what is meant in the diagram: $t(\sigma)$ is a generator for $\mathbb{Z}/3$, and the $t(\sigma_i)$ are the three transpositions in S_3 , for certain elements σ resp. $\sigma_i \in \text{Gal}(\bar{k}|k)$ that restrict to the fixed field of t, which is a Galois cubic extension K|k resp. an extension K|k with Galois group isomorphic to S_3 . The rational root system was computed to be of type G_2 in 3.4.1. We show that similar to the quadratic cases above, every Galois extension K|k with Galois group of order 3 or isomorphic to S_3 , respectively, can occur as fixed field of the t-action.

What one patches here is a copy of $\mathfrak{sl}_2(k)$ – giving the central vertex – with a suitable scalar restriction $R_{K_0|k} \mathfrak{sl}_2(K_0)$. Namely, $K_0 = K$ for the $\mathbb{Z}/3$ case, whereas for the S_3 case we take as K_0 any of the three (non-Galois) subextensions of K|k of degree 3. By Section 4.1 the diagrams are, respectively,



and K is the fixed field of the twisted Galois action.

When one knows the split forms, the classification of quasi-split forms is thus entirely equivalent to the arithmetic-algebraic problem to determine all Galois extensions of k with Galois group isomorphic to $\mathbb{Z}/2, \mathbb{Z}/3$ and S_3 , the last being in turn equivalent to that of non-Galois cubic extensions. This is also equivalent to the determination of $H^1(\text{Gal}(\bar{k}|k), A(R)/W(R))$ (= the homomorphisms of $\text{Gal}(\bar{k}|k)$ to A(R)/W(R) up to conjugacy), which fits in the Galois cohomological approach.

CHAPTER 5

CLASSIFICATION OVER SPECIAL FIELDS

Again, k is a field of characteristic zero, and every extension of it is understood to be in a fixed algebraic closure $\bar{k}|k$.

5.1. C1 fields and unramified splitting for p-adic fields

LEMMA 5.1.1 (Springer, [Spr1]). — If k is C1, every non-zero semisimple Lie algebra contains a non-zero nilpotent element.

Proof. — Let \mathfrak{g} be a semisimple Lie algebra of dimension $n \geq 1$. For $x \in \mathfrak{g}$, the characteristic polynomial of $\mathrm{ad}_{\mathfrak{g}}(x) \in \mathrm{End}_k(\mathfrak{g})$ with variable T is of the form

$$\sum_{i=r}^{n} p_i(x) T^i$$

where $r = \operatorname{rk}(\mathfrak{g})$ and for i < n, p_i is a homogeneous polynomial map from \mathfrak{g} to k of degree n - i, meaning that after a choice of basis x_1, \ldots, x_n of \mathfrak{g} , there are homogenous polynomials $P_i \in k[X_1, \ldots, X_n]$ of degree n - i such that $p_i(\sum_{i=1}^n c_i x_i) = P_i(c_1, \ldots, c_n)$. By the C1 property, in particular there is a non-zero $y \in \mathfrak{g}$ with $p_r(y) = 0$ or in other words, y is non-regular.

We prove the assertion by induction over $n = \dim_k \mathfrak{g}$. For $n = 0 \Leftrightarrow \mathfrak{g} = 0$, there is nothing to show. There are no semisimple Lie algebras of dimension 1 or 2, and if n = 3, then $\mathfrak{g}_{\bar{k}} \simeq \mathfrak{sl}_2(\bar{k})$. We identify \mathfrak{g} with a subalgebra of $R_{\bar{k}|k}\mathfrak{g}_{\bar{k}}$ and use that the characteristic polynomial "remains the same" (cf. [**Bo2**, VII.2.2, Rem. 2]) so that by *ibid.*, example 2, $p_2 \equiv 0$ and hence the *y* constructed above is nilpotent.

Now let n > 3 and assume the assertion is true for all Lie algebras of dimension < n. Pick a non-regular $y \neq 0$ as above; by Jordan decomposition, w.l.o.g. y is semisimple. Consider $\mathfrak{g}_1 := \mathfrak{z}_{\mathfrak{g}}(y) = \ker(\mathrm{ad}_g(y)) = \mathfrak{g}^0(y)$. By [**Bo2**, VII.1, Prop. 11], \mathfrak{g}_1 is reductive, by [**Bo2**, VII.3.3, Cor.] it contains an element z which is regular in \mathfrak{g} . If we had [y, z] = 0, then by the Jacobi identity $\ker(\mathrm{ad}_{\mathfrak{g}}(y)) = \ker(\mathrm{ad}_{\mathfrak{g}}(z))$, but this is a Cartan subalgebra whose dimension is $\operatorname{rk}(\mathfrak{g}) < \dim_k(\mathfrak{g}_1)$. Thus the derived algebra $\mathcal{D}\mathfrak{g}_1$ is nonzero, semisimple, and its dimension is < n. By the induction hypothesis it contains an $x \neq 0$ which is nilpotent in $\mathcal{D}\mathfrak{g}_1$ and hence (2.1.8) in \mathfrak{g} .

This has a strong consequence:

PROPOSITION 5.1.2. — If k is C1, every semisimple Lie algebra is quasi-split.

Namely, this follows from Springer's lemma and the following fact.

PROPOSITION 5.1.3. — The following are equivalent (for the field k):

- i. Every non-zero semisimple Lie algebra contains a non-zero nilpotent element.
- ii. Every semisimple Lie algebra contains a Borel subalgebra (i.e. is quasi-split).
- iii. The only anisotropic semisimple Lie algebra is $\{0\}$.

Proof. — This can be shown directly (cf. [**Bo2**, VIII.10, Exercise 7.a]) but follows easily from our previous results. Indeed, $i \Leftrightarrow iii$ is immediate from 3.2.1.viii. By part ix there, an anisotropic Lie algebra with a Borel subalgebra must be equal to it, hence solvable, hence zero, so ii \Rightarrow iii. Finally, if iii is valid, then in particular every semisimple \mathfrak{g} has trivial anisotropic kernel, so is quasi-split by 3.2.6.iv.

REMARK 5.1.4. — Corresponding statements for algebraic groups are largely due to Serre [Ser1], Springer, and Steinberg [Ste]. Indeed the following are equivalent for our field k (remember char(k) = 0 and every extension is understood to be an intermediate extension of $\bar{k}|k$):

- i. The cohomological dimension of $Gal(\bar{k}|k)$ is 1.
- ii. For every extension K|k, the Brauer group Br(K) is trivial.
- iii. For every extension K|k and every finite Galois extension L|K, the Gal(L|K)-module L^* is cohomologically trivial.
- iv. In the situation of iii, the norm $N_{L|K}: L^* \to K^*$ is surjective.
- v. $H^1(k,G)$ is trivial for every semisimple (or connected linear) algebraic group G.
- vi. Every linear algebraic group contains a Borel subgroup defined over k.
- vii. Every non-trivial semisimple algebraic group contains a unipotent element $\neq 1$.
- viii. k has the properties of 5.1.3.

This is the property $\dim(k) \leq 1$ in [Ser3], and the above equivalences are Proposition 5 in II, 3.1 and Theorems 1, 1', 2 in III, 2 there. It is shown in *ibid*. II, 3.2 that the property C1 implies iv and thus all of the above. Springer's lemma above showed that C1 implies viii, and "viii \Rightarrow ii" was 4.4.4.

Proposition 5.1.2 combined with Lang's theorem 2.2.2 implies:

COROLLARY 5.1.5 (Weisfeiler's Theorem 1). — For a semisimple Lie algebra \mathfrak{g} over a \mathfrak{p} -adic field k, there exists a finite unramified extension $\mathfrak{K}|k$ such that $\mathfrak{g}_{\mathfrak{K}}$ is quasi-split.

100
Proof. — Since the maximal unramified extension k^{nr} is C1, 5.1.2 says that $\mathfrak{g}_{k^{nr}}$ contains a Borel subalgebra \mathfrak{B} . Let b_1, \ldots, b_n be a k^{nr} -basis of \mathfrak{B} . Since we can write

$$\mathfrak{g}_{k^{nr}} = igcup_{k\subseteq\mathfrak{K}\subseteq k^{nr}} \mathfrak{g}_{\mathfrak{K}}$$
 $\mathfrak{g}_{k ext{ finite}}$

there is a finite unramified extension $\mathfrak{K}|k$ such that $\mathfrak{g}_{\mathfrak{K}}$ contains all b_i . The vector space \mathfrak{b} spanned by them in $\mathfrak{g}_{\mathfrak{K}}$ satisfies $\mathfrak{b} = \mathfrak{B} \cap \mathfrak{g}_{\mathfrak{K}}$ and hence is a subalgebra, and by definition a Borel subalgebra, of $\mathfrak{g}_{\mathfrak{K}}$.

5.2. Type AI and Kneser's theorem

We had seen in the general discussion how the inner forms of type A are completely described by the central division algebras over k. Now for a p-adic field k, one has the well-known (cf. [**Rei**, Sections 14 and 31])

THEOREM 5.2.1 (cf. Hasse [Has]). — Br(k) $\simeq \mathbb{Q}/\mathbb{Z}$. For $d \ge 1$, there are exactly $\phi(d)$ isomorphism classes of k-central division algebras of degree d. Each of them contains a copy of, and is split by, any field extension K|k of degree d, and there are $\lfloor \frac{\phi(d)+1}{2} \rfloor$ classes up to isomorphism and anti-isomorphism.

Combined with 4.5.21 this gives a complete classification of inner forms of type A. In particular:

PROPOSITION 5.2.2. — For $d \ge 2$, there are $\lfloor \frac{\phi(d)+1}{2} \rfloor$ anisotropic inner forms of type A_{d-1} over k up to isomorphism. Each of them is split by any extension of degree d.⁽¹⁾

So for $d \in \{2, 3, 4, 6\}$ we can and will speak of "the" anisotropic (inner) form of \mathfrak{sl}_d . We will see that \mathfrak{sl}_d has no outer anisotropic forms. In fact the following theorem, due independently to Weisfeiler and Kneser, states that much more is true.

THEOREM 5.2.3 (Weisfeiler's Theorem 2). — Over a \mathfrak{p} -adic field k, all absolutely simple anisotropic Lie algebras are inner forms of type A.

These are the ones from above, classified by the k-central division algebras. The remarkable assertion is that no other anisotropic forms exist.

Kneser actually noted the corresponding statement for algebraic groups: see [Kne, Satz 3], $^{(2)}$ where it follows from his main

THEOREM 5.2.4 (Kneser). — For a p-adic field k and a connected, simply connected semisimple algebraic group G defined over k, we have $H^1(\text{Gal}(\bar{k}|k), G) = 1$.

^{1.} This means that for any such extension K|k, there is a maximal toral subalgebra of \mathfrak{g} which is split by K. Indeed, the maximal abelian subalgebras of \mathfrak{D} "are" the maximal (= d-dimensional) subfields of D, and their intersections with $\mathcal{D}\mathfrak{D}$ are the maximal toral subalgebras of this.

^{2.} Bruhat and Tits complemented their opus magnum with a completely different and more general proof in [**BT2**, Section 4].

In fact, the whole classification of semisimple algebraic groups, or Lie algebras, over a \mathfrak{p} -adic field k can be derived from Kneser's theorem. This is sketched in [Kne, II.4–5] and also done in [Sat3, II.3.3]. The idea is that the vanishing of Kneser's cohomology group gives upper bounds on the size of $H^1(\text{Gal}(\bar{k}|k), \text{Aut}(X_{\bar{k}}))$, where X is the object of interest. We sketch the procedure, for an absolutely simple Lie algebra \mathfrak{g} of type R. There are algebraic groups G and G_0 defined over k such that G(K) = $\text{Aut}(\mathfrak{g}_K)$ and $G_0(K) = \text{Aut}_0(\mathfrak{g}_K)$ for every subextension K|k of \bar{k} . The group G_0 is connected and of adjoint type (cf. [Bo2, VIII.5.4]), and there is an exact sequence

$$1 \to Z \longrightarrow G_0 \longrightarrow G_0 \to 1$$

where \tilde{G}_0 satisfies the conditions for Kneser's theorem and (the dual of) its centre Z can be identified with P(R)/Q(R), which is listed for all types in the plates of [**Bo2**, VI]. With [**Ser3**, I.5.7] we get an exact sequence

(19)
$$1 \to H^1(k, G_0) \to H^2(k, Z)$$

We first look at the case $G = G_0$ where we only have *inner forms*: e.g. for the types E_8 , F_4 and G_2 , Z and hence $H^2(k, Z)$ are trivial, so there is only the split form. For type E_7 , again $G_0 = G$, but now Z has two elements and so has $H^2(k, Z)$ (as can be seen via Tate duality). We will indeed show the existence of one non-split form of type E_7 over k in 5.5.5. Hence, this and the split one are all of them. In the general case we have $G \simeq G_0 \rtimes S$ (semidirect product) where $S \simeq A(R)/W(R)$ is the automorphism group of the Dynkin diagram (cf. [**Bo2**, VIII.5.3, Cor. 1]). This gives a surjective ϕ in the exact sequence

$$H^1(k,G_0) \longrightarrow H^1(k,G) \xrightarrow{\phi} H^1(k,S)$$

of [Ser3, I.5.5]. $H^1(k, S)$ classifies exactly the quasi-split forms, which is the Galois cohomological interpretation of our results in 4.6. Indeed this sequence can be translated to the statement that every form can be viewed as an "inner form of a uniquely determined quasi-split form". (The "inner forms" in our terminology are then the inner forms of the split form, i.e. the image of the left = kernel of the right map in the sequence). With [Ser3, I.5.5, Cor. 2] for an $\alpha \in H^1(k, S)$ (corresponding to one quasi-split form), the set $\phi^{-1}(\alpha)$ (corresponding to the "forms of this form") can be identified with the quotient of $H^1(k, {}_aG_0)$ under a certain action of $H^0(k, {}_aS)$, where a is an arbitrary cocycle representing an arbitrary \bar{a} with $\phi(\bar{a}) = \alpha$, and the subscripts denote twisting with this cocycle. In particular, the inner forms (in our terminology) are given by orbits of S in $H^1(k, G_0)$. With (19) one can transport this to the orbits of an S-action on $H^2(k, Z)$, which further identifies with (the Galois fixed set of the dual of) Z by Tate duality. In type E_6 , Z is a group of three and S a group of two elements, the S-action permutes the two non-trivial elements of Z, so we expect (at most) one non-split inner form and will indeed see it in 5.5.4. For the outer forms of type E_6 , and also e.g. for the trialitarian types D_4 , it turns out that already the

MÉMOIRES DE LA SMF 151

twisted $H^2(k, {}_aZ)$ is trivial, hence in these cases there exist no outer forms except for the quasi-split ones we have described in 4.6. Tables which summarise the results for all types are found in [**Kne**, p. 254] and [**Sat3**, p. 121].

We believe, however, that this approach is not the only possible way. First, using results about (skew-)hermitian forms over \mathfrak{p} -adic fields, we can completely classify the classical types with our earlier results, and theorem 5.2.3 for these types follows from this. This will be done in Section 5.4. (In [**Kne**, I], Kneser proves his theorem for the classical types using the same results about hermitian forms.) In the further course of the classification in Section 5.5, we will see that each possible diagram can either be excluded by our previous results or shown to be admissible – *except for the anisotropic forms*. Thus the validity of Weisfeiler's table is reduced, without using Kneser's theorem, to the seemingly simpler statement:

REDUCED THEOREM. — There are no anisotropic Lie algebras of exceptional type over p-adic fields.

To prove this, we need some more machinery which we will set up in the next Section 5.3. We will then combine that with our previous results and Weisfeiler's first theorem (5.1.5) to prove the Reduced Theorem in every case except for one form in type E_6 and one in type D_4 .

Kneser himself suggests equivalence of his and Weisfeiler's results in [Kne, II, footnote 4]. The paper [Tit4] deals with anisotropic forms, but for p-adic fields it also refers to Kneser's theorem.

REMARK 5.2.5. — For $k = \mathbb{R}$ the situation is very different. Here it is a classical result due to E. Cartan that for every type, there exists a unique anisotropic form. But only one of them is an inner form of type A, in fact A_1 , the derived Lie algebra of Hamilton's quaternions; whereas e.g. the anisotropic forms of type A_n , $n \ge 2$, are outer forms.

5.3. Tools for ruling out anisotropic forms: The prime degree case

Recall Weisfeiler's first theorem 5.1.5 which said that we have unramified, in particular cyclic splitting extensions, at least for the inner forms of split Lie algebras (cf. definition 4.2.4). A cyclic extension K|k can further be "filtered" to a tower of fields $k \,\subset K_0 \,\subset K_1 \,\subset \,\cdots \,\subset K_d = K$ such that every step $K_r|K_{r-1}$ is of prime degree. These are the ones we are going to attack in this section. Already basic arithmetic constraints rule out most prime degree extensions as splitting fields, as we will see, not only in the p-adic case. The main goal of this section then is a big step towards the Reduced Theorem in the preceding section: we show that over p-adic fields, there are no anisotropic Lie algebras of exceptional type which are split by an unramified extension of prime degree.

We start with a technical lemma. Let R be a root system and V be the \mathbb{R} -vector space spanned by it.

LEMMA 5.3.1. — Let ℓ be a prime, and $\sigma \in A(R)$ such that $\sigma \in GL(V)$ has minimal polynomial $T^{\ell-1} + T^{\ell-2} + \cdots + T + 1$ (in particular $\sigma^{\ell} = id$). Then for every root $\alpha \in R$, after possibly replacing σ by some σ^j (with $1 \leq j \leq \ell - 1$), the roots

$$\alpha_1 := \alpha, \alpha_2 := \sigma(\alpha), \dots, \alpha_{\ell-1} := \sigma^{\ell-2}(\alpha)$$

satisfy the relations of a basis of root system of type $A_{\ell-1}$ (i.e. correspond to the nodes in the associated Dynkin diagram).

Proof. — The result is trivially true for $\ell = 2$, so let $\ell \geq 3$. We choose an A(R)-invariant scalar product $\langle ., . \rangle$ on V. All $\sigma^{j}(\alpha)$ have the same length, without loss of generality let this length be 1. $\alpha, \sigma(\alpha), \ldots, \sigma^{\ell-2}(\alpha)$ are linearly independent, and we have

$$\sigma^{\ell-1}(\alpha) = -\sum_{i=0}^{\ell-2} \sigma^i(\alpha)$$

Multiplying out $\langle \sigma^{\ell-1}(\alpha), \sigma^{\ell-1}(\alpha) \rangle = 1$, using the σ -invariance and $\langle \alpha, \alpha \rangle = 1$, gives

(20)
$$(\ell - 1) + 2\sum_{i=1}^{\ell-2} (\ell - 1 - i) \langle \alpha, \sigma^i(\alpha) \rangle = 1$$

Since $\ell \geq 3$, this means that for some $1 \leq j \leq \ell - 2$, $\langle \alpha, \sigma^j(\alpha) \rangle$ must be negative. Replace σ by this σ^j , so we have $\langle \alpha, \sigma(\alpha) \rangle < 0$ and indeed by [**Bo2**, VI.1.3], $\langle \alpha, \sigma(\alpha) \rangle = -\frac{1}{2}$ because the roots have the same length 1. But then by σ -invariance, $\langle \sigma^{\ell-1}(\alpha), \alpha \rangle = -\frac{1}{2}$ too and thus

(21)
$$\sigma^{\ell-1}(\alpha) + \alpha = -\sum_{i=1}^{\ell-2} \sigma^i(\alpha)$$

is a root which further is of length 1 again ([loc. cit.]). We are finished here if $\ell = 3$. To proceed in the case $\ell > 3$, note first that inserting $\langle \alpha, \sigma(\alpha) \rangle = -\frac{1}{2}$ into (20) gives

$$2\sum_{i=2}^{\ell-2} (\ell-1-i) \langle \alpha, \sigma^i(\alpha) \rangle = 0.$$

Further note that

(22)
$$\langle \alpha, \sigma^i(\alpha) \rangle = \langle \alpha, \sigma^{\ell-i}(\alpha) \rangle$$

for all $0 \le i \le \ell$ by σ -invariance and $\sigma^{\ell} = id$. Coupling these pairs reduces our equation further to

(23)
$$(\ell-2) \cdot \sum_{i=2}^{\ell-2} \langle \alpha, \sigma^i(\alpha) \rangle = 0$$

MÉMOIRES DE LA SMF 151

where we can cancel the constant too. Applying σ^{-1} to (21), we have the root

(24)
$$\sum_{i=0}^{\ell-3} \sigma^i(\alpha)$$

of length 1, which multiplied out gives

$$(\ell-2) + 2\sum_{i=1}^{\ell-3} (\ell-2-i) \langle \alpha, \sigma^i(\alpha) \rangle = 1$$

or, using $\langle \alpha, \sigma(\alpha) \rangle = -\frac{1}{2}$ and coupling the pairs (22) again,

$$(\ell - 3) \cdot \sum_{i=2}^{\ell-3} \langle \alpha, \sigma^i(\alpha) \rangle = 0$$

where again cancelling the constant (we assume $\ell > 3$ now) and comparing with (23) gives:

(25)
$$\langle \alpha, \sigma^{\ell-2}(\alpha) \rangle = \langle \alpha, \sigma^2(\alpha) \rangle = 0.$$

From this one infers that

(26)
$$\sigma^{\ell-1}(\alpha) + \alpha + \sigma(\alpha) = -\sum_{i=2}^{\ell-2} \sigma^i(\alpha)$$

is again a root of length 1. This process can be repeated, showing iteratively that

$$\langle \alpha, \sigma^i(\alpha) \rangle = 0$$

for $i \in \{2, \ldots, \ell - 2\}$. This proves the claim.

Actually, we do not even need ℓ to be a prime – we note the following generalisation for use in later sections:

COROLLARY 5.3.2. — Let $n \in \mathbb{N}$ and $\sigma \in A(R)$ such that $\sigma \in \operatorname{GL}(V)$ has minimal polynomial $T^{n-1} + T^{n-2} + \cdots + T + 1$ (in particular $\sigma^n = \operatorname{id}$). Then for every root $\alpha \in R$ such that the σ -conjugates of α span an (n-1)-dimensional subspace of V, after possibly replacing σ by some σ^j (with $\operatorname{gcd}(j, n) = 1$), the roots

$$\alpha_1 := \alpha, \alpha_2 := \sigma(\alpha), \dots, \alpha_{n-1} := \sigma^{n-2}(\alpha)$$

satisfy the relations of a basis of root system of type A_{n-1} (i.e. correspond to the nodes in the associated Dynkin diagram).

Proof. — By induction on the number of divisors of n, the lemma being the start of the induction. Then for general n, one imitates the proof until the place after equation (20) where we get

$$\langle \alpha, \sigma^j(\alpha) \rangle < 0$$

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

for some $1 \leq j \leq n-2$. If gcd(j, n) were not 1, we could w.l.o.g. assume that j is a divisor of n and use the induction hypothesis on σ^j to see that the sets

$$\{\alpha, \sigma^{j}(\alpha), \sigma^{2j}(\alpha), \dots, \sigma^{(d-1)j}(\alpha)\},$$

$$\{\sigma(\alpha), \sigma^{j+1}(\alpha), \sigma^{2j+1}(\alpha), \dots, \sigma^{(d-1)j+1}(\alpha)\},$$

$$\vdots$$

$$\{\sigma^{j-1}(\alpha), \sigma^{2j-1}(\alpha), \sigma^{3j-1}(\alpha), \dots, \sigma^{(n-1)}(\alpha)\}$$

(where d := n/j) would each span root systems of type A_{d-1} ; and there are j of them. But that would mean the span of all σ -conjugates of α would only have dimension $j \cdot (d-1) = n - j$, contradiction.

So j does not divide n, and the rest of the proof is exactly the same as in the lemma.

REMARK 5.3.3. — i. This is consistent with table 3 in [Car]. Also, a posteriori, σ is a Coxeter element of the sub-root-system of type A_{n-1} .

- ii. Note that in the situation of the lemma, σ splits V into σ -invariant subspaces each of dimension $\ell 1$. In particular, $\operatorname{rk}(R)$ has to be a multiple of $\ell 1$.
- iii. Note however that although the σ -invariant subspace spanned by the σ conjugates of any root α now necessarily contains a sub-root system of type $A_{\ell-1}$,
 the intersection of R with that subspace might be a larger root system. An example is a root system of type G_2 (cf. [**Bo2**, Plate X], or this work's dedication)
 with σ being the rotation by 120°.

The lemma puts severe restrictions on anisotropic forms of exceptional types with splitting fields of prime degree:

PROPOSITION 5.3.4. — Let k be a field with $\operatorname{char}(k) = 0$, and let \mathfrak{g} be an absolutely simple, anisotropic Lie algebra of exceptional type over k, with a cyclic splitting extension K|k of prime degree ℓ . Let $(X, R, \Delta, \Delta_0 = \Delta, t)$ be the index of \mathfrak{g} with respect to $\mathfrak{s} = \{0\}$ and a maximal toral \mathfrak{t} which is split by the extension K. Let σ be a generator of $\Gamma = \operatorname{Gal}(K|k)$ and resume notations like σ_A, X_0, V , etc. from Chapter 3. Then:

- \triangleright If $\ell = 2$, $\sigma_A = -id$.
- \triangleright If $\ell = 3$, R cannot be of type E_7 .
- \triangleright If $\ell = 5$, R must be of type E_8 .
- \triangleright The case $\ell > 5$ cannot occur.

Proof. — Since $X = X_0 = \{\chi \in X : \sum_{i=0}^{\ell-1} \sigma_A{}^i(\chi) = 0\}$, the element $\sigma_A \in A(R)$ has no eigenvector of eigenvalue 1 in $V = \mathbb{R}X$ (this is called an "elliptic element"), and it is of order ℓ . This already implies the assertion for $\ell = 2$, and more generally, the minimal polynomial of $\sigma_A \in \mathrm{GL}(V)$ is the cyclotomic polynomial $T^{\ell-1} + T^{\ell-2} + \cdots + T + 1$, so

we are in the situation of the preceding lemma and remark. For odd ℓ and R not of type D_4 , we also see that $\sigma_A \in W(R)$ because $|A(R)/W(R)| \leq 2$; for R of type D_4 , the same holds true for $\ell > 3$. Then D_4 , F_4 and G_2 are excluded for $\ell > 3$ because their Weyl groups have orders $2^6 \cdot 3$, $2^7 \cdot 3^2$ and $2^2 \cdot 3$. E_6 is excluded for $\ell > 5$ since its Weyl group is of order $2^7 \cdot 3^4 \cdot 5$.

Finally, part ii of the remark implies $\operatorname{rk}(R)/(\ell-1) \in \mathbb{Z}$, which excludes type E_6 for $\ell = 5$ and type E_7 for $\ell \geq 3$.

Now we attack these remaining cases over \mathfrak{p} -adic fields, starting with $\ell = 2$. The following has an analogue in [We3], especially §§8 and 12, but we were not able to verify all statements there. We take a different approach which however uses the same key fact about the norms of the unramified quadratic extension.

PROPOSITION 5.3.5 (Case $\ell = 2$). — Let \mathfrak{g} be a semisimple Lie algebra over a \mathfrak{p} -adic field k which is split by the unramified quadratic extension K|k. If the root system of \mathfrak{g}_K contains roots γ, δ such that $\gamma + \delta$ is a root, then \mathfrak{g} cannot be anisotropic.

In particular, if \mathfrak{g} is absolutely simple and anisotropic, it must be of type A_1 , i.e. the anisotropic form of \mathfrak{sl}_2 .

Proof. — Let σ be the non-trivial element of $\operatorname{Gal}(K|k)$. Assume \mathfrak{g} is anisotropic. In our general terminology of choices, necessarily $\mathfrak{s} = \{0\}$. Choose a maximal toral \mathfrak{t} defined over k and split over K.

We resume notation from Section 3.3.1 but now fix a *Chevalley basis* consisting of H_{α} and bases e_{α} of the root spaces, which means that the structure constants $N_{\alpha,\beta}$ defined in (14) are in $\mathbb{Z} \setminus \{0\}$ (such a basis exists by [**Che**], cf. [**Bo2**, VIII.4.4]). We have $N_{\alpha,\beta} = -N_{\beta,\alpha}$ for all roots α, β as well as $H_{-\alpha} = -H_{\alpha}$. Also, we have the crucial relation

(27)
$$N_{-\alpha,-\beta} = -N_{\alpha,\beta}$$

cf. [Che, p. 23]. (Our sign conventions follow [Che] instead of Bourbaki, that is, we have $[e_{\alpha}, e_{-\alpha}] = H_{\alpha}$. Remember that we chose our basis of \mathfrak{sl}_2 accordingly in 2.1.6.)

Now the root lattice satisfies $X = X_0$ and therefore $\sigma_A = -id$ on X; in particular, $\sigma_A(\alpha) = -\alpha$ for all roots α . We get elements $\xi_\alpha \in K^*$ defined by

$$\sigma(e_{\alpha}) = \xi_{\alpha} \, e_{-\alpha}$$

Because $\sigma(H_{\alpha}) = [\sigma(e_{\alpha}), \sigma(e_{-\alpha})] = \xi_{\alpha}\xi_{-\alpha}H_{\sigma_{A}(\alpha)}$ and $\sigma_{\alpha}(\sigma(H_{\alpha})) = \sigma(\alpha(H_{\alpha})) = 2$, we have $\sigma(H_{\alpha}) = H_{\sigma_{A}(\alpha)} = H_{-\alpha} = -H_{\alpha}$ and $\xi_{-\alpha} = \xi_{\alpha}^{-1}$ for all $\alpha \in R$. We also have $\sigma(\xi_{\alpha}) \cdot \xi_{-\alpha} = 1$ as in (12), hence $\xi_{\alpha} \in k^{*}$, and for each root α , the copy of $\mathfrak{sl}_{2}(K)$

$$\mathfrak{G}(\mathfrak{t},\alpha) := KH_{\alpha} + (\mathfrak{g}_K)_{\alpha} + (\mathfrak{g}_K)_{-\alpha}$$
$$= KH_{\alpha} + Ke_{\alpha} + Ke_{-\alpha}$$

is σ -stable, hence descends to k where it becomes $\mathfrak{g}(\mathfrak{t}, \alpha) := \mathfrak{G}(\mathfrak{t}, \alpha)^{\sigma=\mathrm{id}}$ and necessarily is the anisotropic form of \mathfrak{sl}_2 . Picking $y \in K^*$ with $\sigma(y) = -y$, it has the k-basis

 $yH_{\alpha}, \quad e_{\alpha} + \xi_{\alpha}e_{-\alpha}, \quad y(e_{\alpha} - \xi_{\alpha}e_{-\alpha})$

and visibly is isomorphic to the form described in example 3.2.5, the ξ_{α} here identifying with the "b" there (with our sign conventions; with Bourbaki's conventions instead, it would be -b). We had seen in that example that this \mathfrak{sl}_2 -form being anisotropic is equivalent to $\xi_{\alpha} \notin N_{K|k}(K)$ which, K|k being unramified, means that

(28) for all roots
$$\alpha$$
, we have: $\xi_{\alpha} \equiv \pi_k \mod \mathcal{O}_k^*$

In other words, for all roots α , the constant ξ_{α} must be of odd valuation in k. But if there are roots γ, δ whose sum is a root, calculating $\sigma([e_{\gamma}, e_{\delta}]) = \sigma(N_{\gamma,\delta} e_{\gamma+\delta})$ in two different ways gives

$$N_{-\gamma,-\delta}\cdot\xi_{\gamma}\xi_{\delta}=N_{\gamma,\delta}\cdot\xi_{\gamma+\delta}$$

and thus by (27),

(29)

$$\xi_{\gamma+\delta} = -\xi_{\gamma}\xi_{\delta}$$

which contradicts (28).

REMARK 5.3.6. — Our carefulness about signs in the above proof seems unnecessary (as -1 is of valuation 0 anyway), but it becomes important when one wants to see why the analogous statement in the case $k = \mathbb{R}$ is *not* true, and indeed anisotropic forms exist in *every* type, all split by the quadratic $\mathbb{C}|\mathbb{R}$. There, in our sign convention, we would have $\xi_{\alpha} \equiv -1 \mod \mathbb{R}^{*2}$ for all roots α , and (29) would be perfectly OK; in Bourbaki's sign convention, we would instead have all $\xi_{\alpha} \in \mathbb{R}^{*2}$ (that is, > 0), but in that convention, we have $N_{-\alpha,-\beta} = N_{\alpha,\beta}$ instead of (27) and thus $\xi_{\gamma+\delta} = \xi_{\gamma}\xi_{\delta}$ instead of (29), again without contradiction.

Note that the condition in Proposition 5.3.5 – that R contains two roots whose sum is a root of the same length – can be reformulated as R containing a sub-root-system of type A_2 . We need a bit more than that that to derive a contradiction in the next case, $\ell = 3$.

PROPOSITION 5.3.7 (Case $\ell = 3$). — Let \mathfrak{g} be a semisimple Lie algebra over a \mathfrak{p} -adic field k which is split by the unramified cubic extension K|k. If the root system of \mathfrak{g}_K contains a sub-root-system of type A_3 , then \mathfrak{g} cannot be anisotropic.

In particular, if \mathfrak{g} is absolutely simple of exceptional type and anisotropic, it would need to be of type G_2 (which however will be excluded in the next proposition).

Proof. — Let σ be a generator of $\operatorname{Gal}(K|k)$, and again fix t and a Chevalley basis of e_{α} 's and H_{α} 's as before. Since σ_A is elliptic of order 3, we now have for each root α that: α and $\sigma(\alpha)$ are a basis of the sub-root system consisting of $\{\pm \alpha, \pm \sigma(\alpha), \mp \sigma^2(\alpha) = \pm (\alpha + \sigma(\alpha))\}$ which is of type A_2 , and the corresponding Chevalley basis vectors e_2

MÉMOIRES DE LA SMF 151

generate a subalgebra of \mathfrak{g}_K isomorphic to $\mathfrak{sl}_3(K)$. Again define ξ_α by $\sigma(e_\alpha) = \xi_\alpha e_{\sigma(\alpha)}$ for all $\alpha \in R$. Since the Galois-invariants of each of those subalgebras are subalgebras of the anisotropic \mathfrak{g} , they must each be the anisotropic k-form of \mathfrak{sl}_3 . As an analogue to equation (28) in the quadratic case, one can show ⁽³⁾ that this implies:

For all roots
$$\alpha$$
, $\xi_{\alpha}^2 \cdot \xi_{\sigma(\alpha)} \equiv \pi_k$ or $\pi_k^2 \mod \mathcal{O}_k^*$

or in other words, the valuation $v_k(\xi_{\alpha}^2 \cdot \xi_{\sigma(\alpha)}) \in \mathbb{Z}$ has to be non-trivial modulo 3 for every root $\alpha \in R$. Let us abbreviate the invariant $r(\alpha) := \xi_{\alpha}^2 \cdot \xi_{\sigma(\alpha)}$. Note that replacing σ by σ^2 would switch for each α whether $r(\alpha)$ is congruent to π or π^2 .

On the other hand, using that all occurring Chevalley structure invariants $N_{?,?}$ are ± 1 because all the roots have the same length, one computes

$$r(\alpha + \beta) = \pm r(\alpha)r(\beta)$$

for all roots α, β of the same length such that $\alpha + \beta$ is a root of the same length, which means

$$v_k(r(\alpha + \beta)) = v_k(r(\alpha)) + v_k(r(\beta)).$$

Now let $\alpha_1, \alpha_2, \alpha_3$ be an ordered basis of the sub-root-system of type A_3 existing by hypothesis. This means that all consecutive sums $(\alpha_1 + \alpha_2, \alpha_2 + \alpha_3, \alpha_1 + \alpha_2 + \alpha_3)$ are roots, all of the same length as any α_i . Abbreviate $v(1) := v_k(r(\alpha_1)), v(2) := v_k(r(\alpha_2))$ and $v(3) := v_k(r(\alpha_3))$. Now an application of Dirichlet's box principle leads to a contradiction: Not all of the integers

$$v(1), v(2), v(3)$$

$$v_k(r(\alpha_1 + \alpha_2)) = v(1) + v(2),$$

$$v_k(r(\alpha_2 + \alpha_3)) = v(2) + v(3) \text{ and }$$

$$v_k(r(\alpha_1 + \alpha_2 + \alpha_3)) = v(1) + v(2) + v(3)$$

can be $\not\equiv 0 \mod 3$ at the same time.

For the final assertion, it is readily checked that all exceptional root systems except those of type G_2 contain subsystems of type A_3 (except for type F_4 , their bases even contain bases of such subsystem; in F_4 on the other hand, the long and short roots are systems of type D_4 which in turn contain various copies of A_3).

$$x = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & \xi_\alpha \\ \xi_\alpha \xi_{\sigma(\alpha)} & 0 & 0 \end{pmatrix};$$

^{3.} Computations with matrices as in 3.2.5, from which we derived (28), become a bit awkward now. But one can use our knowledge about the Lie algebra of type A_2 in question being the derived algebra of a cyclic CSA and invoke the theory of [**Rei**, 30.4 and 31]. To this end, note that the decisive generating element in the enveloping associative algebra, written as a matrix, is

and whether the diagonal matrix $x^3 = \xi_{\alpha}^2 \cdot \xi_{\sigma(\alpha)} \cdot I$, or rather its diagonal entry, is a norm of K|k or not, decides whether that CSA contains nilpotent elements or not, i.e. whether our Lie algebra is not anisotropic or is anisotropic. Compare also [loc. cit., 14.6].

We can amend the case G_2 "by hand" – in fact, we have the very general

PROPOSITION 5.3.8. — There is no anisotropic Lie algebra over any (characteristic 0) field k which is of type G_2 and split by a cyclic extension of odd degree.

Proof. — (Cf. [We2, $\S2$].) Assume there is one and fix a maximal toral t as usual. A look at the root system (cf. [Bo2, Plate X], or this work's dedication) shows that the Galois group must act through a rotation of 120° , and hence factors through a cubic extension. The "long roots" form a closed, symmetric and irreducible subsystem, hence \mathfrak{g}_K has a simple subalgebra \mathfrak{H} which consists of the sum of \mathfrak{t}_K and the corresponding root spaces. Now \mathfrak{H} with the adjoint action naturally acts on the 6-dimensional sum V of the root spaces of the "short roots"; indeed \mathfrak{t}_K leaves each root space invariant, and a long root space $(\mathfrak{g}_K)_{\beta}$ sends a short root space $(\mathfrak{g}_K)_{\alpha}$ to 0 if $\alpha + \beta \notin R$, or else to $(\mathfrak{g}_K)_{\alpha+\beta}$ which is again a short root space, as the plate shows. So we have a 6dimensional representation of \mathfrak{H} . Also, V is a vector space complement of \mathfrak{H} in \mathfrak{g}_K . Inspecting the plate one sees further that the short roots decompose into two disjoint sets (in the form of equilateral triangles) stable under "adding a long root, if the result is a root"; correspondingly, V is the direct sum of two (3-dimensional) subspaces V_1 , V_2 invariant under the adjoint action of \mathfrak{H} , corresponding to those two sets of short roots. These representations are non-trivial homomorphisms $\mathfrak{H} \to \mathfrak{sl}_3(K)$ which are necessarily isomorphisms (by simplicity and dimension counting).

 \mathfrak{H} is invariant under our Galois action because the set of the long roots (hence the sum of the corresponding root spaces) and \mathfrak{t}_K are; V_1 and V_2 are invariant under our Galois action because the two equilateral triangles of the short roots are. So taking fixed sets under the Galois (=: Γ) action, $\mathfrak{h} := \mathfrak{H}^{\Gamma}$ is an 8-dimensional simple subalgebra of \mathfrak{g} , and $v_i := V_i^{\Gamma}$ are three-dimensional k-vector spaces whose direct sum is a complement of \mathfrak{h} in \mathfrak{g} , on which \mathfrak{h} acts with the restricted adjoint action. This gives non-trivial homomorphisms $\mathfrak{h} \to \mathfrak{sl}(v_i) \simeq \mathfrak{sl}_3(k)$ which again are necessarily isomorphisms. In other words, the simple subalgebra \mathfrak{h} is split, in contradiction to \mathfrak{g} being anisotropic.

After this digression, we settle the remaining case $\ell = 5$, where according to Proposition 5.3.4 only type E_8 can occur anyways. But it is easy now to guess the generalisation of the method for $\ell = 2$ and 3:

PROPOSITION 5.3.9 (Case $\ell = 5$). — Let \mathfrak{g} be a semisimple Lie algebra over a \mathfrak{p} -adic field k which is split by the unramified extension K|k of degree 5. If the root system of \mathfrak{g}_K contains a sub-root-system of type A_5 , then \mathfrak{g} cannot be anisotropic. In particular, no \mathfrak{g} of exceptional type is anisotropic and split by the unramified extension of degree 5.

Proof. — The same principle as in the proof of Proposition 5.3.7. This time, one first attaches to each root α a Galois-invariant subalgebra of type A_4 whose enveloping

110

matrix algebra contains the decisive element

$$x = \begin{pmatrix} 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & \xi_{\alpha} & 0 & 0 \\ 0 & 0 & 0 & \xi_{\alpha}\xi_{\sigma(\alpha)} & 0 \\ 0 & 0 & 0 & 0 & \xi_{\alpha}\xi_{\sigma(\alpha)}\xi_{\sigma^{2}(\alpha)} \\ \xi_{\alpha}\xi_{\sigma(\alpha)}\xi_{\sigma^{2}(\alpha)}\xi_{\sigma^{3}(\alpha)} & 0 & 0 & 0 & 0 \end{pmatrix},$$

with $x^5 = \xi^4_{\alpha} \xi^3_{\sigma(\alpha)} \xi^2_{\sigma^2(\alpha)} \xi_{\sigma^3(\alpha)} \cdot I$, so that in this case the invariant

$$r(\alpha) := \xi_{\alpha}^4 \xi_{\sigma(\alpha)}^3 \xi_{\sigma^2(\alpha)}^2 \xi_{\sigma^3(\alpha)}^3 \xi_$$

due to anisotropy is not allowed to be a norm of K|k, which boils down to:

For all roots
$$\alpha$$
, $v_k(r(\alpha)) \not\equiv 0 \mod 5$.

But again $v_k(r(?))$ is additive on the positive roots in the sub-root-system in the hypothesis, say with ordered basis $(\alpha_1, \ldots, \alpha_5)$, and Dirichlet's box principle forces r(?) to be of valuation divisible by 5 for at least one of the consecutive sums

$$? = \sum_{i=j}^{k} \alpha_i \quad \text{ for } 1 \le j \le k \le 5,$$

which is a contradiction.

Finally, according to Proposition 5.3.4, the only exceptional type that possibly could exist here is E_8 , which however does contain various copies of A_5 .

Summing up this section, we have proved:

THEOREM 5.3.10. — Let \mathfrak{g} be an absolutely simple Lie algebra of exceptional type over a \mathfrak{p} -adic field k, which is split by an unramified extension K|k of prime degree. Then \mathfrak{g} cannot be anisotropic.

Note that the technique developed apparently can be generalised to prove:

PROPOSITION 5.3.11. — Let \mathfrak{g} be a semisimple Lie algebra over a \mathfrak{p} -adic field, split by the unramified extension of prime degree ℓ . If the root system of \mathfrak{g} contains a subsystem of type A_{ℓ} , \mathfrak{g} cannot be anisotropic.

Caveat: The anisotropic \mathfrak{g} of inner type A which is split by degree ℓ extensions is not of type A_{ℓ} , but of type $A_{\ell-1}$ – the one that featured in Lemma 5.3.1, and was used in the proofs to derive the norm-/valuation-criterion on what we called $r(\alpha)$.

This is of limited use though, since for the exceptional types, only the three cases $\ell \leq 5$ left open by Proposition 5.3.4 are to be considered anyway, whereas for the classical types, we can exclude anisotropy by different, general arguments: which we will do in the next section.

5.4. Types AII, B, C and D over p-adic fields

For a \mathfrak{p} -adic field k, almost everything about these types follows from

THEOREM 5.4.1 (Witt). — A quadratic form in 5 or more variables over k is isotropic.

Proof. — See [Wit, Satz 16] or [Lam, Chap. 6, Thm 2.12]. \Box

This was extended by Jacobson (cf. [**Jac5**]), Ramanathan, Tsukamoto (cf. [**Tsu**]) and others to the following. For a common proof, see [**Kne**, I Anhang]:

PROPOSITION 5.4.2. — The following are isotropic over a p-adic field k:

- i. Hermitian forms in 3 or more variables over a quadratic extension C|k (see type AII).
- ii. Hermitian forms in 2 or more variables over the quaternion division algebra Q with standard involution γ (see type C).
- iii. Skew-hermitian forms in 4 or more variables over the quaternion division algebra (see type D).

5.4.1. Type AII. — The general discussion of the outer forms of type A (4.5.22) showed that they are classified by σ -hermitian forms on vector spaces over D, where (D, σ) is a division algebra over k of type AII; meaning that there is a quadratic extension C|k such that D is a C-central division algebra, and σ induces the non-trivial automorphism of C|k. As promised in the general discussion, our task is simplified by:

PROPOSITION 5.4.3 (Landherr). — If k is a p-adic field with D, σ and C as above, then C = D, so σ is the non-trivial element of Gal(C|k).

Proof. — There is a complicated proof in [Lan1, §4], and a nicer one in [Lan2, p. 211 et seq.] which is close to one by Jacobson and generalises to the following: First one shows that (D_C, σ_C) is isomorphic to $(D \times D^{\text{op}}, (x, y^{\text{op}}) \mapsto (y, x^{\text{op}}))$ as involutorial C-algebra (cf. [Inv, Prop. 2.15]). Consequently, $D \simeq D^{\text{op}}$ and the class of D has order ≤ 2 in Br(C). Thus if $D \neq C$, D would be the quaternion division algebra over C. Secondly, one shows that any quaternion algebra Q over C with σ as given is of the form $Q_0 \otimes_k C$ for a k-quaternion algebra Q_0 (cf. [Inv, Prop. 2.22]). But for \mathfrak{p} -adic k, every quadratic extension splits every quaternion algebra, so that $Q_0 \otimes_k C$ cannot be a skew field.

So the outer forms of type A_n $(n \ge 2)$ are given by hermitian forms on C^{n+1} , where C runs through the different quadratic extensions of k. By 5.4.2.i the Witt index r of such a form has to satisfy $n+1-2r \le 2$, and thus the Satake-Tits diagram (whose number of black vertices was n-2r) will be



if n is even and either



if n is odd. Their rational root systems, determined as in 3.4.1, are $BC_{n/2}$, $C_{(n+1)/2}$ and $BC_{(n+1)/2}$, respectively. The anisotropic kernel in the last case, by the erasing lemma, is an inner form of type A_1 and thus the anisotropic form of \mathfrak{sl}_2 . They all have C as fixed field of the twisted Galois action t, and indeed as minimal splitting field. For the third diagram, any other quadratic extension $C' \neq C$ splits the anisotropic kernel, so that $\mathfrak{g}_{C'}$ has the second diagram, with minimal splitting field C.C'.

So by he isomorphism theorem, there are as many isomorphism classes of outer forms of type A_n as quadratic extensions of k for even n, and twice as many for odd n > 1.

5.4.2. B_n

PROPOSITION 5.4.4. — A Lie algebra of type
$$B_n$$
 over k has Satake-Tits diagram

-0-

or

In the non-split case, the anisotropic kernel is the anisotropic form of \mathfrak{sl}_2 , and the rational root system is of type B_{n-1} . Any quadratic extension is a splitting field.

Proof. — By the general discussion of 4.5.1.2 we know that such a Lie algebra has the diagram



and is given by a quadratic form with Witt index r on k^{2n+1} . If we had r < n-1, we see by the discussion there (or by the erasing Lemma 4.3.2) that there would exist an anisotropic form in $2(n-r) + 1 \ge 5$ variables.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

or

The anisotropic kernel is determined by the erasing lemma and discussion of type A_1 . This also gives the assertion about the splitting field, and the rational root system is easily calculated like in Section 3.4.1.

5.4.3. C_n

or

PROPOSITION 5.4.5. — A Lie algebra of type C_n over k is either split, or its Satake-Tits diagram is either



according to whether n is odd or even. In the non-split case, the anisotropic kernel is a product of $\lfloor \frac{n+1}{2} \rfloor$ copies of the anisotropic form of \mathfrak{sl}_2 , and the rational root system is of type BC_r for odd n and of type C_r for even n, where $r = \lfloor \frac{n}{2} \rfloor$. Any quadratic extension is a splitting field.

Proof. — We had seen already in the general discussion of 4.5.1.4 that d = 1 gives the split form. Otherwise, as noted in 4.5.16, d = 2 and D = Q is the quaternion division algebra with its standard involution γ . By 5.4.2.ii, all non-degenerate hermitian forms are equivalent to the one given by the unit matrix, and hence for the Witt index r we have 2r = n - 1 if n is odd and 2r = n if n is even, which gives the diagrams. $\left|\frac{n+1}{2}\right| = n - r$ is the number of black vertices.

The anisotropic kernel is determined by the erasing lemma and discussion of type A_1 . This also gives the assertion about the splitting field, and the rational root system is easily calculated like in Section 3.4.1.

5.4.4. D_n

PROPOSITION 5.4.6. — A Lie algebra of type D_n over k which comes from an involutorial k-algebra whose underlying CSA is split has Satake-Tits diagram



or

or

In the last two cases, the rational root systems are of type B_{n-1} and B_{n-2} , respectively. The anisotropic kernel in the last case is a product of the anisotropic form of \mathfrak{sl}_2 with itself, and it is split by any quadratic extension. The first and third one are thus completely described up to isomorphism; for the second diagram, there is one isomorphism class of Lie algebras for every quadratic extension of k, and this quadratic extension splits it.

Proof. — Proceeding as for type B, we can exclude the case r < n-2. The first diagram is the case r = n, the split form. For r = n - 1 we have the quasi-split forms discussed in Section 4.5.1. The final case is r = n-2. By [Lam, Chap. 6, Cor. 2.15] up to isomorphism, there is a unique 4-dimensional anisotropic quadratic space (k^4, q) , corresponding to $S_a = \text{diag}(1, -u, -\pi, \pi u)$ (where π is a uniformiser and $u \in \mathcal{O}_k$ such that $k(\sqrt{u})$ is unramified; for $p \neq 2$, any $u \in \mathcal{O}_k \setminus \mathcal{O}_k^2$ will do) whose discriminant is the square $(\pi u)^2$. Its diagram is as described in the general discussion, and the anisotropic kernel and rational root system are determined as usual.

It remains to discuss the forms of type D_n where d = Ind(A) = 2 for the corresponding involutive algebra (A, ι) . Analogous to the discussion for type C, this is equivalent to the description of skew-hermitian forms over the quaternion division algebra Q with its standard involution. By 5.4.2.iii now necessarily $n - 2r \leq 3$ for the dimension n (over Q) and the Witt index r.

The anisotropic skew-hermitian forms are determined in [**Tsu**, Thm 3]: There is exactly one in the dimensions 3 and 0, giving the first and last of the following diagrams. In dimensions 2 and 1 there is one for each quadratic extension, the correspondence being given by the discriminant ⁽⁴⁾ in k^*/k^{*2} , and the corresponding quadratic extension K|k is the fixed field of the *t*-action in the middle two cases. The cases are:

n = 2r + 3 (n odd):



n = 2r (*n* even):

4. Defined as in the general discussion of 4.5.1.3 via the reduced norm, cf. [Tsu, 1.3].



The anisotropic kernels are, from top to bottom:

 \triangleright a product of r copies of the anisotropic form of \mathfrak{sl}_2 with the anisotropic form of \mathfrak{sl}_4 ;

 \triangleright a product of r copies of the anisotropic form of \mathfrak{sl}_2 with $R_{K|k}\mathfrak{a}$, where K|k is the mentioned fixed field of t and \mathfrak{a} is the anisotropic K-form of \mathfrak{sl}_2 ;

 \triangleright a product of r copies of the anisotropic form of \mathfrak{sl}_2 ;

 \triangleright ditto.

Finally, with the method of Section 3.4.1, the rational root systems are seen to be of type B_r in the first three cases and of type C_r in the last.

REMARK 5.4.7. — For n = 4, the phenomenon from 4.5.20.ii occurs and the case d = 1, r = 2 reappears as d = 2, r = 2. So $\mathfrak{S}(M_4(Q), \iota_1) \simeq \mathfrak{S}(M_8(k), \iota_2)$ with

$$\iota_1:\! X \mapsto \begin{pmatrix} 0 & -I_2 \\ I_2 & 0 \end{pmatrix} \cdot {}^t(\gamma(X)) \cdot \begin{pmatrix} 0 & I_2 \\ -I_2 & 0 \end{pmatrix}$$

and ι_2 coming from the quadratic form on k^8 of Witt index 2 (i.e. with anisotropic kernel the 4-dimensional anisotropic space). All other diagrams that we listed, and hence their Lie algebras, are visibly distinct.

5.5. Exceptional Lie algebras over p-adic fields

We now classify, over a p-adic field k, the absolutely simple Lie algebras of the *exceptional types*, by which we mean

 \triangleright those of types G_2, F_4, E_6, E_7, E_8 , and

 \triangleright the so-called *trialitarian forms* which are those of type D_4 where the twisted Galois action does not factor through a quotient of order ≤ 2 .

The main method is to use the results of Section 4.3, combined with the results for the classical types, to exclude most possible Satake-Tits diagrams. It should be noted that large parts of this (e.g. everything where only the result about inner types of A_n in 4.3.7 is used) work just as well over arbitrary fields, as Tits' list in [**Tit1**] suggests. The remaining diagrams are then shown to be admissible, except for the anisotropic ones. For these, we exploit the results in the prime degree case of Section 5.3. This way, we can avoid using Kneser's theorem except for one case in type E_6 and one in type D_4 (although to be clear, one should note that the cases E_7 and E_8 rely on E_6).

5.5.1. G₂

PROPOSITION 5.5.1. — Every Lie algebra of type G_2 over k is split.

MÉMOIRES DE LA SMF 151

 $\mathbf{116}$

a) With Kneser's theorem: The fundamental group P(R)/Q(R) is trivial and hence by Kneser's theorem there is only the split form.

b) Without Kneser's theorem: by 4.3.6, the only other possibilities are anisotropic ones, or a form with diagram

•==0

Assume first we had such a Lie algebra. It is split by the unramified quadratic extension K|k (since its anisotropic kernel is), so fix \mathfrak{s} , \mathfrak{t} as usual with \mathfrak{t}_K split, and call σ the non-trivial element of $\Gamma := \operatorname{Gal}(K|k)$. Looking at the root system (cf. [**Bo2**, Plate X], or this work's dedication) shows that the Galois action would have to consist precisely of the reflection s_{α_1} . In particular we would have $\beta - \sigma_A(\beta) = \alpha_1 \in R$ for $\beta = 2\alpha_1 + \alpha_2$. This is impossible by the following lemma, which looks \mathfrak{p} -adic but is inspired by an analogous assertion over the field \mathbb{R} in [**Ara**, Prop. 1.3].

LEMMA 5.5.2. — Let \mathfrak{g} be a semisimple Lie algebra with \mathfrak{s} , \mathfrak{t} as always such that \mathfrak{t} is split by the unramified quadratic extension K|k. Let σ be the non-trivial element of $\operatorname{Gal}(K|k)$. Then $\beta - \sigma_A(\beta)$ is not a root, for any root β .

Proof. — $\sigma_A = -\operatorname{id}$ on X_0 , so the assertion is clear for $\beta \in R_0$. For the general case, we resume notation from Sections 3.3.1 and 5.3: Again we fix a Chevalley basis consisting of H_{α} and bases e_{α} of the root spaces, with the structure constants $N_{\alpha,\beta}$ defined in (14) being integers. We have $N_{\alpha,\beta} = -N_{\beta,\alpha}$ for all roots α,β . Again we define ξ_{α} 's by $\sigma(e_{\alpha}) = \xi_{\alpha} e_{\sigma_A(\alpha)}$. Because $\sigma(H_{\alpha}) = [\sigma(e_{\alpha}), \sigma(e_{-\alpha})] = \xi_{\alpha} \xi_{-\alpha} H_{\sigma_A(\alpha)}$ and $\sigma_{\alpha}(\sigma(H_{\alpha})) = \sigma(\alpha(H_{\alpha})) = 2$, we have $\sigma(H_{\alpha}) = H_{\sigma_A(\alpha)}$ and $\xi_{-\alpha} = \xi_{\alpha}^{-1}$ for all $\alpha \in \mathbb{R}$.

Now assume there is $\beta \in R$ with $\beta - \sigma_A(\beta) \in R$. Then $N_{\beta, -\sigma_A(\beta)} \in \mathbb{Z} \setminus \{0\}$. Computing $[\sigma(e_\beta), e_{-\beta}] = \sigma([e_\beta, \sigma(e_{-\beta})])$ in two different ways gives

$$\xi_{\beta}N_{\sigma_A(\beta),-\beta} \cdot e_{\sigma_A(\beta)-\beta} = \sigma(\xi_{-\beta})\xi_{\beta-\sigma_A(\beta)}N_{\beta,-\sigma_A(\beta)} \cdot e_{\sigma_A(\beta)-\beta}$$

and hence

$$\xi_{\beta} = -\sigma(\xi_{\beta}^{-1}) \cdot \xi_{\beta - \sigma_{A}(\beta)}$$

or, setting $\gamma = \beta - \sigma_A(\beta)$:

$$\xi_{\gamma} = -N_{K|k}(\xi_{\beta})$$

But on the other hand, the subalgebra $KH_{\gamma} + Ke_{\gamma} + Ke_{-\gamma}$ is stable under the Galois operation, hence descends to a form of $\mathfrak{sl}_2(K)$ which is anisotropic because $\gamma \in R_0$, and the argument in the proof of Proposition 5.3.5 shows that $\xi_{\gamma} \notin \pm N_{K|k}(K)$ or in other words, its π_k -adic valuation is odd, hence we have a contradiction and the lemma is proven.

So there can be only split or anisotropic forms.

Now assume there is an anisotropic one. Choose an unramified splitting field K|k. Since K|k is cyclic, we can "filter" the extension to a tower of fields $k = K_0 \subset K_1 \subset \cdots \subset K_n = K$ such that each $K_{i+1}|K_i$ is unramified of prime degree.

Now since for all *i*, the scalar extension \mathfrak{g}_{K_i} is either split or anisotropic, and of course scalar extensions of a split form remain split, there must be exactly one step such that \mathfrak{g}_{K_i} is anisotropic but $(\mathfrak{g}_{K_i})_{K_{i+1}} = \mathfrak{g}_{K_{i+1}}$ is split. But then \mathfrak{g}_{K_i} is an anisotropic form of type G_2 over the \mathfrak{p} -adic field K_i , with K_{i+1} , an unramified extension of prime degree, as splitting field: which we excluded in Section 5.3. The proposition is proven.

5.5.2. F₄

PROPOSITION 5.5.3. — Every Lie algebra of type F_4 over k is split.

Proof. — a) Again P(R)/Q(R) is trivial and hence by Kneser's theorem, only the split form exists.

b) Without using Kneser's theorem: First, all diagrams containing a white vertex are easily ruled out. Namely, if the rightmost vertex α_4 is white and the form is not split, by applying the erasing Lemma 4.3.2 to that vertex and comparing with type B_3 , the only possibility is



which is excluded by erasing α_2 and seeing an impossible diagram of type A_2 to its right. So α_4 is black. If α_1 were white, erasing it and looking at type C_3 gives the only option



which is impossible by erasing α_3 . So α_1 is black. By 4.3.6 we are left with



which again is impossible as erasing one white vertex shows.

So again, there are only split or anisotropic forms.

The same procedure as for type G_2 , filtering an unramified splitting field to subextensions of prime degrees, then excludes anisotropic forms via Section 5.3, and the proposition is proven.

5.5.3. E_6 , inner forms

PROPOSITION 5.5.4. — Over k, there are two inner forms of Lie algebras of type E_6 : The split one, and one whose Satake-Tits diagram is



MÉMOIRES DE LA SMF 151

and whose anisotropic kernel is the product of two copies of the anisotropic form of \mathfrak{sl}_3 . The rational root system is of type G_2 , and a splitting field is the unramified cubic extension.

Proof. — Below is a list of all possible diagrams, except for the anisotropic one, which are allowed by 4.3.1 (on the left). Most of them are then directly excluded by an application of the erasing Lemma 4.3.2: we signify a part of the diagram which, according to that lemma (that is, after erasing certain white vertices and possibly restricting to a connected component) would have to be k-admissible, but is not, according to our discussion of the classical types.





we note that if it were k-admissible – being given by the Lie algebra \mathfrak{g} , say – then the anisotropic kernel \mathfrak{g}_a would be of inner type A_5 and thus be the anisotropic form of \mathfrak{sl}_6 . This means that it is split by the unramified extension of degree 6, which is the composite field of the unramified extensions of degree 3 and 2, call them K and L. Then $(\mathfrak{g}_a)_K$, being a non-split inner form of type A_5 which is split by the unramified quadratic extension L.K|K, would have Satake-Tits diagram



and consequently the diagram of \mathfrak{g}_K would have to be



which was excluded in the list above.

Let us now describe the non-split admissible form



It exists by the patching Proposition 4.3.3. Concretely, we patch: Firstly, the admissible form of type A_5



and secondly, a product of $\mathfrak{sl}_2(k)$ with two copies of the anisotropic form of \mathfrak{sl}_3 :

Both have anisotropic kernel as described and so has their patching. The assertion about the splitting field is then clear, and the rational root system was calculated in Section 3.4.1.

0

We have thus dealt with all possible forms except for anisotropic ones. Kneser's and Satake's table show that there are two inner forms of type E_6 , so no anisotropic ones can occur.

Without Kneser's theorem, we can at least get the following information: assume \mathfrak{g} is an anisotropic inner form of type E_6 , and choose an unramified splitting field of minimal degree K|k. Again filter the extension as $k = K_0 \subset K_1 \subset \cdots \subset K_n = K$ such that each $K_{i+1}|K_i$ is unramified of prime degree. Look at the forms \mathfrak{g}_{K_i} . Now let $0 \leq i < n$ be the first index such that $\mathfrak{g}_{K_{i+1}}$ is not anisotropic. By Section 5.3, $\mathfrak{g}_{K_{i+1}}$ cannot be split, thus it has to be the form with diagram



Further, due to minimality and what we know about this form, necessarily n = i+2and $[K_n : K_{i+1}] = 3$. Let us replace k by K_i and call $L := K_{i+1}$, then we have a diagram of fields



where $\ell := [L : K]$ is a prime and F is the unramified cubic extension of k. But then, which form is \mathfrak{g}_F ? Again due to minimality, it cannot be the split form; and again

due to Section 5.3, it cannot be anisotropic; so it must be the form with the same diagram as above. Hence, due to minimality, necessarily $\ell = 3$ and actually F = L.

So what we would like to rule out is an anisotropic form, split by the unramified extension of degree 9, whose scalar extension to the intermediate unramified extension of degree 3 would be the non-split non-anisotropic form mentioned above. Sadly, we were not able to lead this to a contradiction. (The method we used so far will however help in cases E_7 and D_4 , see below.)

5.5.4. E₇

PROPOSITION 5.5.5. — Over k, there are two forms of Lie algebras of type E_7 : The split one, and one whose Satake-Tits diagram is



and whose anisotropic kernel is the product of three copies of the anisotropic form of \mathfrak{sl}_2 . The rational root system is of type F_4 . A splitting field is the unramified quadratic extension.

Proof. — As before we rule out other diagrams with at least one white vertex. If the "lower vertex" α_2 is white, the "upper row" has to be a k-admissible diagram of inner type A_6 , so the only non-split possibility is:



To exclude (*), we generalise the method of the erasing Lemma 4.3.2; cf. [**PSt**, Section 5, Lemma 3]. First note the

LEMMA. — Let R be an irreducible root system with basis $B = \{\alpha_1, \ldots, \alpha_r\}$, and let α_0 be the negative of its highest root. Then for any proper subset $B' \subsetneq B$, the intersection of the \mathbb{Z} -span of $B' \cup \{\alpha_0\}$ with R is a root subsystem R' of R (in the vector space it generates) with basis $B' \cup \{\alpha_0\}$.

The proof is straightforward; for the last assertion one uses that the coefficient of any α_j $(j \ge 1)$ in any root is less or equal to its coefficient in the highest root. The lemma implies: if one erases a non-empty set of vertices from the *completed Dynkin diagram* of R, what remains is a (possibly non-connected) Dynkin diagram of a root system. One can indeed check this case-by-case in the plates of [**Bo2**, VI]. Now the completed Dynkin diagram of type E_7 is:



MÉMOIRES DE LA SMF 151

Assume there were a Lie algebra \mathfrak{g} over k having (*) as Satake-Tits diagram for the usual choices including a Γ -basis Δ . Form the root subsystem R' as in the lemma for $\Delta' := \Delta \setminus \{\alpha_2\}$. Setting $\overline{R}' = \rho(R') \setminus \{0\}$, with 3.4.2 one checks that this is the subsystem of \overline{R} generated by $\rho(\Delta' \cup \{\alpha_0\}) \setminus \{0\}$ (which is just $\{\rho(\alpha_0)\}$ in this case) and that $\rho^{-1}(\overline{R}' \cup \{0\}) = R'$. Proceeding then as in the erasing lemma, one would get a semisimple Lie algebra \mathfrak{g}' over k which would be generated by the anisotropic kernel \mathfrak{g}_a and the root space of $\rho(\alpha_0)$, and whose Satake-Tits diagram would be:



This is an impossible form of type A_7 , so (*) is excluded (over any field k). Hence we can assume from now on that the lower vertex α_2 is black. Then the "rightmost vertex" α_7 is black, since if not, erasing it gives a non-admissible diagram of type E_6 .

Case 1. — The "leftmost vertex" α_1 is black. If the second to the right vertex α_6 were white, on the left of it there would be an admissible form of D_5 which looks like



where the gray vertices can be white or black. Comparing with our list there we see that this is impossible: so α_6 is black. The same game for α_5 and forms of type A_4 shows that α_5 must be black, then repeating it for α_4 and A_2 shows that we are left with the diagrams



and

The first two (of k-rank 1) are excluded by 4.3.6. The anisotropic one is left aside.

Case 2. — The "leftmost vertex" α_1 is white. Then erasing it must give an admissible diagram of type D_6 which looks like



where the gray vertices are either black or white. Comparing with our list for type D_6 we see that the only possibility corresponds to the one named in the proposition:



This exists by the patching Proposition 4.3.3. Concretely, we patch: Firstly, the direct product of an admissible form of type D_5 and the anisotropic form of \mathfrak{sl}_2 :



Secondly, the direct product of an admissible form of type A_3 and the anisotropic form of \mathfrak{sl}_2 :



Both have a threefold product of the anisotropic form of \mathfrak{sl}_2 as anisotropic kernel and so has our patched form of type E_7 . The assertion about the splitting field is clear and the rational roots are easily computed like in Section 3.4.1.

We have thus dealt with all possible forms except for anisotropic ones. Kneser's and Satake's table show that there are two forms of type E_7 , so no anisotropic ones can occur.

We can, however, rule out anisotropic forms without Kneser's theorem:

PROPOSITION 5.5.6. — There is no anisotropic form of type E_7 over k.

Proof. — Assume \mathfrak{g} were one and choose an unramified splitting field of minimal degree K|k. Again filter the extension as $k = K_0 \subset K_1 \subset \cdots \subset K_n = K$ such that each $K_{i+1}|K_i$ is unramified of prime degree, and let $0 \leq i < n$ be the first index such that $\mathfrak{g}_{K_{i+1}}$ is not anisotropic. By Section 5.3, $\mathfrak{g}_{K_{i+1}}$ cannot be split, thus it has to be the form with diagram



MÉMOIRES DE LA SMF 151

Further, due to minimality and what we know about this form, necessarily n = i+2and $[K_n : K_{i+1}] = 2$. Let us replace k by K_i and call $L := K_{i+1}$, then we have a diagram of fields



where $\ell := [L:K]$ is a prime and F is the unramified quadratic extension of k. But which form is \mathfrak{g}_F then? Again due to minimality, it cannot be the split form; and again due to Section 5.3, it cannot be anisotropic; so it must be the form with the same diagram as above. Hence, due to minimality, necessarily $\ell = 2$ and actually F = L.

So we have an anisotropic \mathfrak{g} which is split by the unramified extension of degree 4. Make the usual choices of $\mathfrak{t}, \mathfrak{s}, R, \ldots$, and let σ be a generator of $\operatorname{Gal}(K|k)$. Then the minimal polynomial of σ_A as element of $\operatorname{GL}(V)$ (where V is the \mathbb{R} -vector space spanned by the root system R) must divide $T^3 + T^2 + T + 1 = (T+1)(T^2+1)$.

Choose an A(R)-invariant scalar product on V. We will now look at the cyclic $\mathbb{R}[\sigma_A]$ -module generated by any given root α , which will be of \mathbb{R} -dimension 1, 2 or 3. In each case we will derive a contradiction.

1) If the $\mathbb{R}[\sigma_A]$ -module generated by α is one-dimensional over \mathbb{R} – equivalently, $\sigma_A(\alpha) = -\alpha$ – the space spanned by α and thus also its orthogonal complement $\langle \alpha \rangle^{\perp}$ are Galois-stable and thus define subalgebras of \mathfrak{g} which are necessarily anisotropic. But the roots orthogonal to any given root in E_7 form a root system of type D_6 , and we know from the discussion of the classical types that no Lie algebra of this type can be anisotropic. So σ_A cannot be multiplication by -1 on any root.

2) I the $\mathbb{R}[\sigma_A]$ -module generated by α is two-dimensional over \mathbb{R} – equivalently, $\sigma_A^2(\alpha) = -\alpha$ – it is readily seen that α and $\sigma(\alpha)$ must be orthogonal to each other; in this case, the roots orthogonal to their span form a root system of type $A_1 \times D_4$, and again we know that no anisotropic form of classical type D_4 (no triality since $\operatorname{ord}(\sigma_A) = 4$)) exists.

3) So for each root α , the σ_A -invariant subspace spanned by its σ_A -conjugates in V must be 3-dimensional, and the minimal polynomial of σ_A is indeed $T^3 + T^2 + T + 1$. Then by corollary 5.3.2, the σ_A -conjugates of any root α are the basis of a subsystem of type A_3 . But the roots orthogonal to such a (Galois-invariant) subsystem in E_7 form a (Galois-invariant) subsystem of type $A_1 \times A_2$, which is impossible: one can see that there is no way in which σ_A with said minimal polynomial can operate on $A_1 \times A_2$ without stabilising both factors, which would give a 1- and a 2-dimensional Galois-invariant subsystem, contradicting the above results about their non-existence.

5.5.5. E₈

PROPOSITION 5.5.7. — Every Lie algebra of type E_8 over k is split.

Proof. — The root system of type E_8 has trivial P(R)/Q(R), so by Kneser's theorem, only one form can exist, and this is the split one. Still, our usual procedures allow us to rule out the other diagrams directly.

Assume first we had a diagram with at least one white vertex. If the "rightmost vertex" α_8 were white, by the classification of E_7 the only option for a non-split form would be



but the segment formed by α_7 and α_8 is not admissible in type A_2 . So α_8 is black. If α_7 were white, the possibilities (by admissible diagrams of inner type E_6) are



which are both excluded e.g. by erasing the "lower vertex" α_2 . So α_7 is black. Now if the lower vertex α_2 were white, the "upper row" would have to be an admissible form of type A_7 ending in two black vertices. This gives the possibilities



and

The first is excluded by erasing α_5 and looking at the left part, the second is impossible by 4.3.6. So α_2 is black. If the "third to the right" vertex α_6 were white, by the classification of type D_5 , the only option is



which is seen to be impossible by erasing the left white vertex. So α_6 is black. If α_5 were white, by the classification of type A_4 the only option is



which is impossible by 4.3.6. So α_5 is black. If the leftmost α_1 were white, erasing it would give a non-admissible form of type D_7 , so α_1 is black. Since by 4.3.6 there is no form of k-rank 1 left, and



is impossible – as erasing any of the white vertices shows –, we are left with all-black anisotropic forms.

But since we know now that only anisotropic and split forms can exist, the technique of reducing to an unramified splitting field of prime extension, which is settled by Section 5.3, works exactly as in the cases G_2 and F_4 .

5.5.6. D_4 , trialitarian types

PROPOSITION 5.5.8. — Except for the quasi-split forms described in Section 4.6, no other Lie algebras of trialitarian type exist over k.

We had already noted in Section 4.3.3 that the diagrams



are not admissible over any field. Assume we had a \mathfrak{g} with diagram



where K := the fixed field of the twisted Galois action would be an extension of k with $\operatorname{Gal}(K|k)$ cyclic of order 3, resp. isomorphic to the symmetric group S_3 . Then \mathfrak{g}_K would have the diagram



– a non-trialitarian form of type D_4 – which however is not admissible over \mathfrak{p} -adic fields according to the list in Subsection 5.4.4.⁽⁵⁾ (That the diagram of \mathfrak{g}_K would be as described is not as trivial as it might seem; but it follows from Section 4.1 (cf. Remark 3.4.1) that the anisotropic kernel \mathfrak{g}_a of such \mathfrak{g} , with diagram



would necessarily be the scalar restriction $R_{K|k}\mathfrak{a}$ of $\mathfrak{a} :=$ the anisotropic K-form of \mathfrak{sl}_2 , and we would have $(\mathfrak{g}_a)_K \simeq \mathfrak{a}^3$. From this one can conclude.)

Now let us rule out all anisotropic forms, whose diagrams would be



Case I (Twisted Galois action via $\mathbb{Z}/3$). — Again let K be the fixed field of the twisted Galois action. Then the diagram for \mathfrak{g}_K contains no arrows and thus \mathfrak{g}_K , due to our results so far (including the classification of non-trialitarian types), is either the split form



5. According to the list in [**Tit1**], such forms *do* exist over number fields; correspondingly, the described diagram of a non-trialitarian form *is* admissible over number fields.

or has, up to permutation of the outer vertices, the diagram



With Galois theory, we will now reduce case Ib to Ia. Namely, \mathfrak{g}_K with this diagram is split by any quadratic extension L|K; choose one such that $\operatorname{Gal}(L|k) \simeq \operatorname{Gal}(L|K) \times \operatorname{Gal}(K|k)$, i.e. L = K[x] with $x^2 \in k, x \notin K$. Set $\ell := k(x)$:



Then our results so far imply that the Lie algebra \mathfrak{g}_{ℓ} would have diagram



and have L as a splitting field, with $\operatorname{Gal}(L|l) \simeq \operatorname{Gal}(K|k) \simeq \mathbb{Z}/3$ being the Galois group that acts on the diagram. Thus in case Ib.1, \mathfrak{g}_{ℓ} with its cubic splitting field L is an instance of case Ia. Also:

LEMMA 5.5.9. — Case Ib.2 cannot occur.

Proof. — We use the method that allowed us to exclude anisotropic forms in type E_7 in Proposition 5.5.6. Namely, we make the usual choices, including a generator $\sigma \in \text{Gal}(L|k) \simeq \mathbb{Z}/6 \simeq \mathbb{Z}/2 \times \mathbb{Z}/3$. The minimal polynomial of $\sigma_A \in A(R)$ must divide

$$T^{5} + T^{4} + T^{3} + T^{2} + T + 1 = (T+1)(T^{2} + T + 1)(T^{2} - T + 1)$$

and again we look at the $R[\sigma_A]$ -module in $V = \mathbb{R}X$ generated by one given root α .

1) If it is one-dimensional over \mathbb{R} , the roots orthogonal to it form a root system of type $A_1 \times A_1 \times A_1$, and without loss of generality, we can pick the three orthogonal roots in that product to be basis roots α_1, α_3 and α_4 ; whereas the root α we started with would be $-\alpha_1 - 2\alpha_2 - \alpha_3 - \alpha_4$, the negative of the highest root. But using that σ_A

is elliptic and of order 6, one concludes that modulo renumbering α_1, α_3 and α_4, σ_A would have to map

 $\alpha_1 \longmapsto -\alpha_3 \longmapsto \alpha_4 \longmapsto -\alpha_1 \longmapsto \alpha_3 \longmapsto -\alpha_4$

and operate as -id on $\{\pm \alpha_2, \pm \alpha_1 + \alpha_2 + \alpha_3 + \alpha_4, \pm \alpha_1 + \alpha_2 + \alpha_3 + \alpha_4\}$. This last set, however, is a root system of type A_2 , and the subalgebra of \mathfrak{g} generated by it would be anisotropic, but split by the quadratic extension $\ell = L^{\sigma^2 = id} | k$ (and also be an outer form), which is impossible by our classification of type A.

So on no root α , σ_A operates as -id.

2) Now assume the $\mathbb{R}[\sigma_A]$ -module generated by a root α is two-dimensional over \mathbb{R} . Considering the minimal polynomial, we must either have

$$\sigma_A^2(\alpha) + \sigma_A(\alpha) + \alpha = 0$$
 or $\sigma_A^2(\alpha) - \sigma_A(\alpha) + \alpha = 0.$

In both cases, α and $\sigma_A(\alpha)$ would be the basis of a sub-root-system of type A_2 ; in the second case however, it is easily seen (similar to Lemma 5.3.1) that σ_A would be the rotation by 60°, which is not an element of the Weyl group, thus again we would find an anisotropic *outer* form of type A_2 as subalgebra in \mathfrak{g} , which is impossible. So if the $\mathbb{R}[\sigma_A]$ -module spanned by a root α is two-dimensional, we have $\sigma_A^2(\alpha) + \sigma_A(\alpha) + \alpha = 0$. This means, however, that $\sigma_A^3 = \mathrm{id}$ on these submodules; so if this would be the case for all $\alpha \in R$, the whole algebra would be split by the cubic extension $K = L^{\sigma^3 = \mathrm{id}}|k$, a contradiction.

So there must be a root $\alpha \in R$ such that the $\mathbb{R}[\sigma_A]$ -module generated by it is of dimension 3 or 4 as \mathbb{R} -vector space.

3) If its dimension is 3, the roots in that subspace must form a sub-root-system of rank 3. Since we are in type D_4 , the possibilities for such a sub-root-system would be

$$(A_1)^3$$
 and A_3 .

(There is no subsystem of type $A_1 \times A_2$ in a root system of type D_4 .) In the first case (which means that we have three orthogonal roots), there is a fourth root in R orthogonal to all of them, and we are back in the excluded case of a one-dimensional $\mathbb{R}[\sigma_A]$ -module. The second case is impossible because an anisotropic form of type A_3 , is not split by an extension of degree 6.

4) Remains the case that for every root α , the σ_A -conjugates of α span the whole space V. In this case, the minimal polynomial of σ_A would have to have degree 4, hence be

$$(T2 + T + 1)(T2 - T + 1) = T4 + T2 + 1.$$

Applying the proof of Lemma 5.3.1 to σ_A^2 we see that for any root α , α and $\sigma_A^2(\alpha)$ are a basis of a root system of type A_2 . Choosing an A(R)-invariant scalar product $\langle ., . \rangle$ on V and fixing the (unique) root length as $\langle \beta, \beta \rangle = 1$ for every root β , we have that $\langle \alpha, \sigma_A(\alpha) \rangle = \langle \sigma_A(\alpha), \sigma_A^2(\alpha) \rangle \in \{0, \pm \frac{1}{2}\}$. However, the orthogonal complement of any sub-root-system of type A_2 in D_4 contains no roots, so $\langle \alpha, \sigma_A(\alpha) \rangle = 0$ is excluded.

MÉMOIRES DE LA SMF 151

But if $\langle \alpha, \sigma_A(\alpha) \rangle = \frac{1}{2}$ (i.e. the angle between them is 60°), one computes (using $\langle \alpha, \sigma_A^2(\alpha) \rangle = -\frac{1}{2}$)

$$\langle \sigma_A(\alpha) - \alpha - \sigma_A^2(\alpha), \sigma_A(\alpha) - \alpha - \sigma_A^2(\alpha) \rangle = 0$$

hence $\sigma_A(\alpha) = \alpha + \sigma_A^2(\alpha)$. With a similar computation for $\sigma_A^3(\alpha)$, one sees that the σ -conjugates of α remain in a two-dimensional subspace, in contradiction to our assumption. Analogously, if $\langle \alpha, \sigma_A(\alpha) \rangle = -\frac{1}{2}$ (i.e., the angle is 120°), we have

$$\langle \sigma_A(\alpha) + \alpha + \sigma_A^2(\alpha), \sigma_A(\alpha) + \alpha + \sigma_A^2(\alpha) \rangle = 0,$$

hence $\sigma_A(\alpha) = -\alpha - \sigma_A^2(\alpha)$, and again one arrives at a contradiction.

The upshot of all this is:

COROLLARY 5.5.10. — If, over every \mathfrak{p} -adic field, there is no anisotropic form with diagram



and split by the cubic fixed field of the twisted Galois action (case Ia above), then there is no anisotropic form with that diagram over any \mathfrak{p} -adic field at all.

Case II (Twisted Galois action via S_3)

We can reduce this case to the same case I.a above, showing

PROPOSITION 5.5.11. — If, over every \mathfrak{p} -adic field, there is no anisotropic form with diagram



and split by the cubic fixed field of the twisted Galois action (case Ia above), then there is no anisotropic form of type D_4 over any \mathfrak{p} -adic field at all.

Proof. — We have to rule out diagrams

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017



where the Galois group operates via all of the symmetric group S_3 . Assume a form \mathfrak{g} over a \mathfrak{p} -adic field k with such a diagram given. Let K be the fixed field of the twisted Galois action, so that $\operatorname{Gal}(K|k) \simeq S_3$, and let F be the fixed field of the alternating subgroup A_3 , so that F|k is the unique quadratic subextension of K|k:



In the diagram of \mathfrak{g}_F , the twisted Galois action operates via $A_3 \simeq \mathbb{Z}/3$, so it is



The first case is excluded by our assumption and the last corollary. So we are in the second case. Then K is a splitting field for \mathfrak{g} . As usual, choose $\mathfrak{t}, \mathfrak{s}, R, \ldots$ and an A(R)-invariant scalar product on V, the vector space around the root system. Let τ be a 3-cycle and σ a transposition in S_3 . Note that due to anisotropy $(X = X_0 = X_a)$, we have

$$v + \tau_A(v) + \tau_A^2(v) + \sigma_A(v) + \sigma_A\tau_A(v) + \sigma_A\tau_A^2(v) = 0$$

for all $v \in V$, in particular there is no $v \neq 0$ fixed by both τ_A and σ_A . We look at the $\mathbb{R}[S_3]$ -modules generated by a root α (and what we just said amounts to: this module is not allowed to contain the trivial representation).

1) If one such module is one-dimensional over \mathbb{R} , τ_A operates trivially and σ_A as - id on it (in the language of S_3 -representations, it is the sign representation). The roots in its orthogonal complement are a system of type $(A_1)^3$, and τ_A must permute the factors in this triple cyclically. If σ_A operated trivially on that triple, then for $v \neq 0$ in it, $v + \tau_A(v) + \tau_A^2(v) \neq 0$ would be a vector fixed by all S_3 , which is impossible.

One easily sees (using that $\sigma \tau \sigma = \tau^2$ which excludes $\sigma_A = -id$ on the triple) that choosing any root β in the triple, with

$$\beta_1 := \beta, \quad \beta_2 := \tau_A(\beta), \quad \beta_3 = \tau_A^2(\beta),$$

 σ_A must fix one β_i and $\sigma_A(\beta_j) = -\beta_k$ for the other two indices $j, k \neq i$. The Lie subalgebra of \mathfrak{g} generated by the root spaces of this $(A_1)^3$ thus has the full $S_3 \simeq \operatorname{Gal}(K|k)$ operating on it, and its diagram is



It follows from Section 4.1 (cf. remark 3.4.1) that this Lie algebra is the scalar restriction $R_{K|k}\mathfrak{a}$ of $\mathfrak{a} :=$ the anisotropic K-form of \mathfrak{sl}_2 . In particular, it would become \mathfrak{a}^3 after scalar extension to K, hence not be split by K, contradicting our assumption.

2) By the representation theory of S_3 , a two-dimensional module must be the socalled standard representation. The roots in it would be a subsystem of type A_2 , with τ_A operating as rotation by 120° in one direction, and σ_A operating as a reflection s_β associated with one of the contained roots β (i.e. S_3 operates as the full Weyl group $W(A_2)$). However, the relations between the roots in a system of type D_4 make it impossible to define actions of τ_A and σ_A on all of V consistently. (Any root outside our subsystem of type A_2 will be non-orthogonal to several roots inside it; and all these angles will have to be held invariant under the operations of σ_A and τ_A , often already prescribing how these elements will act on the roots; going through the calculations, one always reaches a contradiction.)

3) If the $\mathbb{R}[S_3]$ -module generated by a root α is three-dimensional, the root system in it must be of rank 3, so of type $(A_1)^3$ or A_3 (there is no subsystem of type $A_2 \times A_1$ in D_4). The first case leads, via orthogonal complement, back to a one-dimensional module already excluded; whereas the anisotropic subalgebra corresponding to a root system of type A_3 would not be split by an extension of degree 6.

4) According to the representation theory of S_3 , there is no cyclic $\mathbb{R}[S_3]$ -module of \mathbb{R} -dimension > 3 without containing the trivial representation.

REMARK 5.5.12. — For the quasi-split trialitarian types, it remains to add to the description in 4.6 which Galois extensions K|k with Gal(K|k) isomorphic to $\mathbb{Z}/3$ or S_3 exist for a p-adic field k.

i. Cubic extensions of k are either unramified or totally ramified. The unique unramified one is Galois. The ramified ones are given (not one-to-one) by Eisenstein

polynomials $X^3 + aX^2 + bX + c \in \mathcal{O}_k[X]$ where a, b, c are divisible by π_k , and $\pi_k^2 \nmid c$. Its discriminant is $\Delta = a^2b^2 - 4b^3 - 4a^3c - 27c^2 + 18abc$, and it is known that the corresponding extension is Galois if and only if $\Delta \in k^{*2}$.

If $p \neq 3$, this can be arranged if and only if k contains a primitive third root of unity; equivalently $3 \mid q - 1$, where q is the order of the residue field of k; or equivalently, $-3 \in k^{*2}$. In this case, the $\mathbb{Z}/3$ -extensions are in obvious bijection with the subgroups of $k^*/(k^*)^3$ of order 3; in particular $k(\sqrt[3]{\pi_k})$ is a ramified Galois cubic extension of k.

ii. An extension K|k with $\operatorname{Gal}(K|k) \cong S_3$ is the Galois closure of any of its three subextensions L|k with [L:k] = 3, which are not Galois; and conversely, any non-Galois cubic extension L|k has such a K|k as normal closure. As noted in part i., a non-Galois cubic L|k is given by an Eisenstein polynomial of degree 3 whose discriminant Δ is not a square in k. We then have $K = L(\sqrt{\Delta})$, and the inertia degree f(K|k) is 1 or 2, according to whether $f(k(\sqrt{\Delta})|k)$ is 1 or 2, i.e. whether $v_k(\Delta)$ is odd or even. As Weisfeiler's table claims, both cases can occur: if k does not contain a primitive third root of unity, the splitting field K of $X^3 - \pi_k$ is an example, for which f(K|k) = 2 if $p \neq 3$ and f(K|k) = 1 if $k = \mathbb{Q}_3$.

5.5.7. E_6 , outer forms

PROPOSITION 5.5.13. — The only outer forms of type E_6 over k are the quasi-split ones which we described in Section 4.6 as being in bijection with k^*/k^{*2} .

As noted in Section 5.2, the cohomological approach shows that the outer forms of type E_6 are quasi-split. But we can do this without Kneser's theorem. First, we rule out other diagrams with white vertices. If the "lower vertex" α_2 is white, erasing it must give an admissible diagram of outer type A_5 with at least one black vertex, so the whole diagram is



But then erasing the two proper orbits leaves a non-admissible diagram of type A_2 . So α_2 is black. Now the same reasoning rules out



and



Erasing the outer proper orbit in



and comparing with the classification of type D_4 shows that these diagrams are not k-admissible. Neither is



as can be seen, for example, by the generalised erasing method with which we excluded the diagram (*) of type E_7 in 5.5.5: The completed Dynkin diagram for E_6 is



and erasing the upper middle vertex in the diagram in question leaves a connected component

Ì

which does not occur in type A_2 over any field.

Invoking 4.3.1 and 4.3.6 we see that the all-black *anisotropic form* is the only possibility left. So the proposition is reduced to the following one, which we can prove with our methods (although, to be fair, relying on the non-existence of anisotropic *inner* forms of type E_6):

PROPOSITION 5.5.14. — There is no anisotropic outer form of type E_6 over k.

Proof. — Assume there is one, and let K be the quadratic extension of k which is the fixed field of the twisted action. \mathfrak{g}_K is an inner form of type E_6 , hence by their classification either the split one



In the first case, \mathfrak{g}_K split, we choose $\mathfrak{t}, \mathfrak{s}, R...$ and a generator $\sigma \in \operatorname{Gal}(K|k)$ as usual and see that $\sigma_A = -\operatorname{id}$, hence any given root $\alpha \in R$ is part of a Galois-stable A_1 (namely, $\pm \alpha$); but the roots orthogonal to α (with respect to an A(R)-invariant scalar product) form a root system of type A_5 and are again stable under σ_A , giving rise to an anisotropic subalgebra of type A_5 which is split by a quadratic extension, contradicting what we know about type A.

So \mathfrak{g}_K must be the non-split form with the diagram above, which is split by any extension L|K of degree 3. Choose one such that $\operatorname{Gal}(L|k) \simeq \mathbb{Z}/6 \simeq \mathbb{Z}/2 \times \mathbb{Z}/3$ (the unramified extension would do).

Again we choose $\mathfrak{t}, \mathfrak{s}, R, \ldots$, a generator $\sigma \in \operatorname{Gal}(L|k)$, and an A(R)-invariant scalar product on V. We know that $\sigma_A \in GL(V)$ is of order 6 and operates without eigenvalue 1, so its minimal polynomial has to be a divisor of

$$T^{5} + T^{4} + T^{3} + T^{2} + T + 1 = (T+1)(T^{2} + T + 1)(T^{2} - T + 1)$$

and we look at the cyclic $\mathbb{R}[\sigma_A]$ -modules generated by roots, case by case arriving at a contradiction:

1) Not all those modules can be of \mathbb{R} -dimension 1, since that would imply $\sigma_A = -id$.

2) If there was a two-dimensional one, the roots contained in it would be a system either of type $A_1 \times A_1$ or of type A_2 .

In the first case, one easily sees $\sigma_A^2(\alpha)$ is either α or $-\alpha$. If it is α , $\alpha + \sigma_A(\alpha)$ is fixed by σ_A , hence 0, so $\sigma_A = -$ id and the module is not two-dimensional; if it is $-\alpha$, we have $\sigma_A^2 = -$ id and σ_A has $\pm \sqrt{-1}$ as eigenvalues over an algebraic closure, in contradiction to what we know about the minimal polynomial.

In the second case, on our two-dimensional module, σ_A has minimal polynomial $T^2 + T + 1$, i.e. operates by rotating the roots by 120°. (If it were $T^2 - T + 1$, σ_A would rotate by 60° and we would have an outer form, which cannot be anisotropic.) The roots orthogonal to any system of type A_2 in E_6 form a system of type $A_2 \times A_2$, Galois-stable in our case. If each factor in this pair were Galois-stable, again σ_A would have minimal polynomial $T^2 + T + 1$ on both of them, hence on all V, which is impossible
since it is of order 6. So σ_A has to "switch" the factors. Actually, choose any β in that $A_2 \times A_2$; then we can already name the roots conveniently, so that σ_A maps

$$\alpha_1 := \beta \longmapsto \alpha_6 := \sigma_A(\beta) \longmapsto \alpha_3 := \sigma_A^2(\beta)$$
$$\longmapsto \alpha_5 := \sigma_A^3(\beta) \longmapsto -\alpha_1 - \alpha_3 \longmapsto -\alpha_5 - \alpha_6 \longmapsto \alpha_1$$

and

$$\alpha_0 := \alpha \longmapsto \alpha_2 := \sigma_A(\alpha) \longmapsto -\alpha_2 - \alpha_0$$

in the diagram



This diagram is related to a basis of the whole root system by dropping α_0 and introducing a root α_4 with the relation

$$\alpha_0 = -\alpha_1 - 2\alpha_2 - 2\alpha_3 - 3\alpha_4 - 2\alpha_5 - \alpha_6.$$

But this relation forces σ_A to send the root α_4 to $-\alpha_2 - \alpha_3 - 2\alpha_5 - \alpha_6$, which is not a root, contradicting $\sigma_A(R) = R$.

3) A three-dimensional module generated by a root would contain a root system of rank 3 corresponding to an anisotropic form. Type $A_1 \times A_2$ is excluded by easy combinatorics; type A_3 would not be split by our field extension of degree 6. Remains the case $A_1 \times A_1 \times A_1$. Orthogonal to such a subsystem in E_6 would be a fourth A_1 , stable under σ_A and hence with $\sigma_A = -$ id on it. On our three-dimensional $A_1 \times A_1 \times A_1$ A_1 , one readily sees that $\sigma_A^3 = \pm id$, and "+" is impossible since otherwise for any root in there, $\alpha + \sigma(\alpha) + \sigma^2(\alpha) \neq 0$ would be fixed by σ . We name basis roots conveniently so that σ_A maps

$$\alpha_1 \mapsto -\alpha_0 \mapsto \alpha_6 \mapsto -\alpha_1 \mapsto \alpha_0 \mapsto -\alpha_6$$

and

$$\alpha_4 \mapsto -\alpha_4$$

in the diagram

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017



Now those roots can be amended by roots α_3 , α_5 and α_2 uniquely determined by being at angle 120° to both α_1 and α_4 (resp. α_4 and α_6 , resp. α_4 and α_0), so that $\alpha_1, \ldots, \alpha_6$ are again a basis of R and α_0 is the negative of the highest root in that basis. One checks that this forces σ_A to act on them as follows:

$$\alpha_3 \longmapsto -\alpha_2 \longmapsto \alpha_5 \longmapsto -\alpha_3 \longmapsto \alpha_2 \longmapsto -\alpha_5$$

so that, like in the case D_4 (and indeed we have a subsystem of type D_4 here with basis $\alpha_2, \ldots, \alpha_5$), we find that σ_A operates as -id on the subsystem consisting of

$$\{\pm \alpha_4, \pm (\alpha_2 + \alpha_3 + \alpha_4 + \alpha_5), \pm (\alpha_2 + \alpha_3 + 2\alpha_4 + \alpha_5)\}$$

This subsystem is of type A_2 and would give rise to an anisotropic form of this type, but an *outer* form (since $-id \notin W(A_2)$), which does not exist. Contradiction.

4) On a four-dimensional cyclic module, σ_A would have to operate with minimal polynomial

$$(T2 + T + 1)(T2 - T + 1) = T4 + T2 + 1.$$

Exactly as in the case D_4 (see step 4 in the proof of Lemma 5.5.9) one shows that α and $\sigma_A^2(\alpha)$ form a basis of a sub-root-system of type A_2 ; if $\sigma_A(\alpha)$ was not orthogonal to α , one would arrive at a contradiction (the module would have \mathbb{R} -dimension 2) like there. So necessarily $\sigma_A(\alpha), \sigma_A^3(\alpha)$ are the basis of a root system of type A_2 orthogonal to the other one, so the roots in our module are a system of type $A_2 \times A_2$. But orthogonal to this, and Galois-stable, is another subsystem of type A_2 , and we are back in the case we ruled out in step 2:



5) A five-dimensional module generated by a root would contain a root system of rank 5 corresponding to an anisotropic form. According to our classification of the classical types, there is no anisotropic form of type D_5 . A subsystem of type $D_4 \times A_1$ does not exist in R, all other sub-root systems of rank 5 would be products of some A_n

with the *n*'s adding up to 5. Except for A_5 , all of these are easily ruled out by basic combinatorics, or reduced to cases already treated; remark also that there are now 5 mutually orthogonal roots in R, hence no subsystem of type $(A_1)^5$.

As for the case A_5 , the orthogonal complement of such a sub-root-system is a Galois-invariant subsystem of type A_1 on which $\sigma_A = -id$; name one of its roots α_0 . On the A_5 factor, being anisotropic and split by our degree 6 extension, by corollary 5.3.2 we can name the basis roots $\alpha_1, \alpha_3, \ldots, \alpha_6$ so that σ_A maps

$$\alpha_1 \longmapsto \alpha_3 \longmapsto \alpha_4 \longmapsto \alpha_5 \longmapsto \alpha_6 \longmapsto -\alpha_1 - \alpha_3 - \alpha_4 - \alpha_5 - \alpha_6$$

The diagram to look at here is





and again we can switch to a basis of our original R by dropping α_0 and introducing α_2 subject to

 $\alpha_0 = -\alpha_1 - 2\alpha_2 - 2\alpha_3 - 3\alpha_4 - 2\alpha_5 - \alpha_6.$

With this one computes that σ_A maps

 $\alpha_2 \mapsto -\alpha_2 - \alpha_3 - 2\alpha_4 - 2\alpha_5 - \alpha_6 \mapsto \alpha_1 + \alpha_2 + 2\alpha_3 + 2\alpha_4 + \alpha_5 \mapsto -\alpha_2$

so that the $\mathbb{R}[\sigma_A]$ -module generated by α_2 is of \mathbb{R} -dimension 2 and thus already shown to be impossible: It would be an outer form of type A_2 again.

6) Since the degree of the minimal polynomial of σ_A is ≤ 5 , no cyclic $\mathbb{R}[\sigma_A]$ -module of \mathbb{R} -dimension ≥ 6 exists.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

CHAPTER 6

CONCLUDING REMARKS

6.1. The field $k = \mathbb{R}$

As noted in the introduction, already E. Cartan had classified the semisimple Lie algebras over \mathbb{R} , and the theory of real Lie algebras is strongly connected to that of real Lie groups. A comprehensive treatment is [Hel]. For Lie algebras, [OVi, Chapter 5] uses methods close to ours, including split toral ("diagonalizable") subalgebras, rational root systems and Satake-Tits diagrams, and combines them with classical "real" methods, namely the Cartan involution and Cartan decomposition.

Araki's paper [**Ara**] uses only a few basic lemmata – some of which, however, are peculiar to \mathbb{R} or at least real closed fields – and then goes through all possible Satake-Tits diagrams with lots of combinatorial reasoning. A very streamlined version of this is M. Sugiura's appendix to [**Sat3**], where much is built on Cartan's theorem quoted in 5.2.5 that for every type there exists a unique anisotropic ("compact") real form. With the notation of Section 4.2.1, this compact real form corresponds to the cocycle given by $a_{\sigma} \leftrightarrow (-\operatorname{id}, (-1)_{\alpha \in \Delta})$ where of course σ is the complex conjugation. The twisted Galois action $t(\sigma)$ on this compact form is always given by the opposition involution from Section 4.3.1, so that the compact forms are outer forms for the types A_n $(n \geq 2)$, D_n (n odd) and E_6 , and inner forms for all other types. (The parity distinction for type D_n was visible in the proof of 4.5.15.)

For comprehensive lists of Satake-Tits diagrams, rational root systems and the connection to Cartan's labelling of the forms, see [**OVi**, Table 9] and [**Oni**, Table 5].

6.2. k-rational approaches (Allison, Seligman)

Let k be a field with char(k) = 0. Allison's paper [All] proposes an isomorphism theorem for central simple (= absolutely simple, see Section 4.1) Lie algebras which is "k-rational" in the sense that both its statement and proof do not make use of scalar extension. It starts from maximal split toral subalgebras \mathfrak{s} (called \mathcal{T} there) and corresponding k-rational root systems \overline{R} , like in our Section 3.1.1. The precise statement of the criterion for isomorphism of two such Lie algebras $(\mathfrak{g}, \mathfrak{s})$ and $(\mathfrak{g}', \mathfrak{s}')$ is a little technical, but it can be summarised as

▷ an isomorphism $(\mathfrak{z}_{\mathfrak{g}}(\mathfrak{s}),\mathfrak{s}) \xrightarrow{\simeq} (\mathfrak{z}_{\mathfrak{g}'}(\mathfrak{s}'),\mathfrak{s}')$, which induces an isomorphism of the rational root systems $\overline{R} \simeq \overline{R}'$, which further

- \triangleright is compatible with some extra structures corresponding to the k-rational Weyl groups, and
- \triangleright for "very few" roots $\alpha \in \overline{R}$ and corresponding $\alpha' \in \overline{R}'$, an isomorphism of the root spaces, compatible with everything above.

"Very few" here means that only one root for every root length in \overline{R} has to be taken, and so "very few" has cardinality 1 if \overline{R} is of type A, D or E, cardinality 3 if \overline{R} is of type BC_n with $n \geq 2$, and cardinality 2 in all other cases.

Note that isomorphism of the anisotropic kernels is included in the first point. The compatibility with the Weyl group structure allows the reduction to few root spaces, since the roots are permuted by the Weyl group.

Along the way, some interesting results on Lie algebras are achieved via their rational root systems. For example [All, Cor. 6.6] says that an absolutely simple \mathfrak{g} with rational root system \overline{R} of type D or E is automatically split (i.e. $R = \overline{R}$); one can indeed check directly that our method from Section 3.4.1 can never produce a root system of these types (except in the split case). The "rational" method uses much more calculation with ideals and subalgebras. See Seligman's [Sel2] and [Sel3] for an extensive version of this approach.

6.3. Explicit constructions for exceptional types

Besides the connection of the classical types to certain associative algebras like we have presented it, there are similar constructions for the exceptional types. One highlight is "Freudenthal's magic square" which in a way gives a construction for all exceptional types at once; except for Freudenthal's works, see e.g. [**Tit2**] and [**Vin**]. In [**GPe**], the Satake-Tits diagrams for some forms of type E_6 in this construction are computed.

We mention one recent invention: In [Inv, X], the authors construct trialitarian algebras which produce the trialitarian types of our Lie algebras. The idea is close in spirit to the construction for the classical types. Like it was necessary there, in the type AII, to allow as underlying objects pairs of associative algebras (which were "linked" by a switching involution), here one has triples of algebras which are linked by an extra structure. This allows to catch the extra operation of the symmetric group S_3 and gives further insights into phenomena like the one mentioned in 4.5.20.ii and 5.4.7: Here it turns out that each of the algebras ($M_4(Q), \iota_1$) and ($M_8(k), \iota_2$) is a factor of the Clifford algebra of the other.

- [Alb] Albert, A. Adrian. Involutorial Simple Algebras and Real Riemann Matrices. Annals of Mathematics 36 (1935), pp. 886–964.
- [All] Allison, B. N. Isomorphism of Simple Lie Algebras. Transactions of the American Mathematical Society 177 (1973), pp. 173–190.
- [Ara] Araki, Shôrô. On root systems and an infinitesimal classification of irreducible symmetric spaces. Journal of Mathematics, Osaka City University 13 (1962), pp. 1–34.
- [Bo1] Bourbaki, Nicolas. Algèbre. Hermann, Paris 1959/Masson, Paris et al., 1970–1981/Springer, Berlin et al., 2007/2012.
- [Bo2] Bourbaki, Nicolas. Groupes et algèbres de Lie. Hermann, Paris 1972– 1975/Masson, Paris et al., 1981–1990/Springer, Berlin et al., 2006/2007.
- [BT1] Borel, Armand and Tits, Jacques. Groupes réductifs. Publications Mathematiques de l'Institut des Hautes Études Scientifiques 27 (1965), pp. 55–151.
- [BT2] Bruhat, F. and Tits, J. Groupes algébriques [sic] sur un corps local. Chapitre III. Compléments et applications à la cohomologie galoisienne. Journal of the Faculty of Sciences of the University of Tokyo 34 (1987), pp. 671– 698.
- [Car] Carter, R. W. Conjugacy classes in the Weyl group. Compositio Mathematica 25 (1972), pp. 1–59.
- [Che] Chevalley, Claude. Sur certains groupes simples. Tohoku Mathematical Journal 7 (1955), pp. 14–66.
- [**GPe**] Garibaldi, Skip and Petersson, Holger P. *Groups of outer type* E_6 with trivial Tits algebras. Transformation Groups **12** (2007), pp. 443–474.

- [Has] Hasse, Helmut. Über p-adische Schiefkörper und ihre Bedeutung für die Arithmetik hyperkomplexer Zahlsysteme. Mathematische Annalen 104 (1931), pp. 495–534.
- [Hec] Heck, André. Involutive automorphisms of root systems. Journal of the Mathematical Society of Japan 36 (1984), pp. 644–658.
- [Hel] Helgason, Sigurdur. Differential geometry, Lie groups, and symmetric spaces. Academic Press, London New York San Francisco, 1978.
- [Hum] Humphreys, James E. Introduction to Lie Algebras and Representation Theory (Graduate Texts in Mathematics 9). Springer, Berlin et al., 1972.
- [Inv] Knus, M.-A.; Merkurjev, A.; Rost, M.; Tignol, J.-P. The Book of Involutions (American Mathematical Society Colloquium Publications 44). American Mathematical Society, Providence, 1998.
- [Jac1] Jacobson, Nathan. Rational methods in the theory of Lie algebras. Annals of Mathematics 36 (1935), pp. 875–881.
- [Jac2] Jacobson, Nathan. A note on non-associative algebras. Duke Mathematical Journal 3 (1937), pp. 544–548.
- [Jac3] Jacobson, Nathan. A class of normal simple Lie algebras of characteristic zero. Annals of Mathematics 38 (1937), pp. 508–517.
- [Jac4] Jacobson, Nathan. Simple Lie algebras over a field of characteristic zero. Duke Mathematical Journal 4 (1938), pp. 534–551.
- [Jac5] Jacobson, Nathan. A note on hermitian forms. Bulletin of the American Mathematical Society 46 (1940), pp. 264–268.
- [Jac6] Jacobson, Nathan. Lie Algebras (Interscience Tracts in Pure and Applied Mathematics 10). Interscience Publishers, New York London, 1962.
- [Jac7] Jacobson, Nathan. Exceptional Lie Algebras (Lecture Notes in Pure and Applied Mathematics 1). Marcel Dekker, New York, 1971.
- [Kne] Kneser, Martin. Galois-Kohomologie halbeinfacher algebraischer Gruppen über p-adischen Körpern. I: Mathematische Zeitschrift 88 (1965), pp. 40–47. II: Mathematische Zeitschrift 89 (1965), pp. 250–272.
- [Lam] Lam, T. Y. The Algebraic Theory of Quadratic Forms. W.A. Benjamin, Reading, 1973.
- [Lan1] Landherr, Walther. Über einfache Liesche Ringe. Abhandlungen aus dem Mathematischen Seminar der Universität Hamburg 11 (1935), pp. 41–64.

- [Lan2] Landherr, Walther. Liesche Ringe vom Typus A über einem algebraischen Zahlkörper (Die lineare Gruppe) und hermitesche Formen über einem Schiefkörper. Abhandlungen aus dem Mathematischen Seminar der Universität Hamburg 12 (1937), pp. 200–241.
- [Lang] Lang, Serge. On quasi algebraic closure. Annals of Mathematics 55 (1952), pp. 373–390.
- [Oni] Onishchik, Arkady L. Lectures on Real Semisimple Lie Algebras and Their Representations (ESI Lectures in Mathematics and Physics). European Mathematical Society Publishing House, Zürich, 2004.
- [OVi] Onishchik, A.L. and Vinberg, E.B. *Lie Groups and Algebraic Groups*. Springer, Berlin et al., 1990.
- [Pie] Pierce, Richard S. Associative Algebras (Graduate Texts in Mathematics 88). Springer, Berlin et al., 1982.
- [PRa] Platonov, Vladimir and Rapinchuk, Andrei. Algebraic Groups and Number Theory. Academic Press, London New York San Francisco, 1994.
- [PSt] Petrov, Victor and Stavrova, Anastasia. The Tits indices over semilocal rings. Transformation Groups 16 (2011), pp. 193–217.
- [Rei] Reiner, Irving. Maximal Orders (LMS Monographs 5). Academic Press, London New York San Francisco, 1975.
- [Sat1] Satake, Ichiro. On representations and compactifications of symmetric Riemannian spaces. Annals of Mathematics 71 (1960), pp. 77–110.
- [Sat2] Satake, Ichiro. On the theory of reductive algebraic groups over a perfect field. Journal of the Mathematical Society of Japan 15 (1963), pp. 210–235.
- [Sat3] Satake, Ichiro. Classification theory of semi-simple algebraic groups (Lecture Notes in Pure and Applied Mathematics 3). Marcel Dekker, New York, 1971.
- [Sel1] Seligman, George B. Modular Lie Algebras (Ergebnisse der Mathematik und ihrer Grenzgebiete 40). Springer, Berlin et al., 1967.
- [Sel2] Seligman, George B. Rational methods in Lie algebras (Lecture Notes in Pure and Applied Mathematics 17). Marcel Dekker, New York, 1976.
- [Sel3] Seligman, George B. Constructions of Lie Algebras and their Modules (Lecture Notes in Mathematics 1300). Springer, Berlin et al., 1988.

- [Ser1] Serre, Jean-Pierre. Cohomologie galoisienne des groupes algébriques linéaires. In: Colloque sur la théorie des groupes algébriques (Bruxelles 1962), pp. 53–68.
- [Ser2] Serre, Jean-Pierre. Local Fields (Graduate Texts in Mathematics 67). Springer, Berlin et al., 1995.
- [Ser3] Serre, Jean-Pierre. Galois Cohomology. Springer, Berlin et al., 1997.
- [Sch] Schattschneider, Doris J. On restricted roots of semi-simple algebraic groups. Journal of the Mathematical Society of Japan 21 (1969), pp. 94–115.
- [Spr1] Springer, T.A. Quelques résultats sur la cohomologie galoisienne. In: Colloque sur la théorie des groupes algébriques (Bruxelles 1962), pp. 129–135.
- [Spr2] Springer, T.A. Linear Algebraic Groups. In: Parshin, A. N. and Shafarevich, I.R. (editors): Algebraic Geometry IV (Encyclopedia of Mathematical Sciences 55), Springer, Berlin et al., 1994, pp. 1–121.
- [Spr3] Springer, T.A. *Linear Algebraic Groups*, Second Edition (Progress in Mathematics 9). Birkhäuser, Boston, 1998.
- [Ste] Steinberg, Robert. Regular elements of semi-simple algebraic groups. Publications Mathematiques de l'Institut des Hautes Études Scientifiques 25 (1965), pp. 49–80. Reproduced as appendix 1 in [Ser3, III].
- [Tit1] Tits, Jacques. Classification of Algebraic Semisimple Groups. In: Algebraic Groups and Discontinuous Subgroups. Proceedings of symposia in pure mathematics, IX (Boulder 1965), pp. 33–62.
- [Tit2] Tits, Jacques. Algèbres alternatives, algèbres de Jordan et algèbres de Lie exceptionnelles. I. Construction. Indagationes Mathematicae 28 (1966), pp. 223–237.
- [Tit3] Tits, Jacques. Reductive groups over local fields. In: Automorphic forms, representations and L-functions. Proceedings of symposia in pure mathematics XXXIII (Corvallis 1977), pp. 29–70.
- [Tit4] Tits, Jacques. Strongly Inner Anisotropic Forms of Simple Algebraic Groups. Journal of Algebra 131 (1990), pp. 648–677.
- [TiW] Tits, J. and Weiss, R. M. *Moufang Polygons*. Springer, Berlin et al., 2002.
- [Tsu] Tsukamoto, Takashi. On the local theory of quaternionic anti-hermitian forms. Journal of the Mathematical Society of Japan 13 (1961), pp. 387–400.

- [Vin] Vinberg, E. B. Construction of the exceptional simple Lie algebras. American Mathematical Society Translations 213 (2005), pp. 241–242. (Original article from 1966.)
- [Wei] Weil, André. Algebras with involutions and the classical groups. Journal of the Indian Mathematical Society 24 (1960), pp. 589–623.
- [We1] Veĭsfeĭler, Boris Ju. (= Weisfeiler, Boris). Classification of semisimple Lie algebras over a p-adic field. Soviet Mathematics Doklady 5 (1964), pp. 1206– 1208.
- [We2] Veĭsfeĭler, Boris Ju. (= Weisfeiler, Boris). Some properties of exceptional semisimple algebraic groups over nonclosed fields. Transactions of the Moscow Mathematical Society 20 (1969), pp. 109–134.
- [We3] Veĭsfeĭler, Boris Ju. (= Weisfeiler, Boris). Semisimple algebraic groups which are split over a quadratic extension. Mathematics of the USSR-Izvestija 5 (1971), pp. 57–72.
- [Win] Winter, David J. Abstract Lie Algebras. The MIT Press, Cambridge and London, 1972.
- [Wit] Witt, Ernst. Theorie der quadratischen Formen in beliebigen Körpern. Journal für die reine und angewandte Mathematik 176 (1937), pp. 31–44.
- [Xue] Xue, Xiaoxi. A note on the centralizer of a maximal toral subalgebra of a semisimple Lie algebra. Journal of Systems Science and Mathematical Sciences. Xitong Kexue yu Shuxue 4 (1984).2, p. 102.

MÉMOIRES DE LA SOCIÉTÉ MATHÉMATIQUE DE FRANCE Nouvelle série

2017

- 151. T. SCHOENEBERG Semisimple Lie Algebras and their classification over p-adic Fields
- 150. P.G. LEFLOCH , YUE MA The Mathematical Validity of the f(R) Theory of Modified Gravity

2016

- 149. BEUZART-PLESSIS R. La conjecture locale de Gross-Prasad pour les représentations tempérées des groupes unitaires
- 148. MOKHTAR-KHARROUBI M. Compactness properties of perturbed sub-stochastic C_0 semigroups on $L^1(\mu)$ with applications to discreteness and spectral gaps
- 147. CHITOUR Y., KOKKONEN P. Rolling of manifolds and controllability in dimension three
- 146. KARALIOLIOS, N. Global aspects of the reducibility of quasiperiodic cocycles in compact Lie groups
- 145. V. BONNAILLIE-NOËL, M. DAUGE, N. Ground state energy of the magnetic Laplacian on corner domains
- 144. P. AUSCHER, S. STAHLHUT A priori estimates for boundary value elliptic problems via first order systems & A priori estimates for boundary value elliptic problems via first order systems

2015

- 143. R. DANCHIN, P.B. MUCHA Critical functional framework and maximal regularity in action on systems of incompressible flows
- 142. Joseph AYOUB Motifs des variétés analytiques rigides
- 140/141. Yong LU, Benjamin TEXIER A Stability Criterion for High-Frequency Oscillations

2014

- 138/139. Takuro MOCHIZUKI Holonomic D-Modules with Betti Structures
 - 137. Paul SEIDEL Abstract Analogues of Flux as Symplectic Invariants
 - 136. Johannes SJÖSTRAND Weyl law for semi-classical resonances with randomly perturbed potentials

2013

- 135. L. PRELLI Microlocalization of subanalytic sheaves
- 134. P. BERGER Persistence of Stratification of Normally Expanded Laminations
- 133. L. DESIDERI Problème de Plateau, équations fuchsiennes et problème de Riemann Hilbert
- 132. X. BRESSAUD, N. FOURNIER One Dimensional General Forest Fire Processes

2012

- 130-131. Y. NAKKAJIMA Weight filtration and slope filtration on the rigid cohomology of a variety in characteristic p > 0
 - 129. W. A STEINMETZ-ZIKESCH Algèbres de Lie de dimension infinie et théorie de la descente
 - 128. D. DOLGOPYAT Repulsion from resonances

2011

127. B. LE STUM – The overconvergent site

- 125/126. J. BERTIN, M. ROMAGNY Champs de Hurwitz
 - 124. G. HENNIART, B. LEMAIRE Changement de base et induction automorphe pour GL_n en caractéristique non nulle

2010

- 123. C.-H. HSIAO Projections in several complex variables
- 122. H. DE THÉLIN, G. VIGNY Entropy of meromorphic maps and dynamics of birational maps
- 121. M. REES A Fundamental Domain for V_3
- 120. H. CHEN Convergence des polygones de Harder-Narasimhan

2009

- 119. B. DEMANGE Uncertainty principles associated to non-degenerate quadratic forms
- 118. A. SIEGEL, J. M. THUSWALDNER Topological properties of Rauzy fractals
- 117. D. HÄFNER Creation of fermions by rotating charged black holes
- 116. P. BOYER Faisceaux pervers des cycles évanescents des variétés de Drinfeld et groupes de cohomologie du modèle de Deligne-Carayol

2008

- 115. R. ZHAO, K. ZHU Theory of Bergman Spaces in the Unit Ball of \mathbb{C}^n
- 114. M. ENOCK Measured quantum groupoids in action
- 113. J. FASEL Groupes de Chow orientés
- 112. O. BRINON Représentations p-adiques cristallines et de de Rham dans le cas relatif

2007

- 111. A. DJAMENT Foncteurs en grassmanniennes, filtration de Krull et cohomologie des foncteurs
- 110. S. SZABÓ Nahm transform for integrable connections on the Riemann sphere
- 109. F. LESIEUR Measured quantum groupoids
- 108. J. GASQUI, H. GOLDSCHMIDT Infinitesimal isospectral deformations of the Grassmannian of 3-planes in \mathbb{R}^6

2006

- 107. I. GALLAGHER, L. SAINT-RAYMOND Mathematical study of the betaplane model: Equatorial waves and convergence results
- 106. N. BERGERON Propriétés de Lefschetz automorphes pour les groupes unitaires et orthogonaux
- 105. B. HELFFER, F. NIER Quantitative analysis of metastability in reversible diffusion processes via a Witten complex approach: the case with boundary
- A. FEDOTOV, F. KLOPP Weakly resonant tunneling interactions for adiabatic quasi-periodic Schrödinger operators

Instructions aux auteurs

Les *Mémoires* de la SMF publient, en français ou en anglais, de longs articles de recherche, des monographies originales ou des séries d'articles centrés sur un même sujet. Les volumes ont entre 100 et 150 pages environ.

Le manuscrit doit être envoyé en *double* exemplaire au secrétariat des publications en précisant le nom de la revue.

Le fichier source T_EX (un seul fichier par article ou monographie) peut aussi être envoyé par courrier électronique ou ftp, sous réserve que sa compilation par le secrétariat SMF soit possible. Contacter le secrétariat à l'adresse électronique revues@smf.ens.fr pour obtenir des précisions.

La SMF recommande vivement l'utilisation d' \mathcal{AMS} -LATEX avec les classes smfart ou smfbook, disponibles ainsi que leur documentation sur le serveur http://smf.emath.fr/ ou sur demande au secrétariat des publications SMF. Les fichiers $\mathcal{AMS}\text{-}\ensuremath{\text{LTEX}}$ au format amsart ou amsbook, ainsi que les fichiers $\ensuremath{\text{LTEX}}$ au format article ou book sont aussi les bienvenus. Ils seront saisis suivant les normes suivantes :

- taille des caractères égale à 10 points (option 10pt);
- largeur du texte (textwidth) de 13 cm;
- hauteur du texte (textheight) égale à 21.5 cm;
- le texte étant en outre centré sur une feuille A4 (option a4paper).

Les autres formats TEX et les autres types de traitement de texte ne sont pas utilisables par le secrétariat et sont *fortement* déconseillés.

Avant de saisir leur texte, les auteurs sont invités à prendre connaissance du document *Recommandations aux auteurs* disponible au secrétariat des publications de la SMF ou sur le serveur de la SMF.

Instructions to Authors

In the *Mémoires* of the SMF are published, in French or in English, long research articles, original research monographs or series of articles on the same topic of current research. Each volume should be approximately 100-150 pages long.

Two copies of the original manuscript should be sent to the editorial board of the SMF, indicating to which publication the paper is being submitted.

The TEX source file (a single file for each article or monograph) may also be sent by electronic mail or ftp, in a format suitable for typesetting by the secretary. Please, send an email to revues@smf.ens.fr for precise information.

The SMF has a *strong* preference for \mathcal{AMS} -LATEX together with the documentclasses smfart or smfbook, available with their user's guide at http://smf.emath.fr/ (Internet) or on request from the editorial board of the SMF. The \mathcal{AMS} - \mathbb{ATEX} files using the documentclasses amsart or amsbook, or the \mathbb{ATEX} files using the documentclasses article or book are also encouraged. They will be prepared following the rules below:

- font size equal to 10 points (10pt option);
- text width (textwidth): 13 cm;
- text height (textheight): 21.5 cm;
- the text being centered on a A4 page (a4paper option).

Files prepared with other TEX dialects or other word processors cannot be used by the editorial board and are *not* encouraged.

Before preparing their electronic manuscript, the authors should read the *Advice to authors*, available on request from the editorial board of the SMF or from the web site of the SMF. We study semisimple Lie algebras over fields of characteristic zero, with emphasis on \mathfrak{p} -adic fields and aiming at classification. We first transfer parts of the structure theory of reductive algebraic groups to our setting, with some variations. Classifying invariants are attached to Lie algebras and visualised with Satake-Tits diagrams. We give necessary and sufficient criteria for these diagrams. Over general fields of characteristic zero, we then classify all quasi-split forms, and we adapt an older classification theory for the classical types A-D to our language. Finally we focus on \mathfrak{p} -adic fields, where we achieve a complete classification by combination of certain well-known properties of these fields with our general results and methods, and we discuss the relation of this with a theorem of Kneser. This extends work by Weisfeiler.

Nous étudions les algèbres de Lie semi-simples sur les corps de caractéristique nulle, où l'accent est mis sur les corps p-adiques, l'objectif étant leur classification. Nous transférons d'abord certaines parties de la théorie de la structure des groupes réductifs dans notre contexte, avec quelques variations. Des invariants classifiants sont attachés aux algèbres de Lie et sont visualisés à l'aide de diagrammes de Satake-Tits. Nous donnons des critères nécessaires et suffisants pour ces diagrammes. Sur les corps généraux de caractéristique nulle, nous classifions ensuite toutes les formes quasi-déployées et nous traduisons une théorie ancienne de classification pour les types classiques A-D dans notre langue. Nous mettons enfin l'accent sur les corps p-adiques, où nous obtenons une classification complète par combinaison de certaines propriétés bien connues sur ces corps avec nos résultats généraux et nos méthodes, et nous abordons la relation de ces résultats avec un théorème de Kneser. Tout cela prolonge un travail de Weisfeiler.